UNITED STATES SECURITIES AND EXCHANGE COMMISSION

WASHINGTON D.C. 20549

FORM 10-Q

(Mark One)

☐ Quarterly Report Pursuant to Section 13 or 15(d) of the Securities Exchange Act of 1934

For the Quarterly Period Ended March 31, 2010

Or

o Transition Report Pursuant to Section 13 or 15(d) of the Securities Exchange Act of 1934

For the Transition Period from

to

Commission file number 001-10716

TRIMAS CORPORATION

(Exact name of registrant as specified in its charter)

Delaware

38-2687639

(State or other jurisdiction of incorporation or organization)

(IRS Employer Identification No.)

39400 Woodward Avenue, Suite 130 Bloomfield Hills, Michigan 48304

(Address of principal executive offices, including zip code)

(248) 631-5450

(Registrant's telephone number, including area code)

Indicate by check mark whether the Registrant (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding 12 months (or for such shorter period that the Registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days. Yes \boxtimes No o

Indicate by check mark whether the registrant has submitted electronically and posted on its corporate Web site, if any, every Interactive Data File required to be submitted and posted pursuant to Rule 405 of Regulation S-T during the preceding 12 months (or for such shorter period that the registrant was required to submit and post such files). Yes o No o

Indicate by check mark whether the registrant is a large accelerated filer, an accelerated filer, a non-accelerated filer, or a smaller reporting company. See definitions of "large accelerated filer," "accelerated filer," and "smaller reporting company" in Rule 12b-2 of the Exchange Act. (Check one):

Large accelerated filer o

Accelerated filer \boxtimes

Non-accelerated filer o (Do not check if a smaller reporting company)

Smaller reporting company o

Indicate by check mark whether the registrant is a shell company (as defined in Rule 12b-2 of the Exchange Act). Yes o No 🗵

As of April 30, 2010, the number of outstanding shares of the Registrant's common stock, \$.01 par value, was 34,008,016 shares.

Index

Part I.	<u>Financial</u>	<u>Information</u>	
		Forward-Looking Statements	2
	Item 1.	Consolidated Financial Statements	<u>3</u>
		Consolidated Balance Sheet—March 31, 2010 and December 31, 2009	<u>3</u>
		Consolidated Statement of Operations for the Three Months Ended March 31, 2010 and 2009	<u>4</u>
		Consolidated Statement of Cash Flows for the Three Months Ended March 31, 2010 and 2009	<u>5</u>
		Consolidated Statement of Shareholders' Equity for the Three Months Ended March 31, 2010	<u>6</u>
		Notes to Unaudited Consolidated Financial Statements	<u>Z</u>
	Item 2.	Management's Discussion and Analysis of Financial Condition and Results of Operations	<u>29</u>
	Item 3.	Quantitative and Qualitative Disclosure about Market Risk	<u>47</u>
	Item 4.	Controls and Procedures	<u>48</u>
Part II.	Other Info	rmation and Signatures	
	Item 1.	<u>Legal Proceedings</u>	<u>49</u>
	Item 1A.	Risk Factors	<u>49</u>
	Item 2.	<u>Unregistered Sales of Equity Securities and Use of Proceeds</u>	<u>49</u>
	Item 3.	Defaults Upon Senior Securities	<u>49</u>
	Item 4.	Submission of Matters to a Vote of Security Holders	<u>49</u>
	Item 5.	Other Information	<u>49</u>
	Item 6.	<u>Exhibits</u>	<u>49</u>
	<u>Signature</u>		<u>55</u>
		1	

Forward-Looking Statements

This report contains forward-looking statements (as that term is defined by the federal securities laws) about our financial condition, results of operations and business. You can find many of these statements by looking for words such as "may," "will," "expect," "anticipate," "believe," "estimate" and similar words used in this report.

These forward-looking statements are subject to numerous assumptions, risks and uncertainties. Because the statements are subject to risks and uncertainties, actual results may differ materially from those expressed or implied by the forward-looking statements. We caution readers not to place undue reliance on the statements, which speak only as of the date of this report.

The cautionary statements set forth above should be considered in connection with any subsequent written or oral forward-looking statements that we or persons acting on our behalf may issue. We do not undertake any obligation to review or confirm analysts' expectations or estimates or to release publicly any revisions to any forward-looking statement to reflect events or circumstances after the date of this report or to reflect the occurrence of unanticipated events.

You should carefully consider the factors discussed in Part I, Item 1A, "Risk Factors," in our Annual Report on Form 10-K for the year ended December 31, 2009, which could materially affect our business, financial condition or future results. The risks described in our Annual Report on Form 10-K are not the only risks facing our Company. Additional risks and uncertainties not currently known to us or that we currently deemed to be immaterial also may materially adversely affect our business, financial position and results of operations or cash flows.

We disclose important factors that could cause our actual results to differ materially from our expectations under Part I, Item 2., "Management's Discussion and Analysis of Financial Condition and Results of Operations," and elsewhere in this report. These cautionary statements qualify all forward-looking statements attributed to us or persons acting on our behalf. When we indicate that an event, condition or circumstance could or would have an adverse effect on us, we mean to include effects upon our business, financial and other condition, results of operations, prospects and ability to service our debt.

PART I. FINANCIAL INFORMATION

Item 1. Financial Statements

TriMas Corporation

Consolidated Balance Sheet

(Unaudited—dollars in thousands)

	March 31, 2010		De	cember 31, 2009
Assets				
Current assets:				
Cash and cash equivalents	\$	6,630	\$	9,480
Receivables, net of reserves of approximately \$5.7 million as of March 31, 2010 and				
December 31, 2009		128,650		93,380
Inventories		135,730		141,840
Deferred income taxes		24,320		24,320
Prepaid expenses and other current assets		6,420		6,500
Assets of discontinued operations held for sale	_	4,070		4,250
Total current assets		305,820		279,770
Property and equipment, net		157,430		162,220
Goodwill		194,120		196,330
Other intangibles, net		161,410		164,080
Other assets	_	23,170		23,380
Total assets	\$	841,950	\$	825,780
Liabilities and Shareholders' Equity				
Current liabilities:				
Current maturities, long-term debt	\$	12,720	\$	16,190
Accounts payable		103,150		92,840
Accrued liabilities		63,670		65,750
Liabilities of discontinued operations		1,040		1,070
Total current liabilities		180,580		175,850
Long-term debt		505,800		498,360
Deferred income taxes		43,790		42,590
Other long-term liabilities		46,190		47,000
Total liabilities		776,360		763,800
Preferred stock \$0.01 par: Authorized 100,000,000 shares;				
Issued and outstanding: None		_		_
Common stock, \$0.01 par: Authorized 400,000,000 shares;				
Issued and outstanding 34,010,616 shares at March 31, 2010 and 33,895,503 shares at				
December 31, 2009		340		330
Paid-in capital		529,020		528,370
Accumulated deficit		(504,950)		(510,380)
Accumulated other comprehensive income		41,180	_	43,660
Total shareholders' equity		65,590		61,980
Total liabilities and shareholders' equity	\$	841,950	\$	825,780
	_		_	

Consolidated Statement of Operations

(Unaudited—dollars in thousands, except for share amounts)

		Three months ended March 31,			
N-41	\$	2010	ch.	2009	
Net sales Cost of sales	\$	220,060	\$	201,720	
		(157,000)	_	(155,260)	
Gross profit		63,060		46,460	
Selling, general and administrative expenses Gain (loss) on dispositions of property and equipment		(37,700)		(41,300) 40	
	<u> </u>	25,050	_	5,200	
Operating profit	_	25,050	_	5,200	
Other income (expense), net:		(1.4.1.40)		(12, 400)	
Interest expense		(14,140)		(12,480)	
Gain on extinguishment of debt Other, net		(510)		15,310 (700)	
Other income (expense), net		(14,650)		2,130	
Income from continuing operations before income tax expense		10,400		7,330	
Income tax expense		(4,650)		(2,710)	
Income from continuing operations	<u> </u>	5,750	_	4,620	
Loss from discontinued operations, net of income tax benefit		(320)		(8,300)	
Net income (loss)	\$	5,430	\$	(3,680)	
Earnings per share—basic:	=				
Continuing operations	\$	0.17	\$	0.14	
Discontinued operations, net of income tax benefit		(0.01)		(0.25)	
Net income (loss) per share	\$	0.16	\$	(0.11)	
Weighted average common shares—basic		33,569,677		33,459,502	
Earnings per share—diluted:	_				
Continuing operations	\$	0.17	\$	0.14	
Discontinued operations, net of income tax benefit		(0.01)		(0.25)	
Net income (loss) per share	\$	0.16	\$	(0.11)	
Weighted average common shares—diluted	=	34,314,020	_	33,487,526	

Consolidated Statement of Cash Flows

(Unaudited—dollars in thousands)

-		Three months ended March 31,			
Net income (loss) \$ 5,400 \$ (3,680) Adjustments to reconcile net income (loss) to net cash provided by (used for) operating activities, net of acquisition impact: ************************************			2010		2009
Adjustments to reconcile net income (loss) to net cash provided by (used for) operating activities, net of acquisition impact: (Gain) loss on dispositions of property and equipment 6,000 8,110 Amortization of intangible assets 7,30 6,500 Amortization of debt issue costs 730 6,500 Deferred income taxes 7,30 6,500 Gain on extinguishment of debt 7,400 Gain on extinguishment of debt 8,300 Non-cash compensation expense 480 30 Note proceeds from (reductions in) sale of receivables and receivables securitization 3,80 (6,130) Increase in receivables 6,606 18,090 Decrease in inventories 6,606 18,090 Decrease in inventories 6,606 18,090 Decrease in prepaid expenses and other assets 7,000 Increase in accounts payable and accrued liabilities 7,100 Net cash provided by (used for) operating activities, net of acquisition impact 7,100 Asset Flows from Investing Activities 7,100 Ret proceeds from disposition of businesses and other assets 30 20,680 Net cash provided by (used for) investing activities net of acquisition impact 7,100 Cash Flows from Investing Activities 7,100 Repayments of borrowings on term loan facilities 7,100 Repayments of borrowings on revolving credit facilities 134,90 27,200 Repayments of borrowings on revolving credit facilities 134,90 27,200 Repayments of borrowings on revolving credit facilities 134,90 27,000 Repayments of borrowings on revolving credit facilities 134,90 27,000 Repayments of borrowings on revolving credit facilities 134,90 27,000 Repayments of borrowings on revolving credit facilities 14,000 Repayments of borrowings on r					(0.000)
Inest of acquisition impact: (50) (50) (Gain) loss on dispositions of property and equipment 6,020 8,110 Depreciation 6,020 8,110 Amortization of intangible assets 3,590 3,650 Amortization of debt issue costs 380 (2,740) Deferred income taxes 380 (2,740) Gain on extinguishment of debt (15,310) Non-cash compensation expense 480 30 Net proceeds from (reductions in) sale of receivables and receivables securitization 3,830 (6,130) Net crase in inventories (3,600) 12,630 Decrease in inventories 270 1,660 Increase in accounts payable and accrued liabilities 7,910 1,180 Other, net (2,500) 1,200 Net cash provided by (used for) operating activities, net of acquisition impact (2,500) 1,800 Net cash provided by (used for) operating activities and acquisition impact (2,500) 1,820 Net cash provided by (used for) investing activities (2,500) 1,820 Seath Flows from Financing Activities <t< td=""><td></td><td>\$</td><td>5,430</td><td>\$</td><td>(3,680)</td></t<>		\$	5,430	\$	(3,680)
(Gain) loss on dispositions of property and equipment 310 (50) Depreciation 6,02 8,110 Amortization of intangible assets 3,550 3,550 Amortization of debt issue costs 730 610 Deferred income taxes (380) (2,740) Gain on extinguishment of debt - (15,310) Non-cash compensation expense 480 30 Net proceeds from (reductions in) sale of receivables and receivables securitization 3,830 (6,130) Increase in receivables (38,960) 2,630 Decrease in prepaid expenses and other assets 270 1,660 Increase in accounts payable and accrued liabilities 7,910 1,180 Other, net 620 10,900 Net cash provided by (used for) operating activities, net of acquisition impact 4,090 1,800 Tapital expenditures (2,590) 3,280 Net cash provided by (used for) investing activities, net of acquisition impact 2,590 3,280 Net cash provided by (used for) investing activities (2,590) 3,280 Tapital expenditures (2,590) </td <td></td> <td></td> <td></td> <td></td> <td></td>					
Depreciation 6,020 8,110 Amortization of intangible assets 3,590 3,550 Amortization of debt issue costs 730 610 Deferred income taxes (380) (2,740) Gain on extinguishment of debt — (15,310) Non-cash compensation expense 480 30 Net proceeds from (reductions in) sale of receivables and receivables securitization 3,830 (2,630) Decrease in incecivables 6,600 18,090 Decrease in prepaid expenses and other assets 2,70 1,660 Increase in accounts payable and accrued liabilities 7,910 1,180 Other, net 620 (990) Net cash provided by (used for) operating activities, net of acquisition impact 4,090 1,800 Net cash provided by (used for) operating activities, net of acquisition impact 2,590 3,280 Net cash provided by (used for) operating activities, net of acquisition impact 2,590 3,280 Net cash provided by (used for) investing activities 2,590 3,280 Net cash provided by (used for) investing activities (2,590 1,740			210		(=0)
Amortization of intangible assets 3,590 3,650 Amortization of debt issue costs 730 610 Deferred income taxes (2,740) Gain on extinguishment of debt — (15,310) Non-cash compensation expense 440 30 Net proceeds from (reductions in) sale of receivables and receivables securitization 38,360 (2,530) Increase in receivables 6,060 18,090 Decrease in inventories 6,060 18,090 Decrease in inventories 6,060 19,090 Increase in accounts payable and accrued liabilities 7,910 1,180 Other, net 620 (990) Net cash provided by (used for) operating activities, net of acquisition impact (4,090) 1,800 Cash Flows from Investing (2,590) (3,280) Net cash provided by (used for) operating activities (2,590) (3,280) Net cash provided by (used for) investing activities (2,590) (3,280) Text cash provided by (used for) investing activities (2,500) (770) Proceeds from borrowings on revolving credit facilities (30,00)					
Amortization of debt issue costs 730 610 Deferred income taxes (380) (2,740) Gain on extinguishment of debt 40 35 Non-cash compensation expense 480 30 Net proceeds from (reductions in) sale of receivables and receivables securitization 3,830 (6,130) Decrease in inventories 6,060 18,090 Decrease in inventories 6,060 18,090 Decrease in prepaid expenses and other assets 270 1,660 Increase in accounts payable and accrued liabilities 7,910 1,180 Other, net 620 1990 Net cash provided by (used for) operating activities, net of acquisition impact 4,090 1,800 Test proceeds from Investing Activities 2,590 3,280 Net proceeds from disposition of businesses and other assets 30 20,880 Net cash provided by (used for) investing activities (2,590 3,280 Net cash provided by (used for) investing activities (2,590 3,280 Tespayments of borrowings on term loan facilities (3 (2,590 2,700 Repaymen					
Deferred income taxes (380) (2,740) Gain on extinguishment of debt — (15,310) Non-cash compensation expense 30 Net proceeds from (reductions in) sale of receivables and receivables securitization 3,830 (6,130) Increase in receivables (38,960) 2,630 Decrease in inventories 6,060 18,090 Decrease in inventories 7,910 1,180 Other, net 620 (990) Net cash provided by (used for) operating activities, net of acquisition impact (4,090) 1,800 Cash Flows from Investing (2,590) (3,280) Ret proceeds from disposition of businesses and other assets 30 20,680 Net cash provided by (used for) investing activities (2,590) (3,280) Net cash provided by (used for) investing activities (2,590) (3,280) Net cash provided by (used for) investing activities (2,590) (7,700) Repayments of borrowings on term loan facilities (30,600) (7,700) Repayments of borrowings on revolving credit facilities (34,000) (274,680) Repayments of borrowin					
Gain on extinguishment of debt (15,310) Non-cash compensation expense 48 30 Net proceeds from (reductions in) sale of receivables and receivables securitization 3,830 (6,130) Increase in receivables (39,600) (2,630) Decrease in inventories 6,060 18,090 Decrease in prepaid expenses and other assets 270 1,660 Increase in accounts payable and accrued liabilities 7,910 1,180 Other, net 620 (990) Net cash provided by (used for) operating activities, net of acquisition impact (4,090) 1,800 Cash Flows from Investing Activities: 2 (990) Net proceeds from disposition of businesses and other assets 30 20,680 Net cash provided by (used for) investing activities (2,590) 17,400 Cash Flows from Financing Activities (2,590) 17,400 Cash Flows from Financing Activities (550) (770) Repayments of borrowings on term loan facilities (550) (770) Proceeds from borrowings on revolving credit facilities (130,670) (27,4600)					
Non-cash compensation expense 480 30 Net proceeds from (reductions in) sale of receivables and receivables securitization 3,830 (6,130) Increase in receivables 38,960 (2,630) Decrease in inventories 6,060 18,090 Decrease in inventories 270 1,660 Increase in accounts payable and accrued liabilities 7,910 1,180 Other, net 620 (990) Net cash provided by (used for) operating activities, net of acquisition impact (4,090) 18,000 Net proceeds from Investing Activities: 250 (3,280) Repayments from disposition of businesses and other assets 30 20,680 Net cash provided by (used for) investing activities (2,590) 17,400 Cash Flows from disposition of businesses and other assets 30 20,880 Net cash provided by (used for) investing activities (550) (770 Cash Flows from Einancing Activities (550) (770 Proceeds from disposition of businesses and other assets 30 20,880 Ret cash provided by (used for) investing activities (550) (770 </td <td></td> <td></td> <td>(380)</td> <td></td> <td></td>			(380)		
Net proceeds from (reductions in) sale of receivables and receivables securitization 3,830 (6,130) Increase in receivables (38,960) (2,630) Decrease in inventories 6,060 18,090 Decrease in prepaid expenses and other assets 270 1,660 Increase in accounts payable and accrued liabilities 7,910 1,180 Other, net 620 (990) Net cash provided by (used for) operating activities, net of acquisition impact (4,090) 1,800 Cash Flows from Investing Activities: 2(5,590) (3,280) Net proceeds from disposition of businesses and other assets 30 20,680 Net cash provided by (used for) investing activities (2,590) 17,400 Cash Flows from Hinancing Activities: (2,560) 17,400 Cash Flows from Financing Activities: (550) 17,400 Cash Flows from Financing Activities: (650) (770) Repayments of borrowings on term loan facilities (650) (770) Proceeds from borrowings on revolving credit facilities 134,940 272,900 Retirement of senior subordinated notes — (16	-		400		
Increase in receivables (38,960) (2,630) Decrease in inventories 6,060 18,090 Decrease in prepaid expenses and other assets 270 1,660 Increase in accounts payable and accrued liabilities 7,910 1,180 Other, net 620 (990) Net cash provided by (used for) operating activities, net of acquisition impact (4,090) 1,800 Cash Flows from Investing Activities: Cash Flows from disposition of businesses and other assets 30 20,680 Net cash provided by (used for) investing activities 30 20,680 Net cash provided by (used for) investing activities (2,590) 17,400 Cash Flows from disposition of businesses and other assets 30 20,680 Net cash provided by (used for) investing activities (2,590) 17,400 Cash Flows from Enancing Activities (650) (770) Repayments of borrowings on term loan facilities (130,670) (274,680) Repayments of borrowings on revolving credit facilities (130,670) (274,680) Retirement of senior subordinated notes					
Decrease in inventories 6,060 18,090 Decrease in prepaid expenses and other assets 270 1,660 Increase in accounts payable and accrued liabilities 7,910 1,180 Other, net 620 (990) Net cash provided by (used for) operating activities, net of acquisition impact (4,090) 1,800 Cash Flows from Investing Activities: 2,590 (3,280) Net proceeds from disposition of businesses and other assets 30 20,680 Net cash provided by (used for) investing activities (2,560) 17,400 Cash Flows from Financing Activities: (2,560) 17,400 Cash Flows from Financing Activities: (650) (770) Proceeds from borrowings on revolving credit facilities 134,940 227,900 Repayments of borrowings on revolving credit facilities (130,670) (274,680) Repayments of borrowings on revolving credit facilities (130,670) (274,680) Repayments of borrowings on revolving credit facilities (130,670) (274,680) Repayments of borrowings on revolving credit facilities (3,80) (16,020) Shares surrendered upon vesting of	-		,		
Decrease in prepaid expenses and other assets 270 1,660 Increase in accounts payable and accrued liabilities 7,910 1,180 Other, net 620 (990) Net cash provided by (used for) operating activities, net of acquisition impact (4,900) 1,800 Cash Flows from Investing Activities: 2,590 (3,280) Net proceeds from disposition of businesses and other assets 30 20,680 Net cash provided by (used for) investing activities (2,500) 17,400 Net rash provided by (used for) investing activities (5,500) 17,400 Cash Flows from Financing Activities: (5,500) 17,400 Repayments of borrowings on term loan facilities (650) (770) Proceeds from borrowings on revolving credit facilities 134,940 272,900 Repayments of borrowings on revolving credit facilities (130,670) (274,680) Retirement of senior subordinated notes — (16,020) Shares surrendered upon vesting of options and restricted stock awards to cover tax obligations (16) — Proceeds from exercise of stock options 3,800 (18,570) Cash an					
Increase in accounts payable and accrued liabilities 7,910 1,180 Other, net 620 (990) Net cash provided by (used for) operating activities, net of acquisition impact (4,090) 1,800 Cash Flows from Investing Activities: Capital expenditures (2,590) (3,280) Net proceeds from disposition of businesses and other assets 30 20,680 Net cash provided by (used for) investing activities (2,560) 17,400 Cash Flows from Financing Activities: (2,560) 17,400 Cash Flows from Financing Activities (650) (770) Proceeds from borrowings on term loan facilities (650) (770) Proceeds from borrowings on revolving credit facilities 134,940 272,900 Repayments of borrowings on revolving credit facilities (130,670) (274,680) Retirement of senior subordinated notes — (16,020) Shares surrendered upon vesting of options and restricted stock awards to cover tax obligations (160) — Proceeds from exercise of stock options 3,800 (18,570) Cash and Cash Equivalents: 280 — <td></td> <td></td> <td>,</td> <td></td> <td></td>			,		
Other, net 620 (990) Net cash provided by (used for) operating activities, net of acquisition impact (4,909) 1,800 Cash Flows from Investing Activities: Capital expenditures (2,590) (3,280) Net proceeds from disposition of businesses and other assets 30 20,680 Net cash provided by (used for) investing activities (2,500) 17,400 Cash Flows from Financing Activities: Repayments of borrowings on term loan facilities (650) (770) Proceeds from borrowings on revolving credit facilities 134,940 272,900 Repayments of borrowings on revolving credit facilities (130,670) (274,680) Repayments of borrowings on revolving credit facilities (130,670) (274,680) Repayments of borrowings on revolving credit facilities (130,670) (274,680) Retirement of senior subordinated notes — (160,020) Shares surrendered upon vesting of options and restricted stock awards to cover tax obligations (160) — Pexcess tax benefits from stock based compensation 3,800 (18,570) Cash and Cash Equivalents					,
Net cash provided by (used for) operating activities, net of acquisition impact (4,090) 1,800 Cash Flows from Investing Activities: (2,590) (3,280) Net proceeds from disposition of businesses and other assets 30 20,680 Net cash provided by (used for) investing activities (2,560) 17,400 Cash Flows from Financing Activities: 8 (650) (770) Proceeds from borrowings on term loan facilities (134,940) 272,900 Repayments of borrowings on revolving credit facilities (130,670) (274,680) Retirement of senior subordinated notes — (16,020) Shares surrendered upon vesting of options and restricted stock awards to cover tax obligations (160) — Proceeds from exercise of stock options 60 — Excess tax benefits from stock based compensation 280 — Net cash provided by (used for) financing activities 3,800 (18,570) Cash and Cash Equivalents: Increase (decrease) for the period (2,850) 630 At end of period 9,480 3,910 At end of period \$6,630 \$4,540 <th< td=""><td></td><td></td><td>,</td><td></td><td></td></th<>			,		
Cash Flows from Investing Activities: Capital expenditures (2,590) (3,280) Net proceeds from disposition of businesses and other assets 30 20,680 Net cash provided by (used for) investing activities (2,560) 17,400 Cash Flows from Financing Activities: Repayments of borrowings on term loan facilities (650) (770) Proceeds from borrowings on revolving credit facilities 134,940 272,900 Repayments of borrowings on revolving credit facilities (130,670) (274,680) Repayments of borrowings on revolving credit facilities (130,670) (274,680) Repayments of borrowings on revolving credit facilities (130,670) (274,680) Repayments of senior subordinated notes — (16,020) Shares surrendered upon vesting of options and restricted stock awards to cover tax obligations (160) — Proceeds from exercise of stock options 280 — Excess tax benefits from stock based compensation 280 — Net cash provided by (used for) financing activities (2,850) 630 At beginning of period (2,850) 630 <td>•</td> <td></td> <td></td> <td>_</td> <td>, ,</td>	•			_	, ,
Capital expenditures (2,590) (3,280) Net proceeds from disposition of businesses and other assets 30 20,680 Net cash provided by (used for) investing activities (2,560) 17,400 Cash Flows from Financing Activities: Repayments of borrowings on term loan facilities (650) (770) Proceeds from borrowings on revolving credit facilities 134,940 272,900 Repayments of borrowings on revolving credit facilities (130,670) (274,680) Retirement of senior subordinated notes — (16,020) Shares surrendered upon vesting of options and restricted stock awards to cover tax obligations (160) — Proceeds from exercise of stock options 60 — Excess tax benefits from stock based compensation 280 — Net cash provided by (used for) financing activities 3,800 (18,570) Cash and Cash Equivalents: Increase (decrease) for the period (2,850) 630 At beginning of period 9,480 3,910 At end of period \$6,630 \$4,540 Supplemental disclosure of cash flow information:			(4,090)	_	1,800
Net proceeds from disposition of businesses and other assets 30 20,680 Net cash provided by (used for) investing activities (2,560) 17,400 Cash Flows from Financing Activities: Repayments of borrowings on term loan facilities (650) (770) Proceeds from borrowings on revolving credit facilities 134,940 272,900 Repayments of borrowings on revolving credit facilities (130,670) (274,680) Retirement of senior subordinated notes — (16,020) Shares surrendered upon vesting of options and restricted stock awards to cover tax obligations (160) — Proceeds from exercise of stock options 60 — Excess tax benefits from stock based compensation 280 — Net cash provided by (used for) financing activities 3,800 (18,570) Cash and Cash Equivalents: Increase (decrease) for the period (2,850) 630 At beginning of period 9,480 3,910 At end of period 5,630 4,540 Supplemental disclosure of cash flow information: 5,250 4,770					
Net cash provided by (used for) investing activities (2,560) 17,400 Cash Flows from Financing Activities: Repayments of borrowings on term loan facilities (650) (770) Proceeds from borrowings on revolving credit facilities 134,940 272,900 Repayments of borrowings on revolving credit facilities (130,670) (274,680) Retirement of senior subordinated notes — (16,020) Shares surrendered upon vesting of options and restricted stock awards to cover tax obligations (160) — Proceeds from exercise of stock options — 60 — 6 Excess tax benefits from stock based compensation 280 — 6 Net cash provided by (used for) financing activities 3,800 (18,570) Cash and Cash Equivalents: Increase (decrease) for the period (2,850) 630 At beginning of period 9,480 3,910 At end of period 9,480 3,910 Supplemental disclosure of cash flow information: Cash paid for interest \$5,250 \$4,770					
Cash Flows from Financing Activities:Repayments of borrowings on term loan facilities(650)(770)Proceeds from borrowings on revolving credit facilities134,940272,900Repayments of borrowings on revolving credit facilities(130,670)(274,680)Retirement of senior subordinated notes—(16,020)Shares surrendered upon vesting of options and restricted stock awards to cover tax obligations(160)—Proceeds from exercise of stock options60—Excess tax benefits from stock based compensation280—Net cash provided by (used for) financing activities3,800(18,570)Cash and Cash Equivalents:Increase (decrease) for the period(2,850)630At beginning of period9,4803,910At end of period\$6,630\$4,540Supplemental disclosure of cash flow information:Cash paid for interest\$5,250\$4,770					20,680
Repayments of borrowings on term loan facilities(650)(770)Proceeds from borrowings on revolving credit facilities134,940272,900Repayments of borrowings on revolving credit facilities(130,670)(274,680)Retirement of senior subordinated notes—(16,020)Shares surrendered upon vesting of options and restricted stock awards to cover tax obligations(160)—Proceeds from exercise of stock options60—Excess tax benefits from stock based compensation280—Net cash provided by (used for) financing activities3,800(18,570)Cash and Cash Equivalents:(2,850)630At beginning of period9,4803,910At end of period\$ 6,630\$ 4,540Supplemental disclosure of cash flow information:\$ 5,250\$ 4,770	Net cash provided by (used for) investing activities		(2,560)		17,400
Proceeds from borrowings on revolving credit facilities 134,940 272,900 Repayments of borrowings on revolving credit facilities (130,670) (274,680) Retirement of senior subordinated notes — (16,020) Shares surrendered upon vesting of options and restricted stock awards to cover tax obligations (160) — Proceeds from exercise of stock options — Excess tax benefits from stock based compensation 280 — Excess tax benefits from stock based compensation 280 — Net cash provided by (used for) financing activities 3,800 (18,570) Cash and Cash Equivalents: Increase (decrease) for the period (2,850) 630 At beginning of period 9,480 3,910 At end of period \$6,630 \$4,540 Supplemental disclosure of cash flow information: Cash paid for interest \$5,250 \$4,770	Cash Flows from Financing Activities:				
Repayments of borrowings on revolving credit facilities(130,670)(274,680)Retirement of senior subordinated notes—(16,020)Shares surrendered upon vesting of options and restricted stock awards to cover tax obligations(160)—Proceeds from exercise of stock options60—Excess tax benefits from stock based compensation280—Net cash provided by (used for) financing activities3,800(18,570)Cash and Cash Equivalents:Increase (decrease) for the period(2,850)630At beginning of period9,4803,910At end of period\$6,630\$4,540Supplemental disclosure of cash flow information:Cash paid for interest\$5,250\$4,770	Repayments of borrowings on term loan facilities		(650)		(770)
Retirement of senior subordinated notes Shares surrendered upon vesting of options and restricted stock awards to cover tax obligations Proceeds from exercise of stock options Excess tax benefits from stock based compensation Net cash provided by (used for) financing activities Cash and Cash Equivalents: Increase (decrease) for the period At beginning of period At end of period Supplemental disclosure of cash flow information: Cash paid for interest (16,020) (16,020) (16,020) (16,020) (160) (20)	Proceeds from borrowings on revolving credit facilities		134,940		272,900
Shares surrendered upon vesting of options and restricted stock awards to cover tax obligations Proceeds from exercise of stock options Excess tax benefits from stock based compensation Net cash provided by (used for) financing activities Cash and Cash Equivalents: Increase (decrease) for the period At beginning of period At end of period Supplemental disclosure of cash flow information: Cash paid for interest Shares surrendered upon vesting of options and restricted stock awards to cover tax obligations (160) — (260) — (280) — (2850) 630 4,540 Supplemental disclosure of cash flow information: Cash paid for interest \$5,250 \$4,770	Repayments of borrowings on revolving credit facilities		(130,670)		(274,680)
Proceeds from exercise of stock options 60 — Excess tax benefits from stock based compensation 280 — Net cash provided by (used for) financing activities 3,800 (18,570) Cash and Cash Equivalents: Increase (decrease) for the period (2,850) 630 At beginning of period 9,480 3,910 At end of period \$6,630 \$4,540 Supplemental disclosure of cash flow information: Cash paid for interest \$5,250 \$4,770	Retirement of senior subordinated notes		_		(16,020)
Excess tax benefits from stock based compensation Net cash provided by (used for) financing activities Cash and Cash Equivalents: Increase (decrease) for the period At beginning of period At end of period Supplemental disclosure of cash flow information: Cash paid for interest 280 — (2,850) 630 (2,850) 630 4,540 Supplemental disclosure of cash flow information: \$ 5,250 \$ 4,770			(160)		_
Net cash provided by (used for) financing activities 3,800 (18,570) Cash and Cash Equivalents: Increase (decrease) for the period (2,850) 630 At beginning of period 9,480 3,910 At end of period \$6,630 \$4,540 Supplemental disclosure of cash flow information: Cash paid for interest \$5,250 \$4,770	Proceeds from exercise of stock options		60		
Cash and Cash Equivalents: Increase (decrease) for the period (2,850) 630 At beginning of period 9,480 3,910 At end of period \$ 6,630 \$ 4,540 Supplemental disclosure of cash flow information: \$ 5,250 \$ 4,770	Excess tax benefits from stock based compensation		280		
Increase (decrease) for the period (2,850) 630 At beginning of period 9,480 3,910 At end of period \$ 6,630 \$ 4,540 Supplemental disclosure of cash flow information: Cash paid for interest \$ 5,250 \$ 4,770	Net cash provided by (used for) financing activities		3,800		(18,570)
Increase (decrease) for the period (2,850) 630 At beginning of period 9,480 3,910 At end of period \$ 6,630 \$ 4,540 Supplemental disclosure of cash flow information: Cash paid for interest \$ 5,250 \$ 4,770	Cash and Cash Equivalents:	_		_	
At beginning of period 9,480 3,910 At end of period \$ 6,630 \$ 4,540 Supplemental disclosure of cash flow information: Cash paid for interest \$ 5,250 \$ 4,770			(2.850)		630
At end of period \$ 6,630 \$ 4,540 Supplemental disclosure of cash flow information: Cash paid for interest \$ 5,250 \$ 4,770					
Cash paid for interest \$ 5,250 \$ 4,770		\$		\$	
Cash paid for interest \$ 5,250 \$ 4,770	Supplemental disclosure of cash flow information:			_	
Cash paid for taxes \$ 1,250 \$ 2,440		\$	5,250	\$	4,770
	Cash paid for taxes	\$	1,250	\$	2,440

Consolidated Statement of Shareholders' Equity

Three Months Ended March 31, 2010

(Unaudited—dollars in thousands)

		nmon	Paid-in	-		mulated	Comp	mulated Other rehensive		
D 1 D4 0000	- 51	ock	Capita		_	eficit		come	Φ.	Total
Balances, December 31, 2009	\$	330	\$ 528,3	370	\$ (5	510,380)	\$	43,660	\$	61,980
Comprehensive income:										
Net income				—		5,430				5,430
Foreign currency translation		_		_				(2,890)		(2,890)
Amortization of unrealized loss on interest rate swaps										
(Note 9)		_		_		_		410		410
Total comprehensive income										2,950
Shares surrendered upon vesting of options and restricted										
stock awards to cover tax obligations			(1	(60		_		_		(160)
Stock option exercises and restricted stock vestings		10		50		_		_		60
Excess tax benefits from stock based compensation			2	280		_		_		280
Non-cash compensation expense		_	4	180		_		_		480
Balances, March 31, 2010	\$	340	\$ 529,0)20	\$ (5	504,950)	\$	41,180	\$	65,590
									_	

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

(unaudited)

1. Basis of Presentation

TriMas Corporation ("TriMas" or the "Company"), and its consolidated subsidiaries, is a global manufacturer and distributor of products for commercial, industrial and consumer markets. The Company is principally engaged in five reportable segments with diverse products and market channels: Packaging, Energy, Aerospace & Defense, Engineered Components and Cequent. See Note 11, "Segment Information," for further information on each of the Company's reportable segments.

The accompanying consolidated financial statements include the accounts of the Company and its subsidiaries and in the opinion of management, contain all adjustments, including adjustments of a normal and recurring nature, necessary for a fair presentation of financial position and results of operations. Results of operations for interim periods are not necessarily indicative of results for the full year. Certain prior year amounts have been reclassified to conform with the current year presentation. The accompanying consolidated financial statements and notes thereto should be read in conjunction with the Company's 2009 Annual Report on Form 10-K.

2. Discontinued Operations and Assets Held for Sale

During the fourth quarter of 2009, the Company committed to a plan to exit its medical device line of business which was part of the Engineered Components operating segment. In connection with the exit plan, the Company recorded impairment and other charges during the fourth quarter of 2009 to reduce the value of the net assets in this line of business to net realizable value. Management believes this estimate of net realizable value remains reasonable as of March 31, 2010.

During the fourth quarter of 2008, the Company entered into a binding agreement to sell certain assets within its specialty laminates, jacketings and insulation tapes line of business, which was part of the Packaging operating segment. The sale was completed in February 2009 for cash proceeds of approximately \$21.0 million. The Company's manufacturing facility is subject to a lease agreement expiring in 2024 that was not assumed by the purchaser of the business. During first quarter 2009, upon the cease use date of the facility, the Company recorded a pre-tax charge of approximately \$10.7 million for future lease obligations on the facility, net of estimated sublease recoveries.

During the fourth quarter of 2007, the Company committed to a plan to sell its property management line of business. The sale was completed in April 2010. Refer to Note 16, "Subsequent Events," for further disclosure regarding the sale.

The results of the aforementioned businesses are reported as discontinued operations for all periods presented. Results of discontinued operations are summarized as follows:

	March 31,			
	2010	2009		
	(dollars in thousands			
Net sales	\$ 530	\$ 9,120		
Loss from discontinued operations before income tax benefit	\$ (500)	\$ (13,500)		
Income tax benefit	180	5,200		
Loss from discontinued operations, net of income tax benefit	\$ (320)	\$ (8,300)		

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS (Continued)

(unaudited)

2. Discontinued Operations and Assets Held for Sale (Continued)

Assets and liabilities of the discontinued operations held for sale are summarized as follows:

	M	March 31, 2010		cember 31, 2009
		(dollars i	n thou	usands)
Receivables, net	\$	30	\$	200
Property and equipment, net		4,040		4,050
Total assets	\$	4,070	\$	4,250
Accounts payable	\$	40	\$	150
Accrued liabilities and other		1,000		920
Total liabilities	\$	1,040	\$	1,070

3. Mosinee Plant Closure

In March 2009, the Company announced plans to close its manufacturing facility in Mosinee, Wisconsin, moving production and distribution functions from Mosinee to lower-cost manufacturing facilities or to third-party sourcing partners. As of December 31, 2009, the Company completed the move and ceased operations in Mosinee. Upon the cease use date of the facility, the Company recorded a pre-tax charge within its Cequent segment of approximately \$5.3 million for future lease obligations on the facility net of estimated lease recoveries. Also during 2009, the Company recorded a pre-tax charge within its Cequent segment of approximately \$1.8 million, of which \$1.6 million was recorded in the three months ended March 31, 2009, primarily related to cash costs for severance benefits for approximately 160 employees to be involuntarily terminated as part of the closure. As of March 31, 2010, the Company had paid approximately \$1.6 million of severance benefits, with the remaining \$0.2 million expected to be paid by the end of 2010.

In addition, the Company recorded approximately \$2.6 million of accelerated depreciation expense during 2009 as a result of shortening the expected lives on certain machinery and equipment assets that the Company no longer utilizes following the closure, of which \$0.5 million was recorded in the first quarter of 2009.

4. Goodwill and Other Intangible Assets

Changes in the carrying amount of goodwill for the three months ended March 31, 2010 are summarized as follows:

	Packaging	Energy	Aerospace & Defense (dollars in	Engineered Components thousands)	Cequent	Total
Balance, December 31, 2009	\$ 115,460	\$ 39,740	\$ 41,130	\$	\$ —	\$ 196,330
Foreign currency translation	(2,410)	200				(2,210)
Balance, March 31, 2010	\$ 113,050	\$ 39,940	\$ 41,130	\$ —	\$ —	\$ 194,120

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS (Continued)

(unaudited)

4. Goodwill and Other Intangible Assets (Continued)

The gross carrying amounts and accumulated amortization of the Company's other intangibles as of March 31, 2010 and December 31, 2009 are summarized below. The Company amortizes these assets over periods ranging from 1 to 30 years.

As of March 31, 2010			As of December			, 2009	
	Gross Carrying Amount		Accumulated Amortization		Gross Carrying Amount		ccumulated mortization
			(dollars in t	hou	sands)		
\$	24,750	\$	(18,860)	\$	24,710	\$	(18,290)
	154,610		(63,300)		154,610		(61,250)
	179,360		(82,160)		179,320		(79,540)
	26,660		(22,630)		25,800		(22,060)
	42,200		(17,150)		42,120		(16,640)
	68,860		(39,780)		67,920		(38,700)
	35,130		_		35,080		_
\$	283,350	\$	(121,940)	\$	282,320	\$	(118,240)
		\$ 24,750 154,610 179,360 26,660 42,200 68,860 35,130	\$ 24,750 \$ 154,610 179,360 26,660 42,200 68,860 35,130	Gross Carrying Amount Accumulated Amortization (dollars in the control of the control	Gross Carrying Amount Accumulated Amortization (dollars in thousand tho	Gross Carrying Amount Accumulated Amortization (dollars in thousands) Gross Carrying Amount thousands) \$ 24,750 \$ (18,860) \$ 24,710 154,610 (63,300) 154,610 179,360 (82,160) 179,320 26,660 (22,630) 25,800 42,200 (17,150) 42,120 68,860 (39,780) 67,920 35,130 — 35,080	Gross Carrying Amount Accumulated Amortization (dollars in thousands) Gross Carrying Amount (dollars in thousands) Accumulated Amount (dollars

Amortization expense related to technology and other intangibles was approximately \$1.0 million for each of the three month periods ended March 31, 2010 and 2009. These amounts are included in cost of sales in the accompanying consolidated statement of operations. Amortization expense related to customer relationship intangibles was approximately \$2.6 million for each of the three month periods ended March 31, 2010 and 2009. These amounts are included in selling, general and administrative expenses in the accompanying consolidated statement of operations.

5. Receivables Facility

On December 29, 2009, the Company entered into a new three year accounts receivable facility through TSPC, Inc. ("TSPC"), a wholly-owned subsidiary, to sell trade accounts receivable of substantially all of the Company's domestic business operations. Under this facility, TSPC, from time to time, may sell an undivided fractional ownership interest in the pool of receivables up to approximately \$75.0 million to a third party multi-seller receivables funding company. The Company did not have any amounts outstanding under the facility as of March 31, 2010 or December 31, 2009, but had \$51.7 million and \$32.1 million, respectively, available but not utilized. The net amount financed under the facility is less than the face amount of accounts receivable by an amount that approximates the purchaser's financing costs. As of March 31, 2010, the cost of funds under this facility consisted of a 3-month London Interbank Offered Rates ("LIBOR")-based rate plus a usage fee of 3.25% and a fee on the unused portion of the facility of 0.75%. Aggregate costs incurred under this facility were \$0.3 million for the three months ended March 31, 2010 and are included in interest expense in the accompanying consolidated statement of operations. The facility expires on December 29, 2012.

The cost of funds fees incurred are determined by calculating the estimated present value of the receivables sold compared to their carrying amount. The estimated present value factor is based on

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS (Continued)

(unaudited)

5. Receivables Facility (Continued)

historical collection experience and a discount rate based on a 3 month LIBOR-based rate plus the usage fee discussed above and is computed in accordance with the terms of the securitization agreement. As of March 31 2010, the costs of funds under the facility was based on an average liquidation period of the portfolio of approximately 1.6 months and an average discount rate of 1.5%.

Through December 28, 2009, TriMas was party to a 364-day accounts receivable facility through TSPC. Under this facility, TSPC, from time to time, was able to sell an undivided fractional ownership interest in the pool of receivables up to approximately \$55.0 million to a third party multi-seller receivables funding company. The net proceeds of the sale of receivables were less than the face amount of the accounts receivable sold by an amount that approximated the purchaser's financing costs. As of March 31, 2009, the cost of funds under this facility consisted of a commercial paper-based rate plus the usage fee of 4.5% and a fee on the unused portion of the facility of 2.25%. In addition, the Company completed its annual renewal of this facility in February 2009, incurring approximately \$0.4 million. Aggregate costs incurred under this facility, including renewal costs, were \$0.7 million for the three months ended March 31, 2009. Such amounts are included in other expense, net in the accompanying consolidated statement of operations.

In addition, the Company from time to time may sell an undivided interest in accounts receivable under factoring arrangements at three of its European subsidiaries. As of March 31, 2010 and December 31, 2009, the Company's funding under these arrangements was approximately \$3.8 million and \$4.5 million, respectively. Sales of the European subsidiaries' accounts receivable were sold at a discount from face value of approximately 1.5% and 1.8%, at March 31, 2010 and 2009, respectively. Costs associated with these transactions were approximately \$0.1 million for each of the three month periods ended March 31, 2010 and 2009, respectively, and are included in other expense, net in the accompanying consolidated statement of operations.

6. Inventories

Inventories consist of the following:

	 Iarch 31, 2010 (dollars i	cember 31, 2009 isands)
Finished goods	\$ 90,420	\$ 95,420
Work in process	16,090	16,270
Raw materials	29,220	30,150
Total inventories	\$ 135,730	\$ 141,840

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS (Continued)

(unaudited)

7. Property and Equipment, Net

Property and equipment, net, consists of the following:

	M	March 31, 2010		ecember 31, 2009	
	(dollars in thousands)				
Land and land improvements	\$	2,140	\$	2,380	
Buildings		44,500		44,810	
Machinery and equipment		276,630		283,710	
		323,270		330,900	
Less: Accumulated depreciation		165,840		168,680	
Property and equipment, net	\$	157,430	\$	162,220	

Depreciation expense was approximately \$6.0 million and \$6.5 million for the three months ended March 31, 2010 and 2009, respectively, of which \$5.3 million and \$5.7 million, respectively, is included in cost of sales in the accompanying statement of operations, and \$0.7 million and \$0.8 million, respectively, is included in selling, general and administrative expenses in the accompanying statement of operations.

In the first quarter of 2009, in connection with the closure of the Mosinee facility (see Note 3), the Company recorded accelerated depreciation expense of approximately \$0.5 million, which is included in the \$5.7 million of depreciation expense recorded in cost of sales. This charge related to shortening the expected useful lives on certain machinery and equipment.

8. Long-term Debt

The Company's long-term debt consists of the following:

	March 31, 2010	December 31, 2009
	(dollars in	thousands)
U.S. bank debt	\$ 264,080	\$ 256,680
Non-U.S. bank debt and other	9,350	12,890
9 ³ /4% senior secured notes, due December 2017	245,090	244,980
	518,520	514,550
Less: Current maturities, long-term debt	12,720	16,190
Long-term debt	\$ 505,800	\$ 498,360

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS (Continued)

(unaudited)

8. Long-term Debt (Continued)

U.S. Bank Debt

The Company is a party to a credit facility consisting of a \$83.0 million revolving credit facility, a \$60.0 million deposit-linked supplemental revolving credit facility and a \$252.2 million term loan facility (collectively, the "Credit Facility"). Under the Credit Facility, the Company is also able to issue letters of credit, not to exceed \$65.0 million in aggregate, against its revolving credit facility commitments. At March 31, 2010 and December 31, 2009, the Company had letters of credit of approximately \$31.0 million and \$31.2 million, respectively, issued and outstanding. The weighted average interest rate on borrowings under the Credit Facility was 5.6% and 3.9% at March 31, 2010 and December 31, 2009, respectively.

At March 31, 2010 and December 31, 2009, the Company had \$13.2 million and \$5.1 million, respectively, outstanding under its supplemental revolving credit facility and had an additional \$98.8 million and \$101.7 million, respectively, potentially available after giving effect to approximately \$31.0 million and \$31.2 million, respectively, of letters of credit issued and outstanding. However, including availability under its accounts receivable facility and after consideration of leverage restrictions contained in the Credit Facility, the Company had \$150.5 million and \$114.3 million, respectively of capacity available to it for general corporate purposes under its revolving credit and accounts receivable facilities.

The bank debt is an obligation of the Company and its subsidiaries. Although the terms of the Credit Facility do not restrict the Company's subsidiaries from making distributions to it in respect of its $9^3/4\%$ senior secured notes, it does contain certain other limitations on the distribution of funds from TriMas Company LLC, the principal subsidiary, to the Company. The restricted net assets of the guarantor subsidiaries of approximately \$283.6 million and \$270.4 million at March 31, 2010 and December 31, 2009, respectively, are presented in the financial information in Note 17, "Supplemental Guarantor Condensed Consolidating Financial Information." The Credit Facility also contains various negative and affirmative covenants and other requirements affecting the Company and its subsidiaries. The Company was in compliance with its covenants at March 31, 2010.

The Company's term loan facility traded at approximately 98.0% and 95.5% of par value as of March 31, 2010 and December 31, 2009, respectively.

Non-U.S. Bank Debt

In the United Kingdom, the Company's subsidiary is party to a revolving debt agreement which is secured by a letter of credit under the Credit Facility. At March 31, 2010, the balance outstanding under this arrangement was \$0.7 million at an interest rate of 2.5%. At December 31, 2009, the balance outstanding under this agreement was approximately \$0.8 million at an interest rate of 2.5%.

In Australia, the Company's subsidiary is party to a debt agreement which matures December 31, 2010 and is secured by substantially all the assets of the subsidiary. At March 31, 2010, the balance outstanding under this agreement was \$8.3 million at an interest rate of 6.8%. At December 31, 2009, the balance outstanding under this agreement was approximately \$11.7 million at an interest rate of approximately 6.6%.

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS (Continued)

(unaudited)

8. Long-term Debt (Continued)

Notes

During the fourth quarter of 2009, the Company issued \$250.0 million principal amount of $9^3/4\%$ senior secured notes due 2017 ("Senior Notes") at a discount of \$5.0 million. The net proceeds of the offering, approximately \$239.7 million, together with \$29.3 million of cash on hand, were used to repurchase \$256.5 million principal amount of the Company's $9^7/8\%$ senior subordinated notes due 2012 ("Sub Notes"), for tender costs and expenses related thereto, and to pay fees and expenses related to the Senior Notes.

During the first quarter of 2009, the Company utilized approximately \$16.0 million of cash on hand to retire \$31.8 million of face value of the Sub Notes, resulting in a net gain of approximately \$15.3 million after considering non-cash debt extinguishment costs of \$0.5 million.

The Notes indenture contains negative and affirmative covenants and other requirements that are comparable to those contained in the Credit Facility. At March 31, 2010, the Company was in compliance with all such covenant requirements.

The Company's Senior Notes traded at approximately 103.0% and 98.5% of par value as of March 31, 2010 and December 31, 2009, respectively.

9. Derivative Instruments

The Company is party to interest rate swaps to fix the variable LIBOR-based portion on \$200.0 million notional amount of the Company's term loan facility. Upon inception, the interest rate swaps were designated as cash flow hedges; however, upon the Company's amendment and restatement of its credit facilities in the fourth quarter of 2009, the Company determined that these interest rate swaps were no longer effective due to the imposition of a LIBOR floor in the determination of the variable interest rate on the term loan facility. In February 2010, the Company amended the interest rate swaps to include a LIBOR floor similar to the term loan facility; however, the amended interest rate swaps have not be designated as hedging instruments. For the three months ended March 31, 2010, approximately \$0.4 million of unrealized loss from accumulated other comprehensive income incurred while the interest rate swaps were effective was amortized into earnings as interest expense. Over the next 12 months, the Company expects approximately \$1.2 million of unrealized loss in accumulated other comprehensive income incurred while the interest rate swaps were effective to be amortized into earnings as interest expense.

As of December 31, 2009, the Company held a foreign exchange forward contract with a notional value of 55.5 million Mexican pesos and a foreign exchange forward contract with a notional value of £6.5 million. These contracts expired during the first quarter of 2010 and were not designated as hedging instruments.

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS (Continued)

(unaudited)

9. Derivative Instruments (Continued)

As of March 31, 2010 and December 31, 2009, the fair value carrying amounts of the Company's derivative instruments not designated as hedging instruments are recorded as follows:

		Asset Derivatives			 Liability	Deri	vatives	
	Balance Sheet Caption	Marc 20	- /	De	cember 31, 2009 (dollars in	 arch 31, 2010 sands)	Do	ecember 31, 2009
Derivatives not designated as hedging					,	ĺ		
instruments								
Interest rate contracts	Accrued liabilities	\$	_	\$	_	\$ 2,510	\$	1,700
Interest rate contracts	Other long-term liabilities		_		_	580		660
Foreign currency forward contracts	Accrued liabilities		_		_	_		150
Total derivatives not designated as				_				
hedging instruments		\$	_	\$	_	\$ 3,090	\$	2,510

The effect of derivative instruments on the consolidated statement of operations for the three months ended March 31, 2010 and 2009, is summarized as follows:

	_	Amour Recogniz on De (Effectiv net	ed in <i>l</i> rivativ	AOCI res tion,		Location of Reclassified		Reclassi AOC Eari	t of Loss fed from I in to nings
			AOCI into Ea (Effective Po		2010	2009			
Derivatives designated as hedging instruments		(dollars in	thou:	sands)				(dollars in	thousands)
Interest rate contracts	\$	(1,250)	\$	(1,660)]	Interest exp	ense	\$ (410)	\$ (390)
						Amount of (Loss) Recog Earnings Derivati	nized in s on		
					_	Three mo ended Mar		(Loss) F	ion of Gain Recognized in
					_	2010 (dollars in the	2009 ousands)		nings on rivative
Derivatives not designated as hedging instrumen	nts								
Interest rate contracts					\$	(1,450)	\$ —	Interest	expense
Foreign exchange contracts					\$	50	\$ —	Other ex	xpense, net

Valuation of the interest rate swaps and foreign currency forward contracts are based on the income approach which uses observable inputs such as interest rate yield curves and forward currency exchange rates. Fair value measurements and the fair value hierarch level for the Company's assets and

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS (Continued)

(unaudited)

9. Derivative Instruments (Continued)

liabilities measured at fair value on a recurring basis as of March 31, 2010 and December 31, 2009, are shown below:

			March 31, 2010						
			Quoted Prices in						
			Active Markets		Significant				
			for Identical	Significant Other	Unobservable				
		Asset /	Assets	Observable Inputs	Inputs				
Description	Frequency	(Liability)	(Level 1)	(Level 2)	(Level 3)				
			(dollars	in thousands)					
Interest rate swaps	Recurring	\$ (3,090)	\$	\$ (3,090)	\$ —				

		December 31, 2009						
			Quoted Prices in					
			Active Markets	CiiCi Oul	Significant			
<u>Description</u>	Frequency	Asset / (Liability)	for Identical Assets (Level 1)	Significant Other Observable Inputs (Level 2)	Unobservable Inputs (Level 3)			
Interest vote or rope	Dogumina	¢ (2.260)	,	in thousands)	φ			
Interest rate swaps	Recurring	\$ (2,360)	5 —	\$ (2,360)	5 —			
Foreign currency forward contracts	Recurring	(150)	_	(150)	_			
		\$ (2,510)	\$ —	\$ (2,510)	\$ —			

10. Commitments and Contingencies

A civil suit was filed in the United States District Court for the Central District of California in December 1988 by the United States of America and the State of California against more than 180 defendants, including TriMas, for alleged release into the environment of hazardous substances disposed of at the Operating Industries, Inc. site in California. This site served for many years as a depository for municipal and industrial waste. The plaintiffs have requested, among other things, that the defendants clean up the contamination at that site. Consent decrees have been entered into by the plaintiffs and a group of the defendants, including TriMas, providing that the consenting parties perform certain remedial work at the site and reimburse the plaintiffs for certain past costs incurred by the plaintiffs at the site. The Company estimates that its share of the clean-up costs will not exceed \$500,000, for which the Company has insurance proceeds. Plaintiffs had sought other relief such as damages arising out of claims for negligence, trespass, public and private nuisance, and other causes of action, but the consent decree governs the remedy. Based upon the Company's present knowledge and subject to future legal and factual developments, the Company does not believe that this matter will have a material adverse effect on its financial position, results of operations or cash flows.

As of March 31, 2010, the Company was a party to approximately 989 pending cases involving an aggregate of approximately 8,044 claimants alleging personal injury from exposure to asbestos containing materials formerly used in gaskets (both encapsulated and otherwise) manufactured or distributed by certain of the Company's subsidiaries for use primarily in the petrochemical refining and exploration industries. The following chart summarizes the number of claimants, number of claims filed, number of claims dismissed, number of claims settled, the average settlement amount per claim

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS (Continued)

(unaudited)

10. Commitments and Contingencies (Continued)

and the total defense costs, exclusive of amounts reimbursed under the Company's primary insurance, at the applicable date and for the applicable periods:

	Claims pending at beginning of period	Claims filed during period	Claims dismissed during period	Claims settled during period	settl amoi claim	erage lement unt per during eriod	otal defense costs during period
Fiscal year ended December 31, 2009	7,524	586	254	40	\$	4,644	\$ 2,652,000
Three months ended March 31, 2010	7,816	359	118	13	\$	5,673	\$ 684,000

In addition, the Company acquired various companies to distribute its products that had distributed gaskets of other manufacturers prior to acquisition. The Company believes that many of its pending cases relate to locations at which none of its gaskets were distributed or used.

The Company may be subjected to significant additional asbestos-related claims in the future, the cost of settling cases in which product identification can be made may increase, and the Company may be subjected to further claims in respect of the former activities of its acquired gasket distributors. The Company is unable to make a meaningful statement concerning the monetary claims made in the asbestos cases given that, among other things, claims may be initially made in some jurisdictions without specifying the amount sought or by simply stating the requisite or maximum permissible monetary relief, and may be amended to alter the amount sought. The large majority of claims do not specify the amount sought. Of the 8,044 claims pending at March 31, 2010, 82 set forth specific amounts of damages (other than those stating the statutory minimum or maximum). 60 of the 82 claims sought between \$1.0 million and \$5.0 million and \$5.0 million in total damages (which includes compensatory and punitive damages) and 4 sought over \$10.0 million (which includes compensatory and punitive damages). Solely with respect to compensatory damages, 62 of the 82 claims sought between \$250,000 and \$600,000, 16 sought between \$1.0 million and \$5.0 million and 4 sought over \$5.0 million. Solely with respect to punitive damages, 60 of the 82 claims sought between \$1.0 million and \$2.5 million, 17 sought between \$2.5 million and \$5.0 million and 5 sought over \$5.0 million. In addition, relatively few of the claims have reached the discovery stage and even fewer claims have gone past the discovery stage.

Total settlement costs (exclusive of defense costs) for all such cases, some of which were filed over 20 years ago, have been approximately \$5.5 million. All relief sought in the asbestos cases is monetary in nature. To date, approximately 50% of the Company's costs related to settlement and defense of asbestos litigation have been covered by its primary insurance. Effective February 14, 2006, the Company entered into a coverage-in-place agreement with its first level excess carriers regarding the coverage to be provided to the Company for asbestos-related claims when the primary insurance is exhausted. The coverage-in-place agreement makes asbestos defense costs and indemnity coverage available to the Company that might otherwise be disputed by the carriers and provides a methodology for the administration of such expenses. Nonetheless, there may be a period prior to the commencement of coverage under this agreement and following exhaustion of the Company's primary

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS (Continued)

(unaudited)

10. Commitments and Contingencies (Continued)

insurance coverage during which the Company would be solely responsible for defense costs and indemnity payments, the duration of which would be subject to the scope of damage awards and settlements paid.

Based on the settlements made to date and the number of claims dismissed or withdrawn for lack of product identification, the Company believes that the relief sought (when specified) does not bear a reasonable relationship to its potential liability. Based upon the Company's experience to date and other available information (including the availability of excess insurance), the Company does not believe that these cases will have a material adverse effect on its financial position and results of operations or cash flows.

The Company is subject to other claims and litigation in the ordinary course of business, but does not believe that any such claim or litigation will have a material adverse effect on its financial position and results of operations or cash flows.

11. Segment Information

TriMas' reportable operating segments are business units that provide unique products and services. Each operating segment is independently managed, requires different technology and marketing strategies and has separate financial information evaluated regularly by the Company's chief operating decision maker in determining resource allocation and assessing performance. TriMas groups its operating segments into five reportable segments, described below. Within these operating segments, there are no individual products or product families for which reported revenues accounted for more than 10% of the Company's consolidated revenues.

Packaging—Steel and plastic closure caps, drum enclosures, rings and levers, and dispensing systems for industrial and consumer markets.

Energy—Natural gas engines, compressors, gas production equipment and chemical pumps engineered at well sites for the oil and gas industry as well as metallic and non-metallic industrial sealant products and fasteners for the petroleum refining, petrochemical and other industrial markets.

Aerospace & Defense—Highly engineered specialty fasteners and screws for the commercial and military aerospace industries and military munitions components for the defense industry.

Engineered Components—High-pressure and low-pressure cylinders for the transportation, storage and dispensing of compressed gases, specialty fittings for the automotive industry, precision cutting instruments for the medical industry and specialty precision tools such as center drills, cutters, end mills and countersinks for the industrial metal-working market.

Cequent—Custom-engineered towing, trailering and electrical products including trailer couplers, winches, jacks, trailer brakes and brake control solutions, lighting accessories and roof racks for the recreational vehicle, agricultural/utility, marine, automotive and commercial trailer markets, functional vehicle accessories and cargo management solutions including vehicle hitches and receivers, sway controls, weight distribution and fifth- wheel hitches, hitch-mounted accessories, and other accessory components.

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS (Continued)

(unaudited)

11. Segment Information (Continued)

The Company's management uses Adjusted Earnings Before Interest, Taxes, Depreciation and Amortization ("Adjusted EBITDA") as a primary indicator of financial operating performance and as a measure of cash generating capability. Adjusted EBITDA is defined as net income (loss) before cumulative effect of accounting change and before interest, taxes, depreciation, amortization, debt extinguishment costs, non-cash asset and goodwill impairment charges and write-offs and non-cash losses on sale-leaseback of property and equipment.

Segment activity is as follows:

	Three months ended March 31,				
	2010 2009				
	(dollars in thousands)				
Net Sales					
Packaging	\$	43,600	\$	30,250	
Energy		43,890		40,270	
Aerospace & Defense		17,080		22,200	
Engineered Components		18,910		18,550	
Cequent		96,580		90,450	
Total	\$	220,060	\$	201,720	
Operating Profit					
Packaging	\$	11,860	\$	5,400	
Energy		5,180		3,520	
Aerospace & Defense		3,860		6,810	
Engineered Components		1,810		380	
Cequent		8,120		(3,350)	
Corporate expenses		(5,780)		(7,560)	
Total	\$	25,050	\$	5,200	
Adjusted EBITDA					
Packaging	\$	14,920	\$	8,640	
Energy		5,900		4,280	
Aerospace & Defense		4,520		7,410	
Engineered Components		2,570		1,120	
Cequent		12,120		1,340	
Corporate (expenses) income		(5,900)		7,630	
Subtotal from continuing operations		34,130		30,420	
Discontinued operations		(330)		(11,790)	
Total company	\$	33,800	\$	18,630	

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS (Continued)

(unaudited)

11. Segment Information (Continued)

The following is a reconciliation of the Company's net income (loss) to Adjusted EBITDA:

	Three months ended March 31,			
	_	2010 (dollars in	thou	2009 isands)
Net income (loss)	\$	5,430	\$	(3,680)
Income tax expense (benefit)		4,470		(2,490)
Interest expense		14,290		12,530
Debt extinguishment costs				510
Depreciation and amortization		9,610		11,760
Adjusted EBITDA, total company	\$	33,800	\$	18,630
Adjusted EBITDA, discontinued operations		(330)		(11,790)
Adjusted EBITDA, continuing operations	\$	34,130	\$	30,420

12. Equity Awards

The Company maintains two long-term equity incentive plans, the TriMas Corporation 2006 Long Term Equity Incentive Plan (the "2006 Plan") and the 2002 Long Term Equity Incentive Plan (the "2002 Plan"). See below for details of awards by plan.

2006 Plan

Information related to stock options at March 31, 2010 is as follows:

	Number of Stock Options	Weighted Average Option Price	Average Remaining Contractual Life (Years) In	Aggregate ntrinsic Value
Outstanding at January 1, 2010	554,000	\$ 1.14		
Granted		_		
Exercised	(10,666)	1.01		
Cancelled	_	_		
Outstanding at March 31, 2010	543,334	\$ 1.15	8.9 \$	2,903,470

As of March 31, 2010, 173,998 stock options were exercisable under the 2006 Plan. In addition, the fair value of options which vested during the three month periods ended March 31, 2010 and 2009 was \$0.1 million and \$0, respectively. As of March 31, 2010, there was approximately \$0.1 million of unrecognized compensation cost related to stock options that is expected to be recorded over a weighted-average period of 1.5 years.

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS (Continued)

(unaudited)

12. Equity Awards (Continued)

Information related to restricted shares at March 31, 2010 is as follows:

	Number of Unvested Restricted Shares	Weighted Ave Grant Date I Value		Average Remaining Contractual Life (Years)	gregate nsic Value
Outstanding at January 1, 2010	251,937	\$	5.99		
Granted			_		
Vested	(131,810)		5.20		
Cancelled	(1,770)		8.48		
Outstanding at March 31, 2010	118,357	\$	6.84	0.9	\$ 768,140

As of March 31, 2010, there was approximately \$0.3 million of unrecognized compensation cost related to unvested restricted shares that is expected to be recorded over a weighted-average period of 0.9 years.

The Company recognized approximately \$0.3 million and \$0.02 million of stock-based compensation expense related to the 2006 Plan during the three months ended March 31, 2010 and 2009, respectively. The stock-based compensation expense is included in selling, general and administrative expenses in the accompanying statement of operations.

2002 Plan

During the first quarter of 2010, the Company granted 97,870 stock options to certain key employees, each of which may be used to purchase one share of the Company's common stock. These stock options have a ten year life, vest ratably over three years from date of grant, have an exercise price of \$6.09 and had a weighted-average fair value at grant date of \$2.60. The fair value of these options at the grant date was estimated using the Black-Scholes option pricing model using the following weighted-average assumptions: expected life of 6 years, risk-free interest rate of 2.7% and expected volatility of 40%.

Information related to stock options at March 31, 2010 is as follows:

Number of			Average Remaining Contractual Life	Aggregate Intrinsic Value
		13.45	(Itals)	Intrinsic value
97,870		6.09		
(47,455)		1.01		
(28,125)		1.01		
1,307,634	\$	13.62	6.2	\$ 2,424,870
	Options 1,285,344 97,870 (47,455) (28,125)	Options Option 1,285,344 \$ 97,870 (47,455) (28,125)	Options Option Price 1,285,344 \$ 13.45 97,870 6.09 (47,455) 1.01 (28,125) 1.01	Number of Options Weighted Average Option Price Remaining Contractual Life (Years) 1,285,344 \$ 13.45 97,870 6.09 (47,455) 1.01 (28,125) 1.01

As of March 31, 2010, 844,164 stock options were exercisable under the 2002 Plan. In addition, the fair value of options which vested during each of the three month periods ended March 31, 2010 and 2009 was \$0.1 million. As of March 31, 2010, there was approximately \$0.3 million of unrecognized

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS (Continued)

(unaudited)

12. Equity Awards (Continued)

compensation cost related to stock options that is expected to be recorded over a weighted-average period of 1.4 years.

During the first quarter of 2010, the Company granted 78,090 restricted shares of common stock to certain employees. These restricted shares are subject only to a service condition, vesting ratably over three years so long as the employee remains with the Company.

Information related to restricted shares at March 31, 2010 is as follows:

	Number of Unvested Restricted Shares	Weighted Average Grant Date Fair Value	Average Remaining Contractual Life (Years)	Aggregate Intrinsic Value
Outstanding at January 1, 2010	126,950	\$ 5.20		
Granted	78,090	6.09		
Vested	(82,960)	5.20		
Cancelled	(190)	5.20		
Outstanding at March 31, 2010	121,890	\$ 5.77	2.2	\$ 791,070

As of March 31, 2010, there was approximately \$0.6 million of unrecognized compensation cost related to unvested restricted shares that is expected to be recorded over a weighted-average period of 1.7 years.

The Company recognized approximately \$0.2 million and \$0.02 million of stock-based compensation expense related to the 2002 Plan during the three months ended March 31, 2010 and 2009, respectively. The stock-based compensation expense is included in selling, general and administrative expenses in the accompanying statement of operations.

13. Earnings per Share

Net earnings are divided by the weighted average number of shares outstanding during the year to calculate basic earnings per share. Diluted earnings (loss) per share are calculated to give effect to stock options and other stock-based awards. The calculation of diluted earnings per share included 217,083 and 28,024 restricted shares for the three months ended March 31, 2010 and 2009, respectively. The calculation of diluted earnings (loss) per share also included 527,260 options to purchase shares of common stock for the three months ended March 31, 2010. Options to purchase 2,050,194 shares of common stock were outstanding at March 31, 2009, but were not included in the calculation of diluted earnings (loss) per share because to do so would have been anti-dilutive.

14. Defined Benefit Plans

Net periodic pension and postretirement benefit costs for the Company's defined benefit pension plans and postretirement benefit plans cover foreign employees, union hourly employees and certain

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS (Continued)

(unaudited)

14. Defined Benefit Plans (Continued)

salaried employees. The components of net periodic pension and postretirement benefit costs for the three months ended March 31, 2010 and 2009 are as follows:

tirement
nefits
Months
nded
rch 31,
2009
lars in
ısands)
\$ —
30
_
(70)
(10)
\$ (50)
e

The Company contributed approximately \$0.4 million to its defined benefit pension plans during the three months ended March 31, 2010. The Company expects to contribute approximately \$1.9 million to its defined benefit pension plans for the full year 2010.

15. New Accounting Pronouncements

In June 2009, the Financial Accounting Standards Board ("FASB") issued amended guidance related to the transfer of financial assets. The new guidance, listed under Accounting Standards Codification ("ASC") Topic 860, "Transfers and Servicing," requires more information about transfers of financial assets, including securitization transactions, and where companies have continuing exposure to the risk related to transferred financial assets. ASC Topic 860 eliminates the concept of a qualifying special purpose entity, changes the requirements for derecognizing financial assets and requires additional disclosure. Under the Company's current domestic receivables facility, an undivided fractional ownership interest in a pool of receivables is sold to a third party with recourse. This guidance, effective January 1, 2010, eliminated the off-balance sheet treatment for such facilities. Therefore, upon adoption of this guidance, amounts outstanding under the facility are classified as receivables and debt in the Company's balance sheet, and related costs are classified as interest expense rather than other expense, net. As of March 31, 2010 and December 31, 2009, there were no amounts outstanding under the Company's receivables facility.

In June 2009, the FASB issued amended guidance related to consolidation of variable interest entities. The new guidance, listed under ASC Topic 810, "Consolidation," changes how a reporting entity determines when a variable interest entity should be consolidated. It also requires additional disclosures about its involvement with variable interest entities and any significant changes in risk exposure due to that involvement. There was no significant impact to the Company's consolidated financial statements as a result of the Company's adoption of this guidance during the first quarter of 2010.

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS (Continued)

(unaudited)

16. Subsequent Events

On April 6, 2010, the Company completed the sale of its property management line of business for cash proceeds of approximately \$13 million. In connection with this sale, the Company expects to recognize a pre-tax gain on sale of approximately \$10 million. The April 2010 gain on sale and prior period results of operations of the property management line of business will be and/or are included in discontinued operations in this Form 10-Q.

On April 30, 2010, the Company's Norris Cylinder subsidiary entered into an asset purchase agreement (the "Asset Purchase Agreement") with Taylor-Wharton International, LLC ("TWI") and its subsidiary, TW Cylinders LLC, to purchase certain assets related to TWI's high and low-pressure cylinder business for the purchase price of \$11 million, payable in cash at closing. The purchase price is subject to a net working capital adjustment, if any, to be determined at closing. The Asset Purchase Agreement contains customary representations, warranties, covenants and indemnities. As TWI and certain of its affiliates had filed bankruptcy petitions and motions for voluntary Chapter 11 reorganization under the United States Bankruptcy Code, the Asset Purchase Agreement is subject to the approval of the United States Bankruptcy Court for the District of Delaware (the "Bankruptcy Court"). Should the Asset Purchase Agreement be approved by the Bankruptcy Court, the Company plans to complete the acquisition pursuant to Section 363 of the U.S. Bankruptcy Code. The foregoing summary of the Asset Purchase Agreement, a copy of which is filed as an exhibit to this Report on Form 10-O.

17. Supplemental Guarantor Condensed Consolidating Financial Information

Under an indenture dated December 29, 2009, TriMas Corporation, the parent company ("Parent"), issued 9³/₄% senior secured notes due 2017 in a total principal amount of \$250.0 million (face value). The net proceeds of the offering were used, together with other available cash, to repurchase the Company's outstanding 9⁷/₈% senior subordinated notes due 2012 pursuant to a cash tender offer. The outstanding Senior Notes are guaranteed by substantially all of the Company's domestic subsidiaries ("Guarantor Subsidiaries"). All of the Guarantor Subsidiaries are 100% owned by the Parent and their guarantee is full, unconditional, joint and several. The Company's non-domestic subsidiaries and TSPC, Inc. have not guaranteed the Senior Notes ("Non-Guarantor Subsidiaries"). The Guarantor Subsidiaries have also guaranteed amounts outstanding under the Company's Credit Facility.

The accompanying supplemental guarantor condensed, consolidating financial information is presented using the equity method of accounting for all periods presented. Under this method, investments in subsidiaries are recorded at cost and adjusted for the Company's share in the subsidiaries' cumulative results of operations, capital contributions and distributions and other changes in equity. Elimination entries relate primarily to the elimination of investments in subsidiaries and associated intercompany balances and transactions.

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS (Continued)

(unaudited)

17. Supplemental Guarantor Condensed Consolidating Financial Information (Continued)

Supplemental Guarantor Condensed Financial Statements Consolidating Balance Sheet (dollars in thousands)

	March 31, 2010							
	Parent	Guarantor	Non- Guarantor	Eliminations	Consolidated Total			
Assets								
Current assets:								
Cash and cash equivalents	\$ —	\$ 140	\$ 6,490	\$ —	\$ 6,630			
Trade receivables, net	_	108,200	20,450	_	128,650			
Receivables, intercompany	_	_	3,510	(3,510)	_			
Inventories		112,130	23,600	_	135,730			
Deferred income taxes	5,400	23,430	890	(5,400)	24,320			
Prepaid expenses and other current assets		4,960	1,460	_	6,420			
Assets of discontinued operations held for sale		4,070			4,070			
Total current assets	5,400	252,930	56,400	(8,910)	305,820			
Investments in subsidiaries	283,560	107,050	_	(390,610)	_			
Property and equipment, net	_	113,160	44,270	_	157,430			
Goodwill	_	148,220	45,900	_	194,120			
Intangibles and other assets	27,940	172,610	5,630	(21,600)	184,580			
Total assets	\$ 316,900	\$ 793,970	\$ 152,200	\$ (421,120)	\$ 841,950			
Liabilities and Shareholders' Equity								
Current liabilities:								
Current maturities, long-term debt	\$ —	\$ 3,730		\$ —	\$ 12,720			
Accounts payable, trade	_	82,750	20,400	_	103,150			
Accounts payable, intercompany		3,510		(3,510)	_			
Accrued liabilities	6,220	48,610	8,840	_	63,670			
Liabilities of discontinued operations		1,040			1,040			
Total current liabilities	6,220	139,640	38,230	(3,510)	180,580			
Long-term debt	245,090	260,710	_	_	505,800			
Deferred income taxes	_	66,340	4,450	(27,000)	43,790			
Other long-term liabilities	_	43,720	2,470	_	46,190			
Total liabilities	251,310	510,410	45,150	(30,510)	776,360			
Total shareholders' equity	65,590	283,560	107,050	(390,610)	65,590			
Total liabilities and shareholders' equity	\$ 316,900	\$ 793,970	\$ 152,200	\$ (421,120)	\$ 841,950			

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS (Continued)

(unaudited)

17. Supplemental Guarantor Condensed Consolidating Financial Information (Continued)

Supplemental Guarantor Condensed Financial Statements Consolidating Balance Sheet (dollars in thousands)

	December 31, 2009									
		Parent	(Guarantor	Non- Guarantor		E	liminations	C	onsolidated Total
Assets		raiciii		Juai antoi		Juai aiitui	_15	illilliations_		Iviai
Current assets:										
Cash and cash equivalents	\$	_	\$	300	\$	9,180	\$	_	\$	9,480
Trade receivables, net		_		76,720		16,660		_		93,380
Receivables, intercompany		_		_		3,550		(3,550)		_
Inventories		_		117,850		23,990				141,840
Deferred income taxes		5,400		23,450		870		(5,400)		24,320
Prepaid expenses and other current assets		80		4,820		1,600				6,500
Assets of discontinued operations held for sale		_		4,250		_		_		4,250
Total current assets	_	5,480		227,390		55,850		(8,950)		279,770
Investments in subsidiaries		270,370		107,170		_		(377,540)		_
Property and equipment, net		_		115,380		46,840		_		162,220
Goodwill		_		148,220		48,110		_		196,330
Intangibles and other assets		31,240		175,190		5,720		(24,690)		187,460
Total assets	\$	307,090	\$	773,350	\$	156,520	\$	(411,180)	\$	825,780
Liabilities and Shareholders' Equity			_		_					
Current liabilities:										
Current maturities, long-term debt	\$	_	\$	3,670	\$	12,520	\$	_	\$	16,190
Accounts payable, trade		_		73,980		18,860		_		92,840
Accounts payable, intercompany		_		3,550		_		(3,550)		
Accrued liabilities		130		56,000		9,620		_		65,750
Liabilities of discontinued operations		_		1,070		_		_		1,070
Total current liabilities		130	_	138,270		41,000		(3,550)		175,850
Long-term debt		244,980		253,380		_		_		498,360
Deferred income taxes		_		66,920		5,760		(30,090)		42,590
Other long-term liabilities		_		44,410		2,590		_		47,000
Total liabilities		245,110		502,980		49,350		(33,640)		763,800
Total shareholders' equity	_	61,980		270,370		107,170		(377,540)		61,980
Total liabilities and shareholders' equity	\$	307,090	\$	773,350	\$	156,520	\$	(411,180)	\$	825,780
	_		_							

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS (Continued)

(unaudited)

17. Supplemental Guarantor Condensed Consolidating Financial Information (Continued)

Supplemental Guarantor Condensed Financial Statements Consolidating Statement of Operations (dollars in thousands)

	Three Months Ended March 31, 2010					
	Parent	Guarantor	Non-Guarantor	Eliminations	Total	
Net sales	\$ —	\$ 167,840	\$ 59,880	\$ (7,660)	\$ 220,060	
Cost of sales	_	(119,900)	(44,760)	7,660	(157,000)	
Gross profit		47,940	15,120	_	63,060	
Selling, general and administrative expenses	_	(32,220)	(5,480)		(37,700)	
Loss on dispositions of property and equipment	_	(70)	(240)	_	(310)	
Operating profit		15,650	9,400		25,050	
Other expense, net:						
Interest expense	(6,480)	(7,010)	(650)		(14,140)	
Other, net	_	(180)	(330)	_	(510)	
Income (loss) before income tax (expense) benefit and						
equity in net income of subsidiaries	(6,480)	8,460	8,420		10,400	
Income tax (expense) benefit	2,270	(4,580)	(2,340)	_	(4,650)	
Equity in net income of subsidiaries	9,640	6,080	_	(15,720)	_	
Income from continuing operations	5,430	9,960	6,080	(15,720)	5,750	
Loss from discontinued operations	_	(320)	_	_	(320)	
Net income	\$ 5,430	\$ 9,640	\$ 6,080	\$ (15,720)	\$ 5,430	

	Three Months Ended March 31, 2009						
	Parent	Guarantor	Non-Guarantor	Eliminations	Total		
Net sales	\$ —	\$ 167,520	\$ 39,550	\$ (5,350)	\$ 201,720		
Cost of sales	_	(128,330)	(32,280)	5,350	(155,260)		
Gross profit		39,190	7,270		46,460		
Selling, general and administrative expenses		(36,540)	(4,760)		(41,300)		
Gain on dispositions of property and equipment	_	30	10	_	40		
Operating profit		2,680	2,520		5,200		
Other income (expense), net:							
Interest expense	(8,340)	(3,900)	(240)		(12,480)		
Gain on extinguishment of debt	15,310	_	_	_	15,310		
Other, net	_	(150)	(550)	_	(700)		
Income (loss) before income tax (expense) benefit and							
equity in net income (loss) of subsidiaries	6,970	(1,370)	1,730	_	7,330		
Income tax (expense) benefit	(2,610)	530	(630)		(2,710)		
Equity in net income (loss) of subsidiaries	(8,040)	1,100	_	6,940	_		
Income (loss) from continuing operations	(3,680)	260	1,100	6,940	4,620		
Loss from discontinued operations	_	(8,300)	_	_	(8,300)		
Net income (loss)	\$ (3,680)	\$ (8,040)	\$ 1,100	\$ 6,940	\$ (3,680)		

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS (Continued)

(unaudited)

17. Supplemental Guarantor Condensed Consolidating Financial Information (Continued)

Supplemental Guarantor Condensed Financial Statements Consolidating Statement of Cash Flows (dollars in thousands)

	Three Months Ended March 31, 2010									
	Pa	rent	Gua	arantor	Non-G	iarantor	Eliminati	ons		Total
Cash Flows from Operating Activities:										
Net cash provided by (used for) operating										
activities	\$ (1,100)	\$	(8,990)	\$	6,000	\$	_	\$	(4,090)
Cash Flows from Investing Activities:										
Capital expenditures		_		(2,130)		(460)		_		(2,590)
Net proceeds from disposition of businesses and										
other assets		_		30		_		_		30
Net cash used for investing activities	-			(2,100)		(460)		_		(2,560)
Cash Flows from Financing Activities:										
Repayments of borrowings on term loan facilities		_		(650)		_		_		(650)
Proceeds from borrowings on revolving credit										
facilities		_	1	133,450		1,490		_		134,940
Repayments of borrowings on revolving credit										
facilities		_	(1	125,400)		(5,270)		_	((130,670)
Shares surrended upon vesting of options and										
restricted stock awards to cover tax obligations		(160)		_		_		_		(160)
Proceeds from exercise of stock options		60		_		_		_		60
Excess tax benefits from stock based compensation		_		280		_		_		280
Intercompany transfers (to) from subsidiaries		1,200		3,250		(4,450)		_		_
Net cash provided by (used for) financing										
activities		1,100		10,930		(8,230)		_		3,800
Cash and Cash Equivalents:					-					
Decrease for the period		_		(160)		(2,690)		_		(2,850)
At beginning of period		_		300		9,180		_		9,480
At end of period	\$		\$	140	\$	6,490	\$		\$	6,630

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS (Continued)

(unaudited)

17. Supplemental Guarantor Condensed Consolidating Financial Information (Continued)

Supplemental Guarantor Condensed Financial Statements Consolidating Statement of Cash Flows (dollars in thousands)

	Three Months Ended March 31, 2009					
	Parent	Guarantor	Non-Guarantor	Eliminations	Total	
Cash Flows from Operating Activities:						
Net cash provided by (used for) operating activities	\$ —	\$ (1,710)	\$ 3,510	\$ —	\$ 1,800	
Cash Flows from Investing Activities:						
Capital expenditures		(2,930)	(350)	_	(3,280)	
Net proceeds from disposition of businesses and						
other assets	_	20,640	40	_	20,680	
Net cash provided by (used for) investing						
activities	_	17,710	(310)	_	17,400	
Cash Flows from Financing Activities:						
Repayments of borrowings on senior credit						
facilities	_	(650)	(120)	_	(770)	
Proceeds from borrowings on revolving credit						
facilities	_	272,900	_	_	272,900	
Repayments of borrowings on revolving credit						
facilities		(273,800)	(880)	_	(274,680)	
Retirement of senior subordinated notes	(16,020)	_	_	_	(16,020)	
Intercompany transfers (to) from subsidiaries	16,020	(14,440)	(1,580)	_		
Net cash used for financing activities		(15,990)	(2,580)	_	(18,570)	
		<u>-</u>				
Cash and Cash Equivalents:						
Increase for the period	_	10	620	_	630	
At beginning of period	_	340	3,570	_	3,910	
At end of period	\$ —	\$ 350	\$ 4,190	\$ —	\$ 4,540	
•						

Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations

The following discussion and analysis of our financial condition contains forward-looking statements regarding industry outlook and our expectations regarding the performance of our business. These forward-looking statements are subject to numerous risks and uncertainties, including, but not limited to, the risks and uncertainties described under the heading "Forward Looking Statements," at the beginning of this report. Our actual results may differ materially from those contained in or implied by any forward-looking statements. You should read the following discussion together with the Company's reports on file with the Securities and Exchange Commission.

Introduction

We are an industrial manufacturer and distributor of highly engineered products serving focused markets in a diverse range of commercial, industrial and consumer applications. We have five reportable segments: Packaging, Energy, Aerospace & Defense, Engineered Components and Cequent. In reviewing our financial results, consideration should be given to certain critical events, particularly our acquisitions and consolidation, integration and restructuring efforts in several of our business operations.

Key Factors and Risks Affecting our Reported Results. Our businesses and results of operations depend upon general economic conditions and we serve some customers in cyclical industries that are highly competitive and themselves adversely impacted by unfavorable economic conditions. During the fourth quarter of 2008, worldwide credit markets and global economic conditions deteriorated significantly, resulting in declines in demand for the Company's products and services. These conditions persisted throughout 2009, resulting in reductions in sales and earnings from comparable prior periods across all of our reportable segments except Packaging. We have experienced generally higher levels of economic activity during the first quarter of 2010, which is one of the significant factors helping to generate year-over-year increases in revenue and earnings in all of our reportable segments except Aerospace & Defense. We expect that, although we've benefitted from some economic recovery in the first quarter of 2010, revenue and earnings will continue to trend below historical levels until the still unfavorable economic conditions improve.

Critical factors affecting our ability to succeed include: our ability to successfully pursue organic growth through product development, cross-selling and extending product-line offerings, and our ability to quickly and cost-effectively introduce new products; our ability to acquire and integrate companies or products that will supplement existing product lines, add new distribution channels, expand our geographic coverage or enable us to better absorb overhead costs; our ability to manage our cost structure more efficiently through improved supply base management, internal sourcing and/or purchasing of materials, selective outsourcing and/or purchasing of support functions, working capital management, and greater leverage of our administrative and overhead functions. If we are unable to do any of the foregoing successfully, our financial condition and results of operations could be materially and adversely impacted.

There is some seasonality in the businesses within our Cequent reportable segment, where sales of towing and trailering products are generally stronger in the second and third quarters, as trailer original equipment manufacturers ("OEMs"), distributors and retailers acquire product for the spring and summer selling seasons. No other reportable segment experiences significant seasonal fluctuation in its businesses. We do not consider sales order backlog to be a material factor in our business. A growing portion of our sales may be derived from international sources, which exposes us to certain risks, including currency risks.

The demand for some of our products, particularly in the Cequent segment, is heavily influenced by consumer sentiment. We experienced decreases in sales and earnings in 2008 and 2009 as a result of the general economic downturn, including an uncertain credit market and interest rate environment

and rising energy costs, among other things. While we are experiencing sales increases in certain of our Cequent businesses in 2010 as compared to 2009 as the economic conditions have begun to improve, we expect the current end market conditions in the Cequent segment will continue to remain weak and/or decline until the U.S. economy recovers from existing recessionary forces, employment levels increase and consumer credit availability improves, thereby increasing consumer discretionary spending.

We are sensitive to price movements in our raw materials supply base. Our largest material purchases are for steel, copper, aluminum, polyethylene and other resins and energy. Historically, we have experienced increasing costs of steel and resin and have worked with our suppliers to manage cost pressures and disruptions in supply. We have also initiated pricing programs to pass increased steel, copper, aluminum and resin costs to customers. Although we may experience delays in our ability to implement price increases, we generally are able to recover such increased costs. Although there have been no significant disruptions in the supply of steel since 2005, we may experience disruptions in supply in the future and we may not be able to pass along higher costs associated with such disruptions to our customers in the form of price increases. We will continue to take actions as necessary to manage risks associated with increasing steel or other raw material costs. However, such increased costs may adversely impact our earnings.

We report shipping and handling expenses, associated with certain businesses within our Cequent segment, for its sales distribution network, as an element of selling, general and administrative expenses in our consolidated statement of operations. As such, gross margins for the Cequent segment may not be comparable to other companies which include all costs related to their distribution network in cost of sales.

We have substantial debt, interest and lease payment requirements that may restrict our future operations and impair our ability to meet our obligations and, in a rising interest rate environment, our performance may be adversely affected by our degree of leverage.

Recent Consolidation, Integration and Restructuring Activities. During the past several years, we have undertaken significant consolidation, integration and other cost-savings programs to enhance our efficiency and achieve cost reduction opportunities which exist in our businesses. In addition to major consolidation projects, there have also been a series of ongoing initiatives to eliminate duplicative and excess manufacturing and distribution facilities, sales forces, and back office and other support functions in order to continue to optimize our cost structure in response to competitor actions and market conditions.

In the fourth quarter of 2008, in response to the deteriorating recent economic conditions, we announced and accelerated our Profit Improvement Plan, which included further consolidation of distribution and manufacturing activities, continued integration of certain business activities, movement of production to lowercost environments and expansion of strategic sourcing initiatives. We have also implemented reductions in salaried headcount and in fixed and variable spending to better align the fixed cost structure of these operating segments with the reality of our current market environment and to maintain or improve operating margins. We have implemented commercial actions to protect and gain market share through continued introduction of new and innovative products and by providing superior delivery and service to our customers. Further, we also have pricing initiatives in place to recover inflationary cost increases and we are continuing actions to leverage our businesses' strong brand names. During 2009, the Company realized savings of approximately \$32 million resulting from the actions taken as a part of the Profit Improvement Plan. These implemented actions were a significant driver of maintaining our gross profit margin in 2009 despite a 20% reduction in sales as compared to 2008, and should facilitate further margin expansion in 2010 should our revenues continue to increase.

The most significant element of our Profit Improvement Plan implemented during the first quarter of 2009 was the restructuring of our legacy towing, trailering and electrical businesses within our

Cequent reportable segment into one business, rationalizing facilities and the management team. This restructuring plan included the announcement and start of the closure process of the Mosinee, WI manufacturing facility, which ceased operations in the fourth quarter of 2009. We incurred approximately \$3.3 million of costs in the first quarter of 2009 associated with this initiative.

There were no significant charges recorded in the first quarter of 2010 related to further implementation of our Profit Improvement Plan initiatives.

Key Indicators of Performance. In evaluating our business, our management considers Adjusted EBITDA as a key indicator of financial operating performance and as a measure of cash generating capability. We define Adjusted EBITDA as net income (loss) before cumulative effect of accounting change, interest, taxes, depreciation, amortization, debt extinguishment costs, non-cash asset and goodwill impairment charges and write-offs and non-cash losses on sale-leaseback of property and equipment. In evaluating Adjusted EBITDA, our management deems it important to consider the quality of our underlying earnings by separately identifying certain costs undertaken to improve our results, such as costs related to consolidating facilities and businesses in an effort to eliminate duplicative costs or achieve efficiencies, costs related to integrating acquisitions and restructuring costs related to expense reduction efforts. Although we may undertake new consolidation, restructuring and integration efforts in the future as a result of our acquisition activity, our management separately considers these costs in evaluating underlying business performance. Caution must be exercised in considering these items as they include substantially (but not necessarily entirely) cash costs and there can be no assurance that we will ultimately realize the benefits of these efforts. Moreover, even if the anticipated benefits are realized, they may be offset by other business performance or general economic issues.

Management believes that consideration of Adjusted EBITDA together with a careful review of our results reported under GAAP is the best way to analyze our ability to service and/or incur indebtedness, as we are a highly leveraged company. We use Adjusted EBITDA as a key performance measure because we believe it facilitates operating performance comparisons from period to period and company to company by excluding potential differences caused by variations in capital structures (affecting interest expense), tax positions (such as the impact on periods or companies of changes in effective tax rates or net operating losses), and the impact of purchase accounting as well as depreciation and amortization expense. Because Adjusted EBITDA facilitates internal comparisons of our historical operating performance on a more consistent basis, we also use Adjusted EBITDA for business planning purposes, in measuring our performance relative to that of our competitors and in evaluating acquisition opportunities.

In addition, we believe Adjusted EBITDA and similar measures are widely used by investors, securities analysts, ratings agencies and other interested parties as a measure of financial performance and debt-service capabilities. Our use of Adjusted EBITDA has limitations as an analytical tool, and you should not consider it in isolation or as a substitute for analysis of our results as reported under GAAP. Some of these limitations are:

- it does not reflect our cash expenditures for capital equipment or other contractual commitments;
- although depreciation, amortization and asset impairment charges and write-offs are non-cash charges, the assets being depreciated, amortized or
 written off may have to be replaced in the future, and Adjusted EBITDA does not reflect cash capital expenditure requirements for such
 replacements;
- it does not reflect changes in, or cash requirements for, our working capital needs;
- it does not reflect the significant interest expense or the cash requirements necessary to service interest or principal payments on our indebtedness;

- it does not reflect certain tax payments that may represent a reduction in cash available to us;
- it includes amounts resulting from matters we consider not to be indicative of underlying performance of our fundamental business operations, as discussed in "Management's Discussion and Analysis of Financial Condition and Results of Operations;" and
- other companies, including companies in our industry, may calculate these measures differently and as the number of differences in the way two different companies calculate these measures increases, the degree of their usefulness as a comparative measure correspondingly decreases.

Because of these limitations, Adjusted EBITDA should not be considered as a measure of discretionary cash available to us to invest in the growth of our Company. We compensate for these limitations by relying primarily on our GAAP results and using Adjusted EBITDA only supplementally. We carefully review our operating profit margins (operating profit as a percentage of net sales) at a segment level, which are discussed in detail in our year-to-year comparison of operating results.

The following is a reconciliation of our net income (loss) to Adjusted EBITDA and cash flows provided by (used for) operating activities for the three months ended March 31, 2010 and 2009:

	Three months ended March 31,			
		2010		2009
		(dollars in	hou	sands)
Net income (loss)	\$	5,430	\$	(3,680)
Income tax expense (benefit)		4,470		(2,490)
Interest expense		14,290		12,530
Debt extinguishment costs		_		510
Depreciation and amortization		9,610		11,760
Adjusted EBITDA, total company	\$	33,800	\$	18,630
Interest paid		(5,250)		(4,770)
Taxes paid		(1,250)		(2,440)
(Gain) loss on dispositions of property and equipment		310		(50)
Gain on extinguishment of debt		_		(15,820)
Receivables sales and securitization, net		3,830		(6,130)
Net change in working capital		(35,530)		12,380
Cash flows provided by (used for) operating activities	\$	(4,090)	\$	1,800
	_			

The following table details certain items relating to our consolidation, restructuring and integration efforts that are included in the determination of net income (loss) under GAAP and are not added

back to net income (loss) in determining Adjusted EBITDA, but that we would consider in evaluating the quality of our Adjusted EBITDA:

	Three months ended March 31,
	2010 2009 (dollars in
	thousands)
Severance and business unit restructuring costs(a)	\$ — \$ 6,260
Gross gain on extinguishment of debt(b)	— (15,820)
	\$ — \$ (9,560)

- (a) Principally employee severance costs associated with business unit restructuring and other cost reduction activities.
- (b) Gains recognized in connection with the extinguishment of our senior subordinated notes due 2012, excluding debt extinguishment costs.

Segment Information and Supplemental Analysis

The following table summarizes financial information for our five current reportable segments for the three months ended March 31, 2010 and 2009:

		Three Months Ended March 31,	
	2010	As a Percentage of Net Sales 2009	As a Percentage of Net Sales
Net Sales:		(dollars in thousands)	
Packaging	\$ 43,600	19.8%\$ 30,250	15.0%
Energy	43,890	·	20.0%
Aerospace & Defense	17,080	,	11.0%
Engineered Components	18,910	8.6% 18,550	9.2%
Cequent	96,580		44.8%
Total	\$ 220,060	100.0% \$ 201,720	100.0%
Gross Profit:			·
Packaging	\$ 16,930	38.8%\$ 10,090	33.4%
Energy	11,580		24.8%
Aerospace & Defense	7,140		41.0%
Engineered Components	3,500	18.5% 2,200	11.9%
Cequent	23,910	24.8% 15,100	16.7%
Total	\$ 63,060	28.7% \$ 46,460	23.0%
Selling, General and Administrative:			
Packaging	\$ 4,810	11.0%\$ 4,750	15.7%
Energy	6,360	14.5% 6,440	16.0%
Aerospace & Defense	3,280	,	10.3%
Engineered Components	1,680	,	9.8%
Cequent	15,790		20.4%
Corporate expenses	5,780		N/A
Total	\$ 37,700	17.1% \$ 41,300	20.5%
Operating Profit:			
Packaging	\$ 11,860	· · · · · · · · · · · · · · · · · · ·	17.9%
Energy	5,180	· · · · · · · · · · · · · · · · · · ·	8.7%
Aerospace & Defense	3,860		30.7%
Engineered Components	1,810		2.0%
Cequent	8,120	· · · · ·	
Corporate expenses	(5,780		
Total	\$ 25,050	11.4%\$ 5,200	2.6%
Adjusted EBITDA:			
Packaging	\$ 14,920	· · · · · · · · · · · · · · · · · · ·	28.6%
Energy	5,900	,	10.6%
Aerospace & Defense	4,520		33.4%
Engineered Components	2,570		6.0%
Cequent	12,120		1.5%
Corporate (expenses) income	(5,900	·	N/A
Subtotal from continuing operations	34,130	<u> </u>	15.1%
Discontinued operations	(330		
Total company	\$ 33,800	15.4% \$ 18,630	9.2%

Results of Operations

The principal factors impacting us during the three months ended March 31, 2010 compared with the three months ended March 31, 2009, were:

- costs incurred and savings realized related to our Profit Improvement Plan, primarily in our Packaging and Cequent segments;
- the impact of an upturn in economic conditions in the first quarter of 2010 as compared to the global economic recession in the first quarter of 2009, contributing to increased net sales in four of our five reportable segments;
- decreases in the value of the U.S. dollar as compared to the currencies in other countries where we operate; and
- gains on extinguishment of debt in the first quarter of 2009 resulting from the repurchase of our 9⁷/8% senior subordinated notes at prices below their face value.

Three Months Ended March 31, 2010 Compared with Three Months Ended March 31, 2009

Overall, net sales increased approximately \$18.4 million, or approximately 9.1%, for the three months ended March 31, 2010, as compared with the three months ended March 31, 2009. During the first quarter of 2010, net sales increased in each of our five reportable segments except Aerospace & Defense, generally due to higher sales volumes resulting from the upturn in economic conditions as compared to the global economic recession we experienced in the first quarter of 2009. In addition, net sales were favorably impacted by approximately \$6.6 million as a result of currency exchange, as our reported results in U.S. dollars were positively impacted by stronger foreign currencies.

Gross profit margin (gross profit as a percentage of sales) approximated 28.7% and 23.0% for the three months ended March 31, 2010 and 2009, respectively. The increase in profit margin is attributed primarily to the realization of savings resulting from our cost reduction and alternate sourcing initiatives implemented in connection with our Profit Improvement Plan, with the largest impact experienced in our Packaging and Cequent segments.

Operating profit margin (operating profit as a percentage of sales) approximated 11.4% and 2.6% for the three months ended March 31, 2010 and 2009, respectively. Operating profit increased approximately \$19.9 million, or 381.7%, to \$25.1 million for the three months ended March 31, 2010, from \$5.2 million for the three months ended March 31, 2009, primarily as a result of higher sales volumes, higher gross profits resulting from savings realized in connection with our Profit Improvement Plan and reduced selling, general and administrative expenses, as our businesses were able to hold or reduce their spending levels year-over-year despite the increase in sales levels.

Adjusted EBITDA margin from continuing operations (Adjusted EBITDA as a percentage of sales) approximated 15.5% and 15.1% for the three months ended March 31, 2010 and 2009, respectively. Adjusted EBITDA increased approximately \$3.7 million for the three months ended March 31, 2010, as compared to the three months ended March 31, 2009. After consideration of the \$15.8 million gross gain on extinguishment of debt in the first quarter of 2009 that did not recur in 2010, \$0.7 million of costs incurred in connection with our receivables facility in the first quarter of 2009 that were included in other, net while the \$0.3 million of costs incurred in the first quarter of 2010 in connection with the facility are included in interest expense, \$0.4 million of losses on transactions denominated in foreign currencies in the three months ended March 31, 2010 as compared to \$0.2 million of gains on transactions denominated in foreign currencies in the three months ended March 31, 2009 and a decrease in year-over-year depreciation expense of approximately \$0.5 million, the change in Adjusted EBITDA is consistent with the change in operating profit between years.

See below for a discussion of operating results by segment.

Packaging. Net sales increased approximately \$13.4 million, or 44.1%, to \$43.6 million in the three months ended March 31, 2010, as compared to \$30.2 million in the three months ended March 31, 2009. Overall, sales increased approximately \$1.3 million due to currency exchange, as our reported results in U.S. dollars were positively impacted as a result of the weaker U.S. dollar relative to foreign currencies. Sales of our specialty dispensing products and new product introductions increased by approximately \$6.8 million in the three months ended March 31, 2010 compared to the three months ended March 31, 2009, due primarily to increased sales into the personal care markets, pharmaceuticals and the food industries. Sales of our industrial closures, rings and levers increased by approximately \$5.3 million in the three months ended March 31, 2010 compared to the three months ended March 31, 2009, primarily as a result of the moderate general economic recovery.

Packaging's gross profit increased approximately \$6.8 million to \$16.9 million, or 38.8% of sales, in the three months ended March 31, 2010, as compared to \$10.1 million, or 33.4% of sales, in the three months ended March 31, 2009. Of the increase in gross profit, approximately \$3.9 million relates to the increase in sales volumes between years. In addition, approximately \$0.6 million is as a result of the favorable currency exchange. The remainder of the increase in gross profit, which approximated a 190 basis point improvement year-over-year, resulted primarily from initiatives implemented in connection with our Profit Improvement Plan, including productivity projects to improve efficiency and throughput and alternate sourcing or improved internal processing for certain production materials.

Packaging's selling, general and administrative expenses increased approximately \$0.1 million to \$4.8 million, or 11.0% of sales, in the three months ended March 31, 2010, as compared to \$4.7 million, or 15.7% of sales, in the three months ended March 31, 2009, as Packaging was able to hold its selling, general and administrative spending at a consistent level despite the significant increase in sales as a result of the fixed cost reductions implemented throughout 2009.

Packaging's operating profit increased approximately \$6.5 million to \$11.9 million, or 27.2% of sales, in the three months ended March 31, 2010, as compared to \$5.4 million, or 17.9% of sales, in three months ended March 31, 2009. The increase in operating profit between years is primarily attributed to increased sales volumes, implemented projects associated with our Profit Improvement Plan and favorable currency exchange.

Packaging's Adjusted EBITDA increased approximately \$6.3 million to \$14.9 million, or 34.2% of sales, in the three months ended March 31, 2010, as compared to \$8.6 million, or 28.6% of sales, in the three months ended March 31, 2009, consistent with the change in operating profit between years.

Energy. Net sales for the three months ended March 31, 2010 increased approximately \$3.6 million, or 9.0%, to \$43.9 million, as compared to \$40.3 million in the three months ended March 31, 2009. Sales of specialty gaskets and related fastening hardware increased approximately \$3.3 million as a result of increased levels of turn-around activity at petrochemical refineries and increased sales demand from the chemical industry, as customers have begun to perform maintenance work and new programs deferred from 2009 that require our replacement and specialty gaskets and hardware. Sales within our engine business increased by approximately \$0.3 million due to increased sales of gas processing equipment.

Gross profit within Energy increased approximately \$1.6 million to \$11.6 million, or 26.4% of sales, in the three months ended March 31, 2010, as compared to \$10.0 million, or 24.8% of sales, in the three months ended March 31, 2009. Gross profit increased approximately \$0.9 million as a result of the increase in sales levels between years. The remaining increase in gross profit is primarily attributable to increased material margins resulting from successful negotiations with our overseas vendors to reduce the purchase price of certain commodities and inbound freight rates.

Selling, general and administrative expenses within Energy remained essentially flat at \$6.4 million year-over-year, but decreased as a percent of net sales from 16.0% for the three months ended

March 31, 2009 to 14.5% for the three months ended March 31, 2010, as the businesses in this segment were able to hold their spending levels constant despite the increase in sales levels between years.

Overall, operating profit within Energy increased approximately \$1.7 million to \$5.2 million, or 11.8% of sales, in the three months ended March 31, 2010, as compared to \$3.5 million, or 8.7% of sales, in the three months ended March 31, 2009, due principally to the leverage gained by higher sales volumes and due to increased material margins.

Energy's Adjusted EBITDA increased \$1.6 million to \$5.9 million, or 13.4% of sales, in the three months ended March 31, 2010, as compared to \$4.3 million, or 10.6% of sales, in the three months ended March 31, 2009, consistent with the increase in operating profit between years.

Aerospace & Defense. Net sales for the three months ended March 31, 2010 decreased approximately \$5.1 million, or 23.1%, to \$17.1 million, as compared to \$22.2 million in the three months ended March 31, 2009. Sales in our aerospace business decreased approximately \$4.2 million, primarily due to continued lower demand from our distribution customers in response to high inventory levels in the supply chain and consolidation within the distribution segment of the aerospace hardware industry. Sales in our defense business decreased approximately \$0.9 million. Our defense business did not sell any cartridge cases in the first quarter of 2010 due to the ongoing relocation of the defense facility, which began in the second quarter of 2009, as compared to \$3.0 million of cartridge case sales in the first quarter of 2009 in advance of the facility relocation. The decrease in cartridge case sales was partially offset by increase of approximately \$2.1 million in revenue primarily associated with managing the relocation and closure of the defense facility.

Gross profit within Aerospace & Defense decreased approximately \$2.0 million to \$7.1 million, or 41.8% of sales, in the three months ended March 31, 2010, from \$9.1 million, or 41.0% of sales, in the three months ended March 31, 2009, due primarily to the decline in sales levels between years. In addition, although our aerospace business experienced a more favorable product sales mix and slightly better material margins due to cost reductions in the first quarter of 2010 as compared to the first quarter of 2009, this impact was approximately offset by unfavorable absorption of fixed costs resulting from the decline in sales volumes.

Selling, general and administrative expenses increased approximately \$1.0 million to \$3.3 million, or 19.2% of sales, in the three months ended March 31, 2010, as compared to \$2.3 million, or 10.3% of sales, in the three months ended March 31, 2009 due primarily to increased attorney and legal fee costs within our defense business.

Operating profit within Aerospace & Defense decreased approximately \$2.9 million to \$3.9 million, or 22.6% of sales, in the three months ended March 31, 2010, as compared to \$6.8 million, or 30.7% of sales, in the three months ended March 31, 2009, primarily due to lower sales volumes and higher selling, general and administrative expenses due to attorney and legal fees.

Aerospace & Defense's Adjusted EBITDA decreased approximately \$2.9 million to \$4.5 million, or 26.5% of sales, in the three months ended March 31, 2010, as compared to \$7.4 million, or 33.4% of sales, in the three months ended March 31, 2009, consistent with the decrease in operating profit between years.

Engineered Components. Net sales for the three months ended March 31, 2010 increased approximately \$0.4 million, or 1.9%, to \$18.9 million, as compared to \$18.5 million in the three months ended March 31, 2009. Sales within our specialty fittings business increased by approximately \$3.0 million, as our new product offerings for automotive fuel systems increased by approximately \$1.9 million and sales of our core tube nut products increased by approximately \$1.1 million as a result of the recent economic upturn. Sales in our precision tool cutting businesses increased by approximately \$0.9 million, primarily as a result of the upturn in the domestic economy and market share gains in our countersink product line. Sales in our industrial cylinder business decreased by

approximately \$3.5 million, primarily due to the global economic recession, as the recent upturn in the economy has not yet manifested in additional sales in this business, and higher inventory levels at two of its largest customers, who significantly lowered orders in the first quarter of 2010 as compared to the first quarter of 2009.

Gross profit within Engineered Components increased approximately \$1.3 million to \$3.5 million, or 18.5% of sales, in the three months ended March 31, 2010, from \$2.2 million, or 11.9% of sales, in the three months ended March 31, 2009. All of the businesses within this segment improved their gross profit margin year-over-year due to cost reduction and internal productivity efforts implemented in 2009 and the first quarter of 2010 in response to the economic slowdown. In addition, Engineered Components experienced higher costs in the first quarter of 2009 due to the impact of lower absorption of fixed costs resulting from lower production and/or sales levels in all three businesses and our industrial cylinder business' sale of higher-cost inventory in excess of their ability to secure price increases.

Selling, general and administrative expenses decreased approximately \$0.1 million to \$1.7 million, or 8.9% of sales, in the three months ended March 31, 2010, as compared to \$1.8 million, or 9.8% of sales, in the three months ended March 31, 2009 as Engineered Components was able to hold its selling, general and administrative spending at a consistent level.

Operating profit within Engineered Components increased \$1.4 million to approximately \$1.8 million, in the three months ended March 31, 2010, as compared to operating profit of \$0.4 million in the three months ended March 31, 2009, primarily due to savings realized resulting from cost reduction and productivity projects implemented in the last twelve months and due to higher costs related to lower absorption of fixed costs and sales of higher-cost inventory, both of which were incurred in the first quarter of 2009 and were not present in the first quarter of 2010.

Engineered Components' Adjusted EBITDA increased \$1.5 million to \$2.6 million, or 13.6% of sales, in the three months ended March 31, 2010, as compared to \$1.1 million, or 6.0% of sales, in the three months ended March 31, 2009, consistent with the increase in operating profit between years.

Cequent. Net sales increased approximately \$6.1 million to \$96.6 million in the three months ended March 31, 2010, as compared to \$90.5 million in the three months ended March 31, 2009. Net sales were favorably impacted by approximately \$5.0 million of currency exchange, as our reported results in U.S. dollars were positively impacted as a result of the weaker U.S. dollar relative to foreign currencies. Sales in our retail business decreased \$4.2 million, as the continued impact of new customers added in the second half of 2009 was more than offset by a one-time inventory pipeline fill for new products sold to a significant customer in the first quarter of 2009 for which there were no similar pipeline fills in the first quarter of 2010. Sales within our performance products business (includes the legacy towing, trailering and electrical businesses) increased by \$1.1 million, due to stabilization of the markets that this business serves resulting from the upturn in the domestic economy and due to new product introductions in both our original equipment manufacturer and aftermarket businesses. After considering the impact of currency exchange, sales in our Australia/Asia Pacific business increased approximately \$4.2 million, due primarily to continued higher original equipment manufacturer and aftermarket sales resulting from new government stimulus incentives announced during the third quarter of 2009, and continued market share gains.

Cequent's gross profit increased approximately \$8.8 million to \$23.9 million, or 24.8% of sales, in the three months ended March 31, 2010, from approximately \$15.1 million, or 16.7% of sales, in the three months ended March 31, 2009. Of this increase, approximately \$0.2 million is due to the increased sales volumes between periods and \$1.5 million is due to favorable currency exchange. The most significant reason for the increase is our cost reduction efforts implemented throughout 2009 as a part of our Profit Improvement Plan to resize the business and its fixed cost structure to recent demand levels, to identify alternate lower-cost foreign-sourced suppliers and to implement productivity

initiatives to increase manufacturing efficiencies. The largest item within the Profit Improvement Plan was the closure of the Mosinee, WI manufacturing facility, which was completed in the fourth quarter of 2009. Cequent incurred approximately \$0.9 million of costs impacting gross profit in the first quarter of 2009 to implement these actions, for which the benefit was realized in the first quarter of 2010 with no significant additional implementation costs incurred in 2010.

Selling, general and administrative expenses decreased approximately \$2.7 million to \$15.8 million, or 16.3% of sales, in the three months ended March 31, 2010, as compared to \$18.5 million, or 20.4% of sales, in the three months ended March 31, 2009. Total expense decreased by \$0.3 million after consideration of approximately \$2.4 million of costs incurred associated with implementing the Profit Improvement Plan in the first quarter of 2009, primarily related to severance charges recorded in connection with the announcement of the closure of the Mosinee, WI facility. Cequent continues to monitor and minimize its selling, general and administrative spending levels consistent with the current sales levels.

Cequent's operating profit increased approximately \$11.5 million to approximately \$8.1 million, or 8.4% of sales, in the three months ended March 31, 2010, from an operating loss of \$3.4 million, or (3.7)% of net sales, in the three months ended March 31, 2009. The increase in operating profit is due primarily to the cost reductions, alternate sourcing arrangements and productivity initiatives that have been implemented as part of the Profit Improvement Plan, for which savings are being realized in the first quarter of 2010, and approximately \$3.3 million of costs incurred in the first quarter of 2009 to implement such actions.

Cequent's Adjusted EBITDA increased approximately \$10.8 million to \$12.1 million, or 12.5% of sales, for the three months ended March 31, 2010, from \$1.3 million, or 1.5% of sales, for the three months ended March 31, 2009. After consideration of approximately \$0.5 million of lower depreciation expense in the three months ended March 31, 2010 compared with the three months ended March 31, 2009, due primarily to the closure of the Mosinee, WI facility, and \$0.1 million of losses on transactions denominated in foreign currencies as compared to \$0.1 million of gains on transactions denominated in foreign currencies, the change in Adjusted EBITDA is consistent with the increase in operating profit between years.

Corporate Expenses (Income). Corporate expenses (income) included in operating profit and Adjusted EBITDA consists of the following:

	2	Three er <u>Mar</u> 010 (in n	nded rch 3	1 31, 2009
Corporate operating expenses	\$	2.6	\$	2.9
Employee costs and related benefits		3.2		4.7
Corporate expenses—operating profit	\$	5.8	\$	7.6
Receivables sales and securitization expenses		_		0.7
Gain on extinguishment of debt		_		(15.8)
Other, net		0.1		(0.1)
Corporate expenses (income)—Adjusted EBITDA	\$	5.9	\$	(7.6)

Corporate expenses decreased approximately \$1.8 million, to \$5.8 million, for the three months ended March 31, 2010, from \$7.6 million for the three months ended March 31, 2009. The decrease between years is primarily attributed to the severance arrangement associated with the termination of our former chief executive officer in January 2009, which was partially offset by increased short and long-term incentive equity and cash compensation expense. Receivables sales and securitization

expenses decreased by approximately \$0.8 million for the three months ended March 31, 2010 compared with the three months ended March 31, 2009, as new accounting guidance effective in the first quarter of 2010 requires that we account for the facility similar to our credit facility debt, where amounts outstanding under the facility are classified on the balance sheet as debt and costs incurred under the facility are classified as interest expense on the statement of operations. In addition, during the first quarter of 2009, we retired approximately \$31.8 million face value of our senior subordinated notes, resulting in a net gain of approximately \$15.3 million after considering non-cash debt extinguishment costs of \$0.5 million.

Interest Expense. Interest expense increased approximately \$1.6 million, to \$14.1 million, for the three months ended March 31, 2010, as compared to \$12.5 million for the three months ended March 31, 2009. The increase is primarily attributable to an unfavorable change in the fair value of our interest rate swaps of \$1.5 million during the first quarter of 2010, \$0.3 million of aggregate costs incurred under the receivables facility as of March 31, 2010, previously recorded in other expense, net and an increase in our effective weighted-average interest rate on variable rate U.S. borrowings to approximately 5.6% during the first quarter of 2010, from approximately 4.1% during the first quarter of 2009. These increases were partially offset by a decrease in our weighted-average U.S. borrowings to approximately \$274.3 million in the three months ended March 31, 2010 from approximately \$308.7 million in the three months ended March 31, 2009. In addition, we recorded approximately \$1.8 million lower interest expense related to our senior subordinated notes in 2010 as compared to 2009, due primarily our repurchase of \$41.4 million face value of our former senior subordinated notes during the second and third quarters of 2009.

Other Expense, Net. Other expense, net decreased approximately \$0.2 million, to \$0.5 million for the three months ended March 31, 2010, as compared to \$0.7 million for the three months ended March 31, 2009. In the first quarter of 2010, we incurred approximately \$0.4 million of losses on transactions denominated in foreign currencies. In the first quarter of 2009, we recognized approximately \$0.2 million in gains on transactions denominated in foreign currencies and incurred \$0.7 million of expenses in connection with the both the use and renewal of our receivables facility. There were no other individually significant amounts incurred or changes in amounts incurred in either of the three month periods ended March 31, 2010 and 2009.

Income Taxes. The effective income tax rates for the three months ended March 31, 2010 and 2009 were 44.7% and 37.0%, respectively. The increase in the effective tax rate in the three months ended March 31, 2010 compared to the three months ended March 31, 2009 is related to a tax charge of approximately \$1.0 million due to the impact of certain Subpart F exceptions under U.S. tax legislation that recently expired.

Discontinued Operations. The results of discontinued operations consist of our medical device and property management lines of business, which are classified as held for sale for all periods presented, and our specialty laminates, jacketings and insulation tapes line of business, which was sold in February 2009. Our loss from discontinued operations, net of income tax benefit, was \$0.3 million and \$8.3 million for the three months ended March 31, 2010 and 2009, respectively. See Note 2, "Discontinued Operations and Assets Held for Sale," to our consolidated financial statements included in Part I, Item 1 of this report on Form 10-Q.

Liquidity and Capital Resources

Cash Flows

Cash used for operating activities for the three months ended March 31, 2010 was approximately \$4.1 million, as compared to cash provided by operating activities of \$1.8 million for the three months

ended March 31, 2009. Significant changes in cash flows provided by (used for) operating activities and the reasons for such changes are as follows:

- For the three months ended March 31, 2010, the Company generated \$16.8 million of cash, based on the reported net income from operations of \$5.4 million and after considering the effects of non-cash items related to gains/losses on dispositions of property and equipment, depreciation and amortization, changes in deferred taxes, stock compensation and other non-cash items. For the three months ended March, 31 2009, the Company used \$10.4 million in cash flows based on the reported net loss from operations of \$3.7 million and after considering the effects of non-cash impacts related to gains/losses on dispositions of property and equipment, depreciation and amortization, changes in deferred taxes, debt extinguishments, stock compensation and other non-cash items.
- For the three months ended March 31, 2010, activity related to receivables sales resulted in net proceeds of approximately \$3.8 million. For the three months ended March 31, 2009, activity related to the sale of receivables and use of our receivables facility resulted in a net cash use of approximately \$6.1 million, as we reduced amounts outstanding under the facility relative to December 31, 2008. During the first three months of 2009, we relied primarily on our revolving credit facility as the principal source of funding our working capital requirements and ordinary course needs, as it was our lowest cost source of borrowings.
- Increases in receivables resulted in a use of cash of approximately \$39.0 million and \$2.6 million for the three months ended March 31, 2010 and 2009, respectively, primarily due to the increase in sales in the first quarter of 2010 and a heavier mix toward sales to foreign customers, who typically have longer terms for payment. Our accounts receivable past-due aging has not changed significantly year-over-year.
- For the three months ended March 31, 2010 and 2009, we reduced our investment in inventory consistent with our management strategy to improve inventory turns and to better align inventory levels with end market demand, which resulted in a cash source of approximately \$6.1 million and \$18.1 million, respectively.
- For the three months ended March 31, 2010 and 2009, accounts payable and accrued liabilities resulted in a net source of cash of approximately \$7.9 million and \$1.2 million, respectively. The increase in accounts payable and accrued liabilities as of March 31, 2010 is primarily a result of increased production activity in the first quarter of 2010 compared with the fourth quarter of 2009 due to the upturn in economic activity.
- Management of prepaid expenses and other assets resulted in a source of cash of approximately \$0.3 million and \$1.7 million for the three months
 ended March 31, 2010 and 2009, respectively, primarily as a result of ongoing initiatives to reduce the relative level of investment in
 manufacturing supplies, spare parts and tooling assets.

Net cash used for investing activities for the three months ended March 31, 2010 was approximately \$2.6 million, as compared to net cash provided by investing activities of \$17.4 million for the three months ended March 31, 2009. During the first three months of 2010, our investing activities related primarily to capital expenditures, which remained below historical levels due to the lower sales levels given the economic downturn. During the first three months of 2009, we generated approximately \$20.7 million of cash from business and asset dispositions, primarily related to the sale of our specialty laminates, jacketings and insulation tapes line of business. We also incurred approximately \$3.3 million in capital expenditures to support our growth initiatives.

Net cash provided by financing activities for the three months ended March 31, 2010 was approximately \$3.8 million, as compared to net cash used for financing activities of approximately \$18.6 million for the three months ended March 31, 2009. During the first quarter of 2010, we

increased our amounts outstanding on our revolving credit facility by approximately \$4.3 million. During the first quarter of 2009, we used approximately \$16.0 million of available cash to retire \$31.8 million face value of our previous 9⁷/8% senior subordinated notes via open market purchases. In addition we decreased amounts outstanding on our revolving credit facilities during the first three months of 2009 by approximately \$1.8 million.

Our Debt and Other Commitments

During the fourth quarter of 2009, the Company amended and restated its credit facilities. Prior to the amendment and restatement, the credit facilities consisted of a \$90.0 million revolving credit facility, a \$60.0 million deposit-linked supplemental revolving credit facility and a \$260.0 million term loan facility. Under the amended and restated credit facilities, the revolving credit facility was reduced to \$83.0 million, while the supplemental revolving credit facility and term loan facility remained at \$60.0 million and \$252.2 million, respectively (collectively, the Amended and Restated Credit Agreement or "ARCA"). Under the ARCA, the Company extended the maturity of \$70.0 million of its revolving credit facility until December 15, 2013, and the maturity of \$226.3 million of its term loan until December 15, 2015. The maturity date of \$8.0 million of its revolving credit facility and the \$60.0 million deposit-linked supplemental revolving credit facility remained at August 2, 2011, and the maturity date of \$25.9 million of its term loan remained at August 2, 2013. At March 31, 2010, approximately \$250.9 million was outstanding on the term loan and \$13.2 million was outstanding on the revolving credit facilities. Under the ARCA, up to \$25.0 million of our revolving credit facility in the aggregate is available in 2010 to be used for one or more permitted acquisitions subject to certain conditions and other outstanding borrowings and issued letters of credit.

Amounts drawn under our revolving credit facilities fluctuate daily based upon our working capital and other ordinary course needs. Availability under our revolving credit facilities depends upon, among other things, compliance with our credit agreement's financial covenants. Our credit facilities contain negative and affirmative covenants and other requirements affecting us and our subsidiaries, including among others: restrictions on incurrence of debt (except for permitted acquisitions and subordinated indebtedness), liens, mergers, investments, loans, advances, guarantee obligations, acquisitions, asset dispositions, saleleaseback transactions, hedging agreements, dividends and other restricted junior payments, stock repurchases, transactions with affiliates, restrictive agreements and amendments to charters, by-laws, and other material documents. The terms of our credit agreement require us and our subsidiaries to meet certain restrictive financial covenants and ratios computed quarterly, including a leverage ratio (total consolidated indebtedness plus outstanding amounts under the accounts receivable securitization facility over consolidated EBITDA, as defined), interest expense coverage ratio (consolidated EBITDA, as defined, over cash interest expense, as defined) and a capital expenditures covenant. The most restrictive of these financial covenants are the leverage ratio and interest expense coverage ratio. Our permitted leverage ratio under the ARCA is 5.00 to 1.00 for January 1, 2010 to March 31, 2010, 5.25 to 1.00 for April 1, 2010 to June 30, 2010, 5.00 to 1.00 for July 1, 2010 to December 31, 2010, 4.75 to 1.00 for January 1, 2011 to June 30, 2011, 4.50 to 1.00 for July 1, 2011 to September 30, 2011, 4.25 to 1.00 for October 1, 2011 to September 30, 2012, 4.00 to 1.00 for October 1, 2012 to June 30, 2013 and 3.50 to 1.00 from July 1, 2013 and thereafter. Our actual leverage ratio was 3.69 to 1.00 at March 31, 2010. Our permitted interest expense coverage ratio under the ARCA is 2.30 to 1.00 for January 1, 2010 to March 31, 2010, 2.15 to 1.00 for April 1, 2010 to June 30, 2010, 2.00 to 1.00 for July 1, 2010 to June 30, 2011, 2.25 to 1.00 for July 1, 2011 to June 30, 2012, 2.40 to 1.00 for July 1, 2012 to December 31, 2012, 2.50 to 1.00 for January 1, 2013 to September 30, 2013 and 2.75 to 1.00 for October 1, 2013 and thereafter. Our actual interest expense coverage ratio was 3.07 to 1.00 at March 31, 2010. At March 31, 2010, we were in compliance with our financial covenants.

The following is a reconciliation of net income (loss), as reported, which is a GAAP measure of our operating results, to Consolidated Bank EBITDA, as defined in our credit agreement, for the twelve months ended March 31, 2010:

	Year Ended ember 31, 2009	Less: Three Months Ended March 31, 2009 (dollars in the	Add: Three Months Ended March 31, 2010 nousands)	Twelve Months Ended March 31, 2010
Net income (loss), as reported	\$ (220)	\$ (3,680)	\$ 5,430	\$ 8,890
Bank stipulated adjustments:				
Interest expense, net (as defined)	45,720	12,530	14,290	47,480
Income tax expense (benefit)(1)	(520)	(2,490)	4,470	6,440
Depreciation and amortization	43,940	11,760	9,610	41,790
Extraordinary non-cash charges(2)	3,270	_	_	3,270
Monitoring fees(3)	2,890	_	_	2,890
Interest equivalent costs(4)	1,530	570	_	960
Non-cash compensation expense	1,370	30	480	1,820
Other non-cash expenses or losses	3,570	420	830	3,980
Non-recurring expenses or costs for cost savings				
projects(6)	10,940	3,500	_	7,440
Debt extinguishment costs(7)	11,400	510		10,890
Negative EBITDA from discontinued operations(8)	3,720	100	440	4,060
Permitted dispositions(9)	12,130	11,520	(110)	500
Consolidated Bank EBITDA, as defined	\$ 139,740	\$ 34,770	\$ 35,440	\$ 140,410

	(rch 31, 2010 dollars in housands)
Total long-term debt	\$	518,520
Aggregate funding under the receivables securitization facility		_
Total Consolidated Indebtedness, as defined	\$	518,520
Consolidated Bank EBITDA, as defined	\$	140,410
Actual leverage ratio		3.69x
Covenant requirement		5.00x

	Year I December	Ended r 31, 2009	Three En March	ess: Months ded 31, 2009 lollars in th	Three E March	Add: Months nded 131, 2010	 lve Months Ended ech 31, 2010
Interest expense, as reported	\$	45,720	\$	12,530	\$	14,290	\$ 47,480
Bank stipulated adjustments:							
Interest equivalent costs from receivables financing		1,530		570		_	960
Interest income		(310)		(100)		(90)	(300)
Non-cash amounts attributable to amortization of							
financing costs		(2,240)		(610)		(720)	(2,350)
Total Consolidated Cash Interest Expense, as defined	\$	44,700	\$	12,390	\$	13,480	\$ 45,790

	Ma	rch 31, 2010
		dollars in housands)
Consolidated Bank EBITDA, as defined	\$	140,410
Total Consolidated Cash Interest Expense, as defined		45,790
Actual interest expense ratio		3.07x
Covenant requirement		2.30x

- (1) Amount includes tax expense (benefits) associated with discontinued operations.
- (2) Non-cash charges associated with tangible and intangible asset impairments, including goodwill.
- (3) Represents management fees and expenses paid to Heartland and/or its affiliates pursuant to the Heartland Advisory Agreement.
- (4) Interest-equivalent costs associated with the Company's receivables securitization facility.
- (5) Non-cash expenses resulting from the grant of restricted shares of common stock and common stock options.
- (6) Non-recurring costs and expenses relating to cost savings projects, including restructuring and severance expenses, not to exceed \$32,000,000 in the aggregate, subsequent to October 1, 2009.
- Costs incurred in connection with amending and restating our credit facilities, issuance of our $9^3/4\%$ senior secured notes and the retirement of our $9^7/8\%$ senior subordinated notes.
- (8) Not to exceed \$10,000,000 in any fiscal year.
- (9) EBITDA from permitted dispositions, as defined.

Two of our international businesses are also parties to loan agreements with banks, denominated in their local currencies.

In the United Kingdom, we are party to a revolving debt agreement with a bank in the amount of £1.0 million, At March 31, 2010 the balance outstanding under this agreement, which is secured by a letter of credit under our credit facilities, was approximately \$0.7 million at an interest rate of 2.5%.

In Australia, we are party to a debt agreement with a bank in the amount of \$23.0 million Australian dollars which expires December 31, 2010. At March 31, 2010, the balance outstanding under this agreement was approximately \$8.3 million at an interest rate of 6.8%. Borrowings under this arrangement are secured by substantially all the assets of our local business which is also subject to

financial and reporting covenants. Financial covenants include a capital adequacy ratio (tangible net worth over total tangible assets) and an interest coverage ratio (EBIT over gross interest cost) and we were in compliance with such covenants at March 31, 2010. In addition to the financial covenants there are other financial restrictions such as: restrictions on dividend payments, U.S. parent loan repayments, negative pledge and undertakings with respect to related entities.

Another important source of liquidity is our \$75.0 million accounts receivable facility, under which we have the ability to sell eligible accounts receivable to a third-party multi-seller receivables funding company. As of March 31, 2010, we had no amounts funded under the facility with \$51.7 million available but not utilized.

At March 31, 2010, our available revolving credit capacity of \$143.0 million under our credit facility was reduced by approximately \$31.0 million of letters of credit outstanding as of that date. The letters of credit are used for a variety of purposes, including support of certain operating lease agreements, vendor payment terms and other subsidiary operating activities, and to meet various states' requirements to self-insure workers' compensation claims, including incurred but not reported claims. After consideration of outstanding letters of credit and \$13.2 million outstanding under our supplemental revolving credit facility at March 31, 2010, we had \$98.8 million of revolving credit capacity available, in addition to \$51.7 million of available liquidity under our accounts receivable facility discussed above. However, after consideration of our leverage covenant, we had aggregate available funding under our revolving credit and accounts receivable facilities of \$150.5 million at March 31, 2010.

Our available revolving credit capacity under our credit facility, after consideration of approximately \$31.0 million in letters of credit outstanding related thereto, is approximately \$112.0 million, while our available liquidity under our accounts receivable securitization facility ranges from \$30 million to \$55 million, depending on the level of our receivables outstanding at a given point in time during the year. We rely upon our cash flow from operations and available liquidity under our revolving credit and accounts receivable facilities to fund our debt service obligations and other contractual commitments, working capital and capital expenditure requirements. Generally, we use available liquidity under these facilities to fund capital expenditures and daily working capital requirements during the first half of the year, as we experience some seasonality in our Cequent operating segment. Sales of towing and trailering products within this segment are generally stronger in the second and third quarters, as original equipment manufacturers (OEMs), distributors and retailers acquire product for the spring and summer selling seasons. None of our other operating segments experience any significant seasonal fluctuations in their respective businesses. During the second half of the year, the investment in working capital is reduced and amounts outstanding under our credit and securitization facilities are paid down. At the end of each quarter, we generally use cash on hand to pay down amounts outstanding under our revolving credit and accounts receivable facilities.

Cash management related to our revolving credit and accounts receivable facilities is centralized. We monitor our cash position and available liquidity on a daily basis and forecast our cash needs on a weekly basis within the current quarter and on a monthly basis outside the current quarter over the remainder of the year. Our business and related cash forecasts are updated monthly. Given aggregate available funding under our revolving credit and accounts receivable facilities of \$150.5 million at March 31, 2010, after consideration of the aforementioned leverage restrictions, and based on forecasted cash sources and requirements inherent in our business plans, we believe that our liquidity and capital resources, including anticipated cash flows from operations, will be sufficient to meet our debt service, capital expenditure and other short-term and long-term obligation needs for the foreseeable future.

We also have \$250.0 million (face value) $9^3/4\%$ senior secured notes ("Senior Notes") outstanding at March 31, 2010, due 2017. Interest on the Senior Notes accrues at the rate of 9.75% per annum and is payable semi-annually in arrears on June 15 and December 15.

The Senior Notes are general senior secured obligations of the Company and are pari passu in right of payment with all existing and future indebtedness of the Company that is not subordinated in right of payment to the Senior Notes.

Prior to December 15, 2012, the Company may redeem up to 35% of the principal amount of Senior Notes at a redemption price equal to 109.750% of the principal amount, plus accrued and unpaid interest to the applicable redemption date plus additional interest, if any, with the net cash proceeds of one or more equity offerings, provided that at least 65% of the original principal amount of Senior Notes issued remains outstanding after such redemption, and provided further that each such redemption occurs within 90 days of the date of closing of each such equity offering.

Principal payments required under our amended and restated credit facility term loan are: \$0.7 million due each calendar quarter through September 30, 2015; \$24.9 million due August 2, 2013 relative to term loan amounts not extended, and; \$211.7 million due on December 15, 2015.

Our credit facility is guaranteed on a senior secured basis by us and all of our domestic subsidiaries, other than our special purpose receivables subsidiary, on a joint and several basis. In addition, our obligations and the guarantees thereof are secured by substantially all the assets of us and the guaranters.

Our exposure to interest rate risk results from variable rates under our credit facility. Borrowings under our credit facility bear interest at various rates some of which are subject to a 2% LIBOR-floor.

At March 31, 2010, 1-Month LIBOR approximated 0.25%. Based on our variable rate-based borrowings outstanding at March 31, 2010, and after consideration of the 2% LIBOR-floor applicable to \$53.1 million of our supplemental revolving credit facility and \$225.2 million of our term loan, a 1% increase in the per annum interest rate for borrowings under our U.S. and foreign credit facilities would increase our interest expense by approximately \$0.3 million annually. The impact of a further decrease in LIBOR on our annual interest expense would not be material.

We have other cash commitments related to leases. We account for these lease transactions as operating leases and annual rent expense for continuing operations related thereto approximated 14.7 million. We expect to continue to utilize leasing as a financing strategy in the future to meet capital expenditure needs and to reduce debt levels.

In addition to rent expense from continuing operations, we also have approximately \$2.3 million in annual future lease obligations related to businesses that have been discontinued, of which approximately 61% relates to the facility for the former specialty laminates, jacketings and insulation tapes line of business (which extends through 2024), 33% relates to the Wood Dale facility in the former industrial fastening business (which extends through 2022), and 6% relates to the facility in our medical device line of business (which extends through 2012).

Market Risk

We conduct business in several locations throughout the world and are subject to market risk due to changes in the value of foreign currencies. We do not currently use derivative financial instruments to manage these risks. The functional currencies of our foreign subsidiaries are the local currency in the country of domicile. We manage these operating activities at the local level and revenues and costs are generally denominated in local currencies; however, results of operations and assets and liabilities reported in U.S. dollars will fluctuate with changes in exchange rates between such local currencies and the U.S. dollar.

Common Stock

We voluntarily transferred our stock exchange listing in the U.S. from The New York Stock Exchange to The NASDAQ Global Market® effective August 24, 2009. The company's stock continues to trade under the symbol "TRS."

Credit Rating

We and certain of our outstanding debt obligations are rated by Standard & Poor's and Moody's. On December 19, 2009, Moody's assigned a rating of Caa1 to our senior secured notes, improved the outlook from negative to stable, and affirmed our corporate family and senior secured credit rating at B3 and B1, respectively. On December 16, 2009, Standard & Poor's assigned a rating of B- to our senior secured notes, affirmed our credit facilities and corporate credit ratings of BB and B+ respectively, but maintained a negative outlook. If our credit ratings were to decline, our ability to access certain financial markets may become limited, the perception of us in the view of our customers, suppliers and security holders may worsen and as a result, we may be adversely affected.

Outlook

We believe the actions completed in 2009 under our Performance Improvement Plan position us for continued profitable future growth and place us in a better competitive position by enabling strategies focused on reduced cycle times and securing our position as best cost producer. Among our top priorities for 2010 are continuing to identify and execute on cost savings and productivity initiatives, to grow revenue via new products and expand our core products in non-U.S. markets, to continue to reduce our indebtedness and increase our available liquidity.

The combination of the savings realized from the Performance Improvement Plan, our ongoing productivity initiatives and the upturn in economic activity in the first quarter of 2010 has helped us to grow our earnings levels as compared to the first quarter of 2009. As there is still some degree of uncertainty in the markets that our businesses serve, there are a range of possible outcomes due to the uncertain financial markets environment, and we can offer no assurances that the economy will continue to improve.

Impact of New Accounting Standards

See Note 15, "New Accounting Pronouncements," included in Part I, Item 1, "Notes to Unaudited Consolidated Financial Statements," within this Form 10-Q.

Critical Accounting Policies

Certain of our accounting policies require the application of significant judgment by management in selecting the appropriate assumptions for calculating financial estimates. By their nature, these judgments are subject to an inherent degree of uncertainty. These judgments are based on our historical experience, our evaluation of business and macroeconomic trends, and information from other outside sources, as appropriate.

During the quarter ended March 31, 2010, there were no material changes to the items that we disclosed as our critical accounting policies in Part II, Item 7, "Management's Discussion and Analysis of Financial Condition and Results of Operations," in the Company's 2009 Annual Report on Form 10-K.

Item 3. Quantitative and Qualitative Disclosures About Market Risk

In the normal course of business, we are exposed to market risk associated with fluctuations in foreign currency exchange rates. We are also subject to interest risk as it relates to long-term debt. See Part I, Item 2, "Management's Discussion and Analysis of Financial Condition and Results of Operations,"

for details about our primary market risks, and the objectives and strategies used to manage these risks. Also see Note 8, "Long-term Debt," in Part I, Item 1, "Notes to Unaudited Consolidated Financial Statements," included within this Form 10-Q for additional information.

Item 4. Controls and Procedures

The Company maintains disclosure controls and procedures that are designed to ensure that information required to be disclosed in the reports that the Company files or submits under the Securities and Exchange Act of 1934, as amended (the "Exchange Act") is recorded, processed, summarized, and reported within the time periods specified in the SEC's rules and forms, and that such information is accumulated and communicated to the Company's management, including its Chief Executive Officer and Chief Financial Officer, as appropriate, to allow timely decisions regarding required disclosures.

Evaluation of disclosure controls and procedures

As of March 31, 2010, an evaluation was carried out by management, with the participation of our Chief Executive Officer and Chief Financial Officer, of the effectiveness of our disclosure controls and procedures (as such term is defined in Rule 13a-15(e) and Rule 15d-15(e) of the Securities Exchange Act of 1934, (the "Exchange Act")) pursuant to Rule 13a-15 of the Exchange Act. Our disclosure controls and procedures are designed only to provide reasonable assurance that they will meet their objectives. Based upon that evaluation, the Chief Executive Officer and Chief Financial Officer concluded that as of March 31, 2010, the Company's disclosure controls and procedures are effective to provide reasonable assurance that they would meet their objectives.

Changes in disclosure controls and procedures

There have been no changes in the Company's internal control over financial reporting during the quarter ended March 31, 2010 that have materially affected, or are reasonably likely to materially affect, the Company's internal control over financial reporting.

PART II. OTHER INFORMATION

TRIMAS CORPORATION

Item 1. Legal Proceedings

See Note 10, "Commitments and Contingencies," included in Part I, Item 1, "Notes to Unaudited Consolidated Financial Statements," within this Form 10-O.

Item 1A. Risk Factors

In addition to the other information set forth in this report, you should carefully consider the factors discussed in Part 1, Item 1A., "*Risk Factors*," in our Annual Report on Form 10-K for the year ended December 31, 2009, which could materially affect our business, financial condition or future results. There have been no significant changes in our risk factors as disclosed in our 2009 Form 10-K.

The risks described in our Annual Report on Form 10-K are not the only risks facing our Company. Additional risks and uncertainties not currently known to us or that we currently deemed to be immaterial also may materially adversely affect our business, financial position and results of operations or cash flows.

Item 2. Unregistered Sales of Equity Securities and Use of Proceeds.

None.

Item 3. Defaults Upon Senior Securities

Not applicable.

Item 4. Submission of Matters to a Vote of Security Holders

Reserved.

Item 5. Other Information

Not applicable.

Item 6. Exhibits.

Exhibits Index:

- 3.1(1) Fourth Amended and Restated Certificate of Incorporation of TriMas Corporation.
- 3.2(l) Second Amended and Restated By-laws of TriMas Corporation.
- 4.1(a) Indenture relating to the 9⁷/8% senior subordinated notes, dated as of June 6, 2002, by and among TriMas Corporation, each of the Guarantors named therein and The Bank of New York as Trustee, (including Form of Note as Exhibit).
- 4.2(c) Supplemental Indenture dated as of March 4, 2003.
- 4.3(d) Second Supplemental Indenture dated as of May 9, 2003.
- 4.4(e) Third Supplemental Indenture dated as of August 6, 2003.
- 4.5(p) Fourth Supplemental Indenture dated as of February 28, 2008.
- 4.6(ad) Fifth Supplemental Indenture dated as of January 26, 2009.
- 4.7(ac) Sixth Supplemental Indenture, dated as of December 29, 2009.

- 4.8(ac) Indenture relating to the $9^3/4\%$ senior secured notes dated as of December 29, 2009, among TriMas Corporation, the Guarantors named therein and The Bank of New York Mellon Trust Company, N.A., as Trustee.
- 10.1(a) Stock Purchase Agreement dated as of May 17, 2002 by and among Heartland Industrial Partners, L.P., TriMas Corporation and Metaldyne Company LLC.
- 10.2(a) Amended and Restated Shareholders Agreement, dated as of July 19, 2002 by and among TriMas Corporation and Metaldyne Corporation.
- 10.3(j) Amendment No. 1 to Amended and Restated Shareholders Agreement dated as of August 31, 2006.
- 10.4(i) Credit Agreement dated as of June 6, 2002, as amended and restated as of August 2, 2006 among TriMas Corporation, TriMas Company LLC, JPMorgan Chase Bank, N.A., as Administrative Agent and Collateral Agent, and Comerica Bank, as Syndication Agent.
- 10.5(ab) Credit Agreement dated as of June 6, 2002, as amended and restated as of August 2, 2006, as further amended and restated as of December 16, 2009, among TriMas Corporation, TriMas Company LLC, JPMorgan Chase Bank, N.A., as Administrative Agent and Collateral Agent, Comerica Bank, as Syndication Agent and J.P. Morgan Securities Inc., as Lead Arranger and Bookrunner.
- 10.6(ac) Credit Agreement dated as of June 6, 2002, as amended and restated as of August 2, 2006, as further amended and restated as of December 16, 2009, as further amended and restated as of January 13, 2010, among TriMas Corporation, TriMas Company LLC, JPMorgan Chase Bank, N.A., as Administrative Agent and Collateral Agent, Comerica Bank, as Syndication Agent, and J.P. Morgan Securities Inc., as Lead Arranger and Bookrunner.
- 10.7(a) Receivables Purchase Agreement, dated as of June 6, 2002, by and among TriMas Corporation, the Sellers party thereto and TSPC, Inc., as Purchaser.
- 10.8(w) Amendment No. 1 as of February 13, 2009 to Receivables Purchase Agreement.
- 10.9(a) Receivables Transfer Agreement, dated as of June 6, 2002, by and among TSPC, Inc., as Transferor, TriMas Corporation, individually, as Collection Agent, TriMas Company LLC, individually as Guarantor, the CP Conduit Purchasers, Committed Purchasers and Funding Agents party thereto, and JPMorgan Chase Bank as Administrative Agent.
- 10.10(k) Amendment dated as of June 3, 2005, to Receivables Transfer Agreement.
- 10.11(h) Amendment dated as of July 5, 2005, to Receivables Transfer Agreement.
- 10.12(n) Amendment dated as of December 31, 2007, to Receivables Transfer Agreement.
- 10.13(0) Amendment dated as of February 22, 2008, to Receivables Transfer Agreement.
- 10.14(w) Amendment dated as of February 13, 2009, to Receivables Transfer Agreement.
- 10.15(p) TriMas Receivables Facility Amended and Restated Fee Letter dated February 22, 2008.
- 10.16(w) TriMas Receivables Facility Amended and Restated Fee Letter dated February 13, 2009.
- 10.17^(ac) Amended and Restated Receivables Purchase Agreement, dated as of December 29, 2009, among TriMas Corporation, the Sellers named therein and TSPC, Inc. as Purchaser.
- 10.18(ac) Receivables Transfer Agreement, dated as of December 29, 2009, among TSPC, Inc., as Transferor, TriMas Corporation, as Collection Agent, TriMas Company LLC, as Guarantor, the persons party thereto from time to time as Purchasers and Wachovia Bank, National Association, as Administrative Agent.

- 10.19(a) Lease Assignment and Assumption Agreement, dated as of June 21, 2002, by and among Heartland Industrial Group, L.L.C., TriMas Company LLC and the Guarantors named therein.
- 10.20(a) TriMas Corporation 2002 Long Term Equity Incentive Plan.
- 10.21(t) First Amendment to the TriMas Corporation 2002 Long Term Equity Incentive Plan.
- 10.22(t) Second Amendment to the TriMas Corporation 2002 Long Term Equity Incentive Plan.
- 10.23(t) Third Amendment to the TriMas Corporation 2002 Long Term Equity Incentive Plan.
- 10.24(t) Fourth Amendment to the TriMas Corporation 2002 Long Term Equity Incentive Plan.
- 10.25(d) Asset Purchase Agreement among TriMas Corporation, Metaldyne Corporation and Metaldyne Company LLC dated May 9, 2003, (including Exhibit A—Form of Sublease Agreement).
- 10.26(f) 2003 Form of Stock Option Agreement.
- 10.27(s) 2008 Annual Value Creation Program.
- 10.28(t) 409A Amendment to TriMas Corporation Annual Value Creation Plan effective September 10, 2008.
- 10.29(g) Form of Indemnification Agreement.
- 10.30(j) Amendment No. 1 to Stock Purchase Agreement, dated as of August 31, 2006 by and among Heartland Industrial Partners, L.P., TriMas Corporation and Metaldyne Corporation.
- 10.31(s) Amendment No. 2 to Stock Purchase Agreement, dated as of November 27, 2006 by and among Heartland Industrial Partners, L.P., TriMas Corporation and Metaldyne Corporation.
- 10.32(j) Advisory Agreement, dated June 6, 2002 between Heartland Industrial Partners, L.P. and TriMas Corporation.
- 10.33(k) First Amendment to Advisory Agreement, dated as of November 1, 2006 between Heartland Industrial Group, L.L.C. and TriMas Corporation.
- 10.34(k) Second Amendment to Advisory Agreement, dated as of November 1, 2006 between Heartland Industrial Group, L.L.C. and TriMas Corporation.
- 10.35(k) Management Rights Agreement.
- 10.36(aa) Executive Severance/Change of Control Policy.
- 10.37(ag) TriMas Corporation 2006 Long Term Equity Incentive Plan Composite Plan Document.
- 10.38(q) Separation Agreement dated April 10, 2008.
- 10.39(r) Letter Agreement dated April 28, 2008.
- 10.40(s) Letter Agreement dated July 1, 2008.
- 10.41(z) ISDA 2002 Master Agreement between JPMorgan Chase Bank, N. A. and TriMas Company LLC dated as of January 29, 2009.
- 10.42(t) Interest Rate Swap Transaction letter Agreement between JPMorgan Chase Bank, N.A. and TriMas Company, LLC effective as of April 29, 2008.
- 10.43(ad) Interest Rate Swap Transaction letter Agreement between JPMorgan Chase Bank, N.A. and TriMas Company, LLC effective as of January 28, 2009.

- 10.44(ad) Interest Rate Swap Transaction letter Agreement between JPMorgan Chase Bank, N.A. and TriMas Company, LLC effective as of October 28, 2009.
- 10.45(w) Asset Purchase Agreement between Lamtec Corporation, Compac Corporation and TriMas Company LLC dated as of December 8, 2008.
- 10.46(u) Offer Letter from TriMas Corporation to David M. Wathen dated as of January 12, 2009.
- 10.47(v) Separation Agreement dated as of January 13, 2009.
- 10.48(y) Separation Agreement dated as of March 5, 2009.
- 10.49(x) TriMas Corporation Long Term Equity Incentive Plan Non-Qualified Stock Option Agreement.
- 10.50(y) 2009 TriMas Incentive Compensation Plan.
- 10.51(af) 2010 TriMas Incentive Compensation Plan.
- 10.52(aa) Flexible Cash Allowance Policy.
- 10.53(ad) TriMas Corporation 2006 Long Term Equity Incentive Plan Restricted Stock Agreement—2009 Additional Grant.
- 10.54(ad) TriMas Corporation 2006 Long Term Equity Incentive Plan Restricted Stock Agreement—2009 162(m) Conversion Grant.
- 10.55(ad) TriMas Corporation 2002 Long Term Equity Incentive Plan Restricted Stock Agreement—2009 Conversion and Additional Grants.
- 10.56(ae) TriMas Corporation 2002 Long Term Equity Incentive Plan Non-Qualified Stock Option Agreement.
- 10.57(ae) TriMas Corporation 2002 Long Term Equity Incentive Plan Restricted Stock Agreement.
- 10.58(ae) TriMas Corporation 2006 Long Term Equity Incentive Plan Restricted Stock Unit Agreement.
- 10.59 Asset Purchase Agreement among TW Cylinders LLC, Taylor-Wharton International LLC and Norris Cylinder Company dated as of April 30, 2010.
- 31.1 Certification pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 302 of the Sarbanes-Oxley Act of 2002.
- 31.2 Certification pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 302 of the Sarbanes-Oxley Act of 2002.
- 32.1 Certification pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of
- 32.2 Certification pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002.

⁽a) Incorporated by reference to the Exhibits filed with our Registration Statement on Form S-4, filed on October 4, 2002 (File No. 333-100351).

⁽b) Incorporated by reference to the Exhibits filed with Amendment No. 2 to our Registration Statement on Form S-4, filed on January 28, 2003 (File No. 333-100351).

⁽c) Incorporated by reference to the Exhibits filed with our Annual Report on Form 10-K filed March 31, 2003 (File No. 333-100351).

- (d) Incorporated by reference to the Exhibits filed with our Registration Statement on Form S-4, filed June 9, 2003 (File No. 333-105950).
- (e) Incorporated by reference to the Exhibits filed with our Quarterly Report on Form 10-Q filed on August 14, 2003 (File No. 333-100351).
- (f) Incorporated by reference to the Exhibits filed with our Quarterly Report on Form 10-Q filed on November 12, 2003 (File No. 333-100351).
- (g) Incorporated by reference to the Exhibits filed with Amendment No. 3 to our Registration Statement on Form S-1/A, filed on June 29, 2004 (File No. 333-113917).
- (h) Incorporated by reference to the Exhibits filed with our Report on Form 8-K filed on July 6, 2005 (File No. 333-100351).
- (i) Incorporated by reference to the Exhibits filed with our Report on Form 8-K filed on August 3, 2006 (File No. 333-100351).
- (j) Incorporated by reference to the Exhibits filed with Amendment No. 1 to our Registration Statement on Form S-1, filed on September 19, 2006 (File No. 333-136263).
- (k) Incorporated by reference to the Exhibits filed with Amendment No. 3 to our Registration Statement on Form S-1, filed on January 18, 2007 (File No. 333-136263).
- (l) Incorporated by reference to the Exhibits filed with our Quarterly Report on Form 10-Q, filed on August 3, 2007 (File No. 333-100351).
- (m) Incorporated by reference to the Exhibits filed with the Registration Statement on Form S-8, filed on August 31, 2007 (File No. 333-145815).
- (n) Incorporated by reference to the Exhibits filed with our Report on Form 8-K filed on January 4, 2008 (File No. 001-10716).
- (o) Incorporated by reference to the Exhibits filed with our Report on Form 8-K filed on February 26, 2008 (File No. 001-10716).
- (p) Incorporated by reference to the Exhibits filed with our Annual Report on Form 10-K filed on March 13, 2008 (File No. 001-10716).
- (q) Incorporated by reference to the Exhibits filed with our Report on Form 8-K filed on April 10, 2008 (File No. 001-10716).
- (r) Incorporated by reference to the Exhibits filed with our Report on Form 8-K filed on June 2, 2008 (File No. 001-10716).
- (s) Incorporated by reference to the Exhibits filed with our Quarterly Report on Form 10-Q filed on August 7, 2008 (File No. 001-10716).
- (t) Incorporated by reference to the Exhibits filed with our Quarterly Report on Form 10-Q filed on November 10, 2008 (File No. 001-10716).
- (u) Incorporated by reference to the Exhibits filed with our Report on Form 8-K filed on January 14, 2009 (File No. 001-10716).
- (v) Incorporated by reference to the Exhibits filed with our Report on Form 8-K filed on February 5, 2009 (File No. 001-10716).
- (w) Incorporated by reference to the Exhibits filed with our Report on Form 8-K filed on February 17, 2009 (File No. 001-10716).

(x)	Incorporated by reference to the Exh	ibits filed with our Report on Form	8-K filed on March 6, 2009 (File No.	001-10716).

- (y) Incorporated by reference to the Exhibits filed with our Report on Form 8-K filed on March 10, 2009 (File No. 001-10716).
- (z) Incorporated by reference to the Exhibits filed with our Annual Report on Form 10-K filed on March 10, 2009 (File No. 001-10716).
- (aa) Incorporated by reference to the Exhibits filed with our Report on Form 8-K filed on December 10, 2009 (File No. 001-10716).
- (ab) Incorporated by reference to the Exhibits filed with our Report on Form 8-K filed on December 17, 2009 (File No. 001-10716).
- (ac) Incorporated by reference to the Exhibits filed with our Report on Form 8-K filed on January 15, 2010 (File No. 001-10716).
- (ad) Incorporated by reference to the Exhibits filed with our Annual Report on Form 10-K filed on March 4, 2010 (File No. 001-10716).
- (ae) Incorporated by reference to the Exhibits filed with our Report on Form 8-K filed on March 4, 2010 (File No. 001-10716).
- (af) Incorporated by reference to the Exhibits filed with our Report on Form 8-K filed on March 15, 2010 (File No. 001-10716).
- (ag) Incorporated by reference to the Exhibits filed with our Report on Form 8-K filed on March 26, 2010 (File No. 001-10716).

Signature

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned thereunto duly authorized.

Date: April 30, 2010 By: A. Mark Zeffiro Chief Financial Officer 55

ASSET PURCHASE AGREEMENT

BY AND AMONG

TW CYLINDERS LLC,

TAYLOR-WHARTON INTERNATIONAL LLC (as to Sections 2.6, 3.1(b), 3.2(b), 3.3, 5.8, 5.18, and ARTICLE IX)

AND

NORRIS CYLINDER COMPANY

DATED AS OF

APRIL 30, 2010

	TABLE OF CONTENTS	
ARTICLE I. DI	EFINITIONS	1
1.1	Certain Defined Terms.	1
1.2	Other Defined Terms.	9
ARTICLE II. P	PURCHASE AND SALE	11
2.1	Purchase and Sale of the Sold Assets.	11
2.2	Excluded Assets.	13
2.3	Assumption of Liabilities.	14
2.4	Purchase Price.	17
2.5	Net Working Capital.	18
2.6	Escrow.	20
2.7	Allocation of Total Consideration.	21
2.8	The Closing.	21
2.9	Deliveries at the Closing.	22
2.10	Further Assurances.	26
ARTICLE III. I	REPRESENTATIONS AND WARRANTIES OF THE SELLER	26
3.1	Organization.	27
3.2	Authorization; Enforceability.	27
3.3	Membership Interests.	27
3.4	Financial Statements; Accounting Controls and Procedures.	28
3.5	No Approvals or Conflicts.	29
3.6	Compliance with Law; Permits.	29
3.7	Proceedings.	30
3.8	Tax Matters.	30
3.9	Employee Benefits.	31
3.10	Labor Relations.	31
3.11	Intellectual Property.	32
3.12		33
3.13	Environmental Matters.	35
3.14	Personal Property Assets.	36
3.15	Seller Owned Real Property.	36
3.16	No Brokers' or Other Fees.	37
3.17	Undisclosed Liabilities.	37
3.18	Customers and Suppliers.	38
3.19		38
3.20		39
3.21	1	40
3.22		40
3.23		40

i

	3.24	Employees; Employment Matters.	41
	3.25	Sufficiency of Sold Assets.	41
	3.26	Absence of Changes.	41
	3.27	Capital Expenditures.	42
	3.28	Disclosure.	42
	3.29	No Other Representations or Warranties.	43
ARTI	CLE IV. RE	PRESENTATIONS AND WARRANTIES OF THE BUYER	43
	4.1	Organization.	43
	4.2	Authorization; Enforceability.	44
	4.3	No Approvals or Conflicts.	44
	4.4	Proceedings.	44
	4.5	Compliance with Laws; Permits.	44
	4.6	Financing.	45
	4.7	No Brokers' or Other Fees.	45
	4.8	Condition of the Business.	45
	4.9	Solvency.	45
ARTI	CLE V. CO	VENANTS AND AGREEMENTS	46
	5.1	Conduct of Business Prior to the Closing.	46
	5.2	Access to Books and Records; Cooperation.	47
	5.3	Tax Matters: Cooperation; Preparation and Filing of Tax Returns; Transfer Taxes and other Tax Matters.	49
	5.4	Treatment of Payments.	52
	5.5	Employees; Employment Matters.	52
	5.6	Labor Matters.	53
	5.7	Contact With Customers and Suppliers.	53
	5.8	Non-Solicitation by Buyer and its Affiliates.	53
	5.9	Corporate Names.	53
	5.10	Further Actions.	54
	5.11	Elimination of Debt Obligations.	55
	5.12	Bulk Transfer Laws.	55
	5.13	Confidentiality.	55
	5.14	No Exclusivity.	56
	5.15	Non-Solicitation and Non-Competition by Seller and its Affiliates.	56
	5.16	Bankruptcy Conditions.	57
	5.17	Advice of Changes.	59
	5.18	Contracts Related to the Huntsville Business Held by TW International.	60
	5.19	Servicing Warranty Claims.	60
	5.20	Harrisburg Assets.	60
ARTI	CLE VI. CC	ONDITIONS TO THE SELLER'S OBLIGATIONS	61
	6.1	Representations and Warranties.	61
	6.2	Performance.	61
		ii	
	6.3	Governmental Orders.	61
ARTI	CLE VII. C	ONDITIONS TO THE BUYER'S OBLIGATIONS	61
	7.1	Representations and Warranties.	62
	7.2	Performance.	62
	7.3	Governmental Orders.	62
	7.4	Plan not Effective.	62
	7.5	Key Employee.	62
	7.6	No Material Adverse Effect.	62
	7.7	Assumption and Assignment of Sold Contracts.	63
	7.8	Approval of the Break-up Fee and Expense Reimbursement.	63
	7.9	Satisfaction with Certain Due Diligence Items.	63
ARTI	CLE VIII. T	ERMINATION	63
	8.1	Termination.	63
	8.2	Procedure and Effect of Termination.	64
ARTI	CLE IX. IN	DEMNIFICATION	65
	0.1	Indomnification by the College	CF
	9.1 9.2	Indemnification by the Seller. Indemnification by the Buyer.	65 66
	5.2	macminication by the buyer.	90

9.3	Exclusive Remedy.	67
9.4	Indemnification Calculations.	67
9.5	Survival.	68
9.6	Notice and Opportunity to Defend.	68
9.7	Additional Limitations.	69
9.8	Subrogation.	69
ARTICLE X. MIS	SCELLANEOUS	70
10.1	Fees and Expenses.	70
10.2	Governing Law; Jurisdiction.	70
10.3	Certain Interpretive Matters.	70
10.4	Amendment.	71
10.5	Assignment.	72
10.6	Waiver.	72
10.7	Notices.	72
10.8	Complete Agreement.	74
10.9	Counterparts.	74
10.10	Publicity.	74
10.11	Severability.	74
10.12	Third Parties.	75
10.13	Non-Recourse.	75
10.14	No Successor Liability.	75

iii

SCHEDULES

Schedule 1.1-3	Harrisburg Assets
Schedule 2.1(a)	Acquired Equipment - Huntsville
Schedule 2.1(b)(ii)	Consigned Inventory
Schedule 2.1(c)	Sold Contracts
Schedule 2.1(e)	Acquired Intellectual Property
Schedule 2.1(f)	Seller Owned Real Property
Schedule 2.1(g)	Motor Vehicles
Schedule 2.2(c)	Excluded Contracts
Schedule 2.9(a)(x)	Material Closing Condition Consents
Schedule 3.4	Financial Statements and Interim Financial Statements
Schedule 3.5	No Approvals or Conflicts
Schedule 3.6(i)	Compliance with Law
Schedule 3.6(ii)	Permits
Schedule 3.7	Proceedings
Schedule 3.8	Tax Matters
Schedule 3.9	Benefit Plans
Schedule 3.10	Labor Relations
Schedule 3.11	
Schedule 3.11(a)	Information Regarding Acquired Intellectual Property Material Contracts
Schedule 3.12(a)	Material Contracts Material Contracts not in Full Force and Effect
Schedule 3.12(c)	Environmental Reports; Environmental Matters
Schedule 3.14	
	Personal Property Assets
Schedule 3.15	Information Regarding Seller Owned Real Property
Schedule 3.15(b)	Proceedings Affecting Seller Owned Real Property
Schedule 3.17	Undisclosed Liabilities
Schedule 3.18	Customers and Suppliers
Schedule 3.19(a)	Product and Services Liability
Schedule 3.19(b)	Product Liability Claims
Schedule 3.19(c)	Rebate Obligations
Schedule 3.19(d)	No Pending Litigation
Schedule 3.20	Consigned Inventory
Schedule 3.21(a)	Acquired Accounts Receivable
Schedule 3.23	Related Party Transactions
Schedule 3.24(a)	Compliance with Laws Related to Employees and Employment Matters
Schedule 3.24(b)	WARN Act
Schedule 3.26	Absence of Changes
Schedule 3.27	Conditions of Sold Assets
Schedule 5.1	Exceptions to Covenants Regarding Conduct of Business Prior to the Closing
Schedule 5.3(h)	Jurisdictions where Seller files Tax Returns
Schedule 5.5(a)(i)	Certain Huntsville Employees
Schedule 5.5(a)(ii)	Certain Designated Employees of the Seller
Schedule 5.6	Labor Matters

TWI Loan Documents

Reference Working Capital Calculation

Schedule 1.1—1 Schedule 1.1—2

EXHIBITS

Δ	Transition	Services	Agreement
Λ	Hansinon	Del vices i	zereemem.

B Products
1.1-A Agreed Principles
2.6 Escrow Agreement

2.9(a)(ii) Bill of Sale

2.9(a)(iii) Assignment and Assumption Agreement
 2.9(a)(xiv) Amendment to Sherwood Supply Agreement
 3.13 Buyer's Phase I Environmental Site Assessment

5.2(d) Scope of Phase II Review 5.2(e) Insurance Requirements 5.16(a)(i) Bidding Procedures 5.16(a)(iii) Sale Order

v

ASSET PURCHASE AGREEMENT

This ASSET PURCHASE AGREEMENT (this "<u>Agreement</u>"), dated as of April 30, 2010, is by and among TW CYLINDERS LLC, a Delaware limited liability company (the "<u>Seller</u>"), TAYLOR-WHARTON INTERNATIONAL LLC, a Delaware limited liability company ("<u>TW International</u>") (as to Sections 2.6, 3.1(b), 3.2(b), 3.3, 5.8, 5.18 and ARTICLE IX), and NORRIS CYLINDER COMPANY, a Delaware corporation (the "<u>Buyer</u>").

RECITALS

WHEREAS, the Seller is engaged in the business of, among other things, the engineering, design, manufacture, marketing and sale of high and low pressure cylinders (the "Business"); and

WHEREAS, the Seller desires to sell, and the Buyer desires to purchase, (i) substantially all of the assets of the Seller that relate to the Huntsville Business (as defined below) and (ii) the Harrisburg Assets (as defined below), on the terms and subject to the conditions set forth in this Agreement; and

WHEREAS, on November 18, 2009 Seller and certain of its Affiliates (as defined below) filed for protection under the Bankruptcy Code (as defined below) in the Bankruptcy Court (as defined below); and

WHEREAS, on December 4, 2009, Seller and certain of its Affiliates filed their Joint Plan of Reorganization dated as of December 4, 2009 (the "Plan"), which in furtherance of such Plan provides for the sale of the Business; and

WHEREAS, the Seller and the Buyer desire, unless the Effective Date of the Plan (as defined below) occurs before the Closing, to effect the transactions contemplated by this Agreement in the manner and subject to the terms and conditions set forth in this Agreement and, the Sale Order (as defined below) and in accordance with Sections 105, 363, 365 and 1146 of the Bankruptcy Code (as defined below).

NOW, THEREFORE, in consideration of the premises and the representations, warranties, covenants and agreements contained in this Agreement and intending to be legally bound hereby, the parties hereto hereby agree as follows:

ARTICLE I. DEFINITIONS

1.1 <u>Certain Defined Terms.</u>

As used in this Agreement, the following terms shall have the following meanings:

"Affiliate" shall mean, with respect to any specified Person, any other Person that directly, or indirectly through one or more intermediaries, Controls, is Controlled by, or is under common Control with, such specified Person.

1

"Agreed Principles" means GAAP except that the value of inventory is determined in accordance with the provisions of Exhibit 1.1-A.

"Ancillary Agreements" shall mean (a) with respect to the Sold Assets, such Deeds, bills of sale, endorsements, assignments, affidavits and other instruments of sale, conveyance, transfer and assignment from the Seller, in form and substance reasonably satisfactory to the Seller and the Buyer, as shall be necessary under Law or contemplated by this Agreement in order to transfer to the Buyer all right, title and interest in and to the Sold Assets in accordance with the terms hereof, (b) with respect to the Assumed Liabilities, such instruments of assumption, in form and substance reasonably satisfactory to the Seller and the Buyer, as shall be necessary under Law or contemplated by this Agreement in order for the Assumed Liabilities to be effectively assumed by the Buyer and (c) the Transition Services Agreement.

"Avoidance Actions" means any and all actions which a trustee or debtor-in-possession or other appropriate party in interest may assert under applicable state statutes or Chapter 5 of the Bankruptcy Code, including actions under one or more provisions of Sections 542, 543, 544, 545, 546, 547, 548, 549, 550, 551 and 553 of the Bankruptcy Code.

"<u>Bankruptcy Case</u>" means the Jointly Administered bankruptcy case of Taylor-Wharton International LLC and its affiliated debtors, Case No. 09-14089 (BLS), pending in the United States Bankruptcy Court for the District of Delaware.

"Bankruptcy Code" shall mean Title 11 of the United States Code, as amended.

"Bankruptcy Court" shall mean the United States Bankruptcy Court for the District of Delaware.

"Business Day" shall mean any day that is not a Saturday, a Sunday or any other day on which banks are required or authorized by Law to be closed in the city of New York, New York, United States.

"Business Real Property." shall mean, collectively, the Seller Leased Real Property and the Seller Owned Real Property.

"Cash" shall mean the sum of cash and cash equivalents, plus all uncollected bank deposits and less all outstanding checks and other draws and drafts (including overdrafts) as of the applicable measurement date (it being understood that "Cash" can be a negative number).

"Change of Control Payments" shall mean any payment in the nature of compensation that becomes payable to any Person by the Seller or any of its Affiliates as a result of, or in connection with, the transactions contemplated by this Agreement, including stay bonuses, sale or transaction bonuses, or similar change of control payments.

"COBRA" means the requirements of Code Section 4980B and Part 6 of Subtitle B of Title I of ERISA and any similar state law legal requirement.

2

"Code" shall mean the Internal Revenue Code of 1986, as amended.

"Competition/Foreign Investment Law" shall mean any Law that prohibits, restricts or regulates (a) foreign investment or (b) antitrust, monopolization, restraint of trade or competition.

"Confidentiality Agreement" shall mean the confidentiality agreement executed by TriMas Corporation and Taylor-Wharton International LLC dated February 5, 2009, as amended by that certain amendment to add Windpoint as a party thereto effective March 15, 2010.

"Control" (including the terms "controlled by" and "under common control with"), with respect to the relationship between or among two or more Persons, shall mean the possession, directly or indirectly, of the power to direct or cause the direction of the affairs or management of a Person, whether through the ownership of voting securities, by contract or otherwise, including the ownership, directly or indirectly, of securities having the power to elect a majority of the board of directors or similar body governing the affairs of such Person.

"Debt Obligations" shall mean, with respect to any Person as of any date without duplication, (a) all indebtedness of such Person for borrowed money, whether or not current, short-term or long-term, secured or unsecured, (b) all indebtedness of such Person for the deferred purchase price of property or services, including obligations represented by a note, earnout or contingent purchase payment agreement, (c) all indebtedness of such Person created or arising under any conditional sale or other title retention agreement with respect to property acquired by such Person (even though the rights and remedies of the seller or lender under such agreement in the event of default are limited to repossession or sale of such property), (d) all indebtedness of such Person secured by a purchase money mortgage or other lien to secure all or part of the purchase price of the property subject to such mortgage or lien, (e) all lease obligations of such Person under leases that are capital leases in accordance with GAAP, (f) all credit extended on behalf of such Person in respect of banker's acceptances and letters of credit (other than stand-by letters of credit in support of ordinary course trade payables), (g) all liabilities of such Person with respect to interest rate swaps, collars, caps and similar hedging obligations, (h) any accrued and unpaid interest, fees and other expenses on any of the foregoing, and (i) all indebtedness referred to in clauses (a) through (h) above of any Person other than such Person that is either guaranteed by, or secured by an Encumbrance upon any property owned by, such Person.

"<u>Deed</u>" shall mean the special or limited warranty deed with covenants against grantor's acts conveying the Seller Owned Real Property to the Buyer, subject only to Permitted Encumbrances.

"<u>Disclosure Schedules</u>" shall mean the schedules delivered by the Seller prior to or concurrently with the execution and delivery of this Agreement.

3

"<u>Due Date</u>" shall mean the due date with respect to any applicable Tax Return (taking into account valid extensions).

"<u>Duty</u>" shall mean any stamp, transaction or registration, duty or similar charge imposed by any Governmental Authority, including any interest, fine, penalty, charge or other amount imposed in respect thereof.

"Effective Date of the Plan" shall mean the Effective Date (as defined in the Plan).

"Encumbrance" shall mean any security interest, pledge, mortgage, lien, transfer restriction, lease, charge, option, easement, claim or right of first refusal.

"Environment" shall mean soil, surface water, groundwater, stream sediment, surface or subsurface strata and ambient air.

"Environmental Claim" shall mean any written notice, claim, demand, action, suit, complaint or proceeding by any Person alleging any actual or potential liability or violation under any Environmental Law.

"Environmental Laws" shall mean all Laws relating to the protection of human health, the environment, and/or regulating the use, presence, testing, storage, treatment, generation, transportation, processing, handling, production, removal or disposal of Hazardous Materials and the enforceable policies, decisions, orders of federal, state and local governmental agencies and courts with respect thereto, including the Comprehensive Environmental Response, Compensation, and Liability Act, 42 U.S.C. §9601 et seq., the Clean Air Act, 42 U.S.C. §401 et seq., the Clean Water Act, 33 U.S.C. §1251 et seq., the Hazardous Materials Transportation Act, 40 U.S.C. §1801 et seq., the Resource Conservation and Recovery Act, 42 U.S.C. §6901 et seq., and the Toxic Substances Control Act, 15 U.S.C. §2601 et seq., all as amended.

"ERISA" shall mean the Employee Retirement Income Security Act of 1974, as amended, and the rules and regulations promulgated thereunder.

"ERISA Affiliate" shall mean any Person that, together with the Seller, is or was at any time treated as a single employer under Section 414 of the Code or Section 4001 of ERISA and any general partnership of which the Seller is or has been a general partner.

"GAAP" shall mean United States generally accepted accounting principles and practices.

"General Enforceability Exceptions" shall mean the effects of bankruptcy, insolvency, fraudulent conveyance, reorganization, moratorium and other similar Laws relating to or affecting creditors' rights generally and general equitable principles (whether considered in a proceeding in equity or at Law).

4

"Governmental Authority" shall mean any government, agency, governmental department, commission, board, bureau, court, arbitration panel or instrumentality of the United States of America or any foreign government or any state, municipality or other political subdivision in or of any of the foregoing (whether now or hereafter constituted and/or existing) and any court, agency, instrumentality, regulatory commission or other entity exercising executive, legislative, judicial, regulatory or administrative functions of or pertaining to the federal government of the United States of America or any foreign government or any state, municipality or other political subdivision in or of any of the foregoing.

"<u>Governmental Order</u>" shall mean any order, writ, injunction, decree, judgment, assessment or arbitration award of a Governmental Authority.

"Harrisburg Assets" shall mean all of the assets listed on Schedule 1.1-3.

"<u>Harrisburg Holdback</u>" shall mean an amount equal to \$375,000 which will be held and paid in accordance with the provisions of Section 2.9(c).

"<u>Hazardous Material</u>" shall mean, without limitation, any flammable material, radon, radioactive materials, asbestos, urea formaldehyde foam insulation, polychlorinated biphenyls, petroleum and petroleum-based materials, methane, pollutants, hazardous materials, hazardous wastes, hazardous or toxic substances and any other substances regulated under any Environmental Law.

"Huntsville Business" means all of the business conducted at the Seller Owned Real Property and the business of designing, manufacturing, repairing, selling and distributing the Products as conducted by Seller anywhere in the world.

"Income Taxes" shall mean (a) all Taxes based upon, measured by, or calculated with respect to (i) net income or profits (including Texas margins tax, Michigan business tax, Michigan single business tax and any capital gains or minimum Tax, but not including any sales, use, real or personal property, transfer or similar Taxes) or (ii) multiple bases (including, but not limited to, corporate franchise or doing business) if one or more Taxes upon which such Tax may be based, measured by or calculated with respect to, is described in clause (a)(i) above; and (b) all U.S., state, local and foreign franchise Taxes, including, in the case of each of clauses (a) and (b), any related interest and penalties, additions to such Tax or additional amounts imposed with respect thereto by any Governmental Authority.

"Intellectual Property." shall mean any and all: (a) inventions (whether patentable or unpatentable and whether or not reduced to practice), all improvements thereto, and all patents, patent applications and patent invention disclosures, and all other rights of inventorship together with all reissuances, continuations, continuations-in-part, divisions, revisions, extensions and re-examinations thereof; (b) registered and unregistered trademarks, trade names, trade dress, brand names, logos, slogans and Internet domain names and their associated goodwill and all registrations thereof and applications therefor; (c) copyrights and copyrightable works and all other rights of authorship recognized by statute or otherwise (including software, source code, object code, databases, schematics, flowcharts and related items) and all

_

applications, registrations and renewals in connection therewith; (d) trade secrets, ideas, processes, formulae, compositions, technology, manufacturing and production processes and techniques, technical data, engineering production and other designs, engineering notebooks, industrial models, discoveries, knowhow, specifications, designs, plans, manuals, drawings, research, financial, marketing and business data, pricing and cost information, business and marketing plans, customer and supplier lists and information and all other confidential or proprietary information; (e) rights to sue for and remedies against past, present and future infringements of any or all of the foregoing and rights of priority and protection of interests therein under the Laws of any jurisdiction; (f) copies and tangible embodiments of all of the foregoing; and (g) all other proprietary, intellectual property or other rights relating to any of the foregoing.

"Knowledge of the Buyer" shall mean the actual knowledge of the following individuals: Jerry Van Auken and Joshua Sherbin. It is understood that any input by such individuals under this Agreement is only in their capacities as employees or officers (as applicable) of the Buyer and not in any personal capacity.

"Knowledge of the Seller" shall mean the actual knowledge of the following individuals: Stuart Jara, Bill Corbin, Mike Camp, Mark Erb, Peter Smith, Robert Gerkins and Leonard York. It is understood that any input by such individuals under this Agreement is only in their capacities as employees or officers (as applicable) of the Seller and/or TW International and not in any personal capacity.

"<u>Law</u>" shall mean any statute, law, ordinance, regulation or rule of any Governmental Authority and binding orders, judgments, decrees, common law or any other provisions having the force or effect of law.

"<u>Liabilities</u>" shall mean any debt, liability, obligation, deficiency, penalty, assessment, fine, claim, cause of action or other loss (whether direct or indirect, absolute or contingent, accrued or unaccrued, liquidated or unliquidated, asserted or unasserted, or due or to become due), including all costs and expenses relating thereto.

"Material Adverse Effect" shall mean any material adverse effect on the business, results of operations, properties, assets, Liabilities, or financial condition of the Huntsville Business, taken as a whole, but shall exclude any effect (a) resulting from general economic conditions, (b) affecting companies in the gas technologies industry generally, (c) resulting from the announcement or performance of this Agreement or the transactions contemplated hereby, or (d) resulting from any changes in applicable Laws or accounting rules.

"Membership Interests" shall mean all of the membership interests of the Seller.

"Permits" shall mean any permits, licenses, certificates, approvals and authorizations of any Governmental Authority and any industry certifications.

"<u>Permitted Encumbrances</u>" shall mean (a) statutory Encumbrances for current Taxes not yet due and payable, (b) Encumbrances in respect of property or assets imposed by

6

Law that were incurred in the ordinary course of business, such as carriers', warehousemen's, materialmen's and mechanics' liens and other similar liens with respect to amounts which are not yet due and owing, and (c) easements, restrictions, covenants or similar matters relating to real property; provided that none of the foregoing described in clause (c) above do or will, individually or in the aggregate, materially impair the value of the property to which they relate or the continued use and operation thereof in the Business as conducted by the Seller since December 7, 2007.

"Person" shall mean any individual, partnership, firm, corporation, association, trust, unincorporated organization, joint venture, limited liability company, Governmental Authority or other entity.

"Pre-Closing Tax Period" shall mean a taxable period (or portion of any Straddle Period) that ends before the Closing Date.

"<u>Proceeding</u>" shall mean any judicial, administrative or arbitral actions, suits or proceedings (public or private) by or before any Governmental Authority or before any arbitrator, mediator or other alternative dispute resolution provider pursuant to any collective bargaining agreement, contractual agreement or Law, and including any audit or examination, or other administrative or court proceeding with respect to Taxes or Tax Returns.

"Products" shall mean any and all products manufactured at the Seller Owned Real Property and all products listed on Exhibit B.

"Reference Working Capital" shall mean \$4,056,000, as determined pursuant to Schedule 1.1—2.

"Release" shall have the meaning provided in 42 U.S.C. Section 9601(22).

"<u>Seller Leased Real Property</u>" shall mean the real property that is the subject of that certain Lease Agreement, dated December 7, 2007, by and between Harsco Corporation and TW International, as amended.

"Straddle Period" shall mean any taxable period that includes but does not end on the Closing Date.

"Subsidiary" shall mean, with respect to any Person, any other Person 50% or more of the voting equity of which is owned, directly or indirectly, by such first Person.

"Taxe" or "Taxes" shall mean all (a) taxes, charges, withholdings, fees, levies, imposts, Duties and governmental fees or other like assessments or charges of any kind whatsoever, whether computed on a separate or consolidated, unitary or combined basis or in any other manner, whether disputed or not, imposed by any United States federal, state, local or foreign or other Governmental Authority (including those related to income, net income, gross income, receipts, commercial activities, Texas margins tax, capital, windfall profit, severance, property (real and personal), production, sales, goods and services, use, business and occupation,

7

Michigan business tax, Michigan single business tax, license, excise, registration, franchise, employment, payroll (including social security contributions), deductions at source, withholding, alternative or add-on minimum, intangibles, ad valorem, transfer, gains, stamp, customs, duties, estimated, transaction, title, capital, paid-up capital, profits, premium, value added, recording, inventory and merchandise, business privilege, federal highway use, commercial rent or environmental tax, and any liability under unclaimed property, escheat, or similar Laws), (b) interest, penalties, fines, additions to tax or additional amounts imposed by any Governmental Authority in connection with (i) any item described in clause (a) or (ii) the failure to comply with any requirement imposed with respect to any Tax Return, and (c) liability in respect of any items described in clause (a) and/or (b) payable by reason of contract (including any Tax Sharing Agreement), assumption, transferee, successor or similar liability, operation of law (including pursuant to Treasury Regulations Section 1.1502-6 (or any predecessor or successor thereof or any analogous or similar state, local, or foreign Law)) and including any obligation to indemnify or otherwise assume or succeed to the Tax Liability of any other Person.

"<u>Tax Return</u>" shall mean any return, declaration, form, report, claim, informational return (including all Forms 1099) or statement required by any Governmental Authority to be filed with respect to Taxes, including any schedule or attachment thereto or amendment thereof.

"<u>Tax Sharing Agreement</u>" shall mean any Tax indemnity agreement, Tax sharing agreement, Tax allocation agreement or similar contract or arrangement, whether written or unwritten.

"<u>Taxing Authority</u>" shall mean, with respect to any Tax or Tax Return, the Governmental Authority that imposes such Tax or requires a Person to file such Tax Return and the agency (if any) charged with the collection of such Tax or the administration of such Tax Return, in each case, for such Governmental Authority.

"Third Party Expenses" shall mean all fees and expenses incurred by the Seller in connection with the negotiation and effectuation of the terms and conditions of this Agreement and the transactions contemplated hereby, including all legal, accounting, financial advisory, consulting and all other fees and expenses of third parties in connection therewith.

"<u>Transition Services Agreement</u>" shall mean a transition services agreement, dated as of the Closing Date, to be entered into by TW International and the Buyer, substantially in the form of <u>Exhibit A</u>.

"<u>Treasury Regulations</u>" shall mean the Treasury regulations promulgated under the Code, as such Treasury Regulations may be amended from time to time. Any reference herein to a particular provision of the Treasury Regulations means, where appropriate, the corresponding successor provision.

"TWI Holdings" shall mean Taylor-Wharton Intermediate Holdings LLC, a Delaware limited liability company.

8

"TW International" shall mean Taylor-Wharton International LLC, a Delaware limited liability company.

"TWI Loan Documents" shall mean the Credit Agreement, dated December 7, 2007, among the Seller, TW International, TW Cryogenics LLC, Sherwood Valve LLC, American Welding & Tank LLC, Structural Composites Industries LLC and General Electric Capital Corporation, the agreements ancillary thereto to which the Seller is a party and the other agreements listed on Schedule 1.1.—1.

"United States" or "U.S." shall mean the United States of America.

1.2 Other Defined Terms.

The following terms shall have the meanings defined for such terms in the Sections set forth below:

<u>Term</u>	Section
Acquired Accounts Receivable	3.21(a)
Acquired Intellectual Property	2.1(e)
Acquired Inventory	2.1(b)
Acquisition Transaction	5.14
Agreement	Preamble
Allocation Schedule	2.6(a)
Alternate Transaction	5.16(b)(ii)
Assumed Liabilities	2.3(a)
Auction	5.16(b)(ii)
Basket	9.1
Benefit Plans	3.9
Bidding Procedures	5.16
Break-Up Fee	5.16
Business	Recitals
Buyer	Preamble
Buyer Indemnified Persons	9.1
Cap	9.1
Claim Notice	9.6(a)
Closing	2.8(a)
Closing Balance Sheet	2.5(b)
Closing Date	2.8
Closing NWC Adjustment	2.4
Closing Purchase Price	2.4
Contracts	3.12(a)
CPA Firm	2.5(c)
Cure Costs	2.3(c)
Customers	3.18
Equipment	2.1(a)
Escrow Agreement	2.6

Escrow Amount	2.6
Estimated Net Working Capital	2.5(a)
Excluded Assets	2.2
Excluded Liabilities	2.3(a)(ii)
Expense Reimbursement	5.16
Final Allocation Schedule	2.6(c)
Final Net Working Capital	2.5(b)
Final NWC Adjustment	2.5(d)
Final Statement of Net Working Capital	2.5(c)
Financial Statements	3.4
Huntsville Business	Recitals
Huntsville Employee	5.5(a)
Indemnification Acknowledgement	9.6(a)
Indemnified Party	9.6(a)
Indemnifying Party	9.6(a)
Indemnity Limitations	9.1
Interim Financial Statements	3.4
Inventory	2.1(b)
Losses	9.1
Material Contracts	3.12(a)
Net Working Capital	2.5(a)
Net Working Capital Objection	2.5(a)
Net Working Capital Statement	2.5(b)
NWC True-Up	2.5(d)
Objections Notice Periodic Taxes	2.6(a)
Phase II Review	5.3(d) 5.2(d)
Potential Bidder	
Plan	5.16(b)(i) Recitals
Plant	
Prime Rate	2.9(c) 2.5(d)
Product Liability Claims	2.5(d) 3.19(b)
Prohibited Parties	5.15(b) 5.15(a)
Purchase Price	2.4
Rebate Obligations	3.19(c)
Related Party	3.13(c) 3.23
Reports	3.13(e)
Response Period	2.6(a)
Sale Hearing	5.16(a)
Sale Order	5.16(a)
Sale Motion	5.16(a)
Sale Motion Filing Date	5.16(a)
Seller	Preamble
Seller Indemnified Persons	
Seller Owned Real Property	9.2 2.1(f)
ca rear reperty	(-)

10

Sold Assets	2.1
Sold Contracts	2.1(c)
Special Warranties	9.5
Statement of Assets and Liabilities	3.4
Suppliers	3.18
Tax Controversy	9.6(b)
Taxing Jurisdiction	5.3(h)
Third Party Claim	9.6(a)
Title Insurer	2.9(a)(i)(A)(1)
Total Consideration	2.4
Transaction Financing	4.6
Transferred Employees	5.5(a)
Transfer Taxes	5.3(a)
WARN Act	3.24(b)

ARTICLE II. PURCHASE AND SALE

2.1 <u>Purchase and Sale of the Sold Assets.</u>

On the Closing Date and subject to the terms and conditions set forth in this Agreement, the Seller shall sell, assign, transfer, convey and deliver to the Buyer, and the Buyer shall purchase and acquire from the Seller, all of the right, title and interest in and to the Sold Assets. The term "Sold Assets" means: (i) the Harrisburg Assets, and (ii) all properties, assets and rights of every nature, kind and description, tangible or intangible, whether real, personal or mixed, whether or not reflected on the books and records of the Seller and whether now existing or hereafter acquired, used in or held for use by Seller in connection with, or related to the Huntsville Business, or arising out of the Huntsville Business, or located at the Seller Owned Real Property, including the assets and rights listed below (excluding only the Excluded Assets):

- (a) all machinery, equipment, spare parts, computer hardware, computer software, tools, office equipment, business machines, furniture, furnishings and other tangible personal property (the "<u>Equipment</u>") of the Seller located at the Seller Owned Real Property, including the Equipment listed on <u>Schedule 2.1(a)</u>;
- (b) all inventories of raw material, work in progress, finished goods, spare parts, replacement and component parts, packaging, office and other supplies and inventory (the "Inventory") of the Seller held by the Seller at the Seller Owned Real Property, or used or held for use in connection with the Huntsville Business (excluding any Inventory manufactured by Seller or its Affiliates in whole or in part at any location other than the Seller Owned Real Property) and (ii) the Inventory of the Seller held by a third party on consignment, that is listed on Schedule 2.1(b)(ii) (the Inventory identified in subparts (i) and (ii) of this Section 2.1(b), collectively, the "Acquired Inventory");

11

- (c) all rights and incidents of interest of, and benefits accruing to, the Seller in and to (i) the contracts and agreements set forth on Schedule 2.1(c), (ii) open sales orders or other contracts for the sale of Products or services with respect to the Huntsville Business, which Products or services have not been delivered, whether or not set forth on Schedule 2.1(c), or (iii) any open purchase orders or other contracts made in the ordinary course of business, consistent with past practice, for the acquisition of materials by the Seller with respect to the Huntsville Business (collectively, the "Sold Contracts");
- (d) all rights and benefits of the credits, prepaid expenses, deferred charges, advance payments, security deposits and prepaid items of the Seller with respect to or arising out of the Huntsville Business;
- (e) all Seller's or TWI's right, title and interest in and to the Intellectual Property used in, held for use in, or arising out of, the Huntsville Business including the Intellectual Property set forth on Schedule 2.1(e) (the "Acquired Intellectual Property");
- (f) all real property set forth on Schedule 2.1(f), including the buildings, structures, fixtures and improvements located thereon together with (i) any and all rights, title and interest of the Seller in and to air, mineral, subsurface and riparian rights and all tenements, easements, hereditaments, privileges and appurtenances appertaining thereto, (ii) any and all right, title and interest of the Seller in and to adjacent streets, alleys or rights of way abutting the real property, (iii) any pending or future award made in condemnation or to be made in lieu thereof, and any unpaid award for damages to the real property; (iv) any rights or awards with respect to tax assessment appeals, including any current or pending appeals; and (v) the use of appurtenant easements, whether or not of record, adjacent, contiguous, or adjoining the real property (the "Seller Owned Real Property");
 - (g) all motor vehicles set forth on <u>Schedule 2.1(g)</u>;
- (h) subject to Section 5.10, all Permits (including applications for issuance or renewal thereof) of the Seller with respect to the Huntsville Business or the other Sold Assets;
- (i) all billed and unbilled accounts, notes and other receivables including trade accounts receivables and all correspondence relating thereto with respect to the Huntsville Business and the lockboxes (but not the corresponding bank accounts) utilized by the Huntsville Business with respect to receipt of customer payments, excluding accounts receivable arising from the sale of Inventory manufactured by Seller or its Affiliates in whole or in part at any location other than the Seller Owned Real Property;
- (j) all copies of the books, records, ledgers, files, documents, correspondence, customer, supplier or other lists, manufacturing and engineering drawings and specifications, patterns, jigs, program maps, sales information, environmental records and files, business and marketing plans and proposals, service, maintenance and warranty records, procedure manuals, computer records, and other technical and business records of the Seller with respect to the Huntsville Business; provided, however, that the Seller shall be entitled to retain copies of any such materials that are necessary in its reasonable judgment for its Tax, accounting, personnel or legal purposes;

12

- (k) all other assets and rights used or held for use in the Huntsville Business;
- (l) all Avoidance Actions relating to any of the Sold Assets;
- (m) any claim, remedy, refund or right of the Seller relating to any asset listed in clauses (a) through (l) above, including any insurance benefits arising from or relating to such assets prior to the Closing and any right under any warranties or guarantees relating to any of the foregoing as of the Closing.

Seller shall transfer, assign or otherwise convey to the Buyer the Sold Assets free and clear of all Encumbrances other than Permitted Encumbrances and Encumbrances that may be created by or on behalf of the Buyer.

2.2 <u>Excluded Assets.</u>

The Sold Assets shall not include, and the Seller shall not sell, assign, transfer, convey or deliver to the Buyer, and the Buyer shall not purchase or acquire, any right, title or interest in or to any of the Excluded Assets. The term "Excluded Assets" shall mean the following:

- (a) the Equipment of the Seller located at the Seller Leased Real Property that is not listed on Schedule 1.1-3;
- (b) all of Seller's right, title and interest in and to Intellectual Property, including any trademarks, corporate names, trade names, logos, domain names, or any variation thereof, and any rights or interests therein, and the goodwill associated therewith, incorporating the name "Taylor-Wharton" or any abbreviation thereof (including "TW"), and including those set forth on subsection 3.11(m) of Schedule 3.11, but excluding the Acquired Intellectual Property;

- (c) any Contract that is not a Sold Contract, including the Contracts set forth on <u>Schedule 2.2(c)</u>;
- (d) all rights and incidents of interest of, and benefits accruing to, the Seller with respect to real property, including the Seller Leased Real Property, but excluding the Seller Owned Real Property;
- (e) all accounts, notes and other receivables carried on the Seller's books with respect to the Business (other than the Huntsville Business) and all lockboxes utilized by the Business with respect to receipt of customer payments;
- (f) all books, records, ledgers, files, documents, correspondence, customer, supplier or other lists, manufacturing and engineering drawings and specifications, patterns, jigs, program maps, sales information, environmental records and files, business and marketing plans and proposals, service, maintenance and warranty records, procedure manuals, computer records, and other technical and business records of the Seller with respect to the Business (other than the Huntsville Business and the Sold Assets);
 - (g) any Cash owned by the Seller;

13

- (h) the organizational documents, taxpayer and other identification numbers, minute and record books and the company seal of the Seller;
- (i) any Inventory that has been used or disposed of in the ordinary course, consistent with past practice, since the date of this Agreement;
- (j) any rights to the Seller's insurance policies, premiums or proceeds from insurance coverages relating to Excluded Liabilities or the assets of the Business (other than the Sold Assets) for any period, and any other recovery by the Seller for the benefit of or otherwise relating to the Business (other than the Huntsville Business or the Sold Assets) from any Person;
- (k) any rights to any refunds, and any deposits, of the Seller with any Governmental Authority, relating to Taxes for Pre-Closing Tax Periods;
- (l) all Tax Returns and financial statements of the Seller and the Business and all records (including working papers) related to the Seller or the Business pertaining to the Excluded Assets or the Excluded Liabilities;
- (m) all of the Seller's causes of action, claims, credits, demands or rights of set-off against third parties, to the extent related to any other Excluded Asset;
 - (n) all rights that accrue to the Seller under this Agreement;
- (o) any Inventory manufactured by Seller or its Affiliates in whole or in part at any location other than the Seller Owned Real Property;
- (p) any accounts receivable arising from the sale of Inventory manufactured by Seller or its Affiliates in whole or in part at any location other than the Seller Owned Real Property;
- (q) all Benefit Plans of Seller and any ERISA Affiliate, and all rights of the Seller under, and any funds and property held in trust or any other funding vehicle pursuant to, or any insurance contract providing funding for, any Benefit Plan;
- (r) the rights and benefits of all credits, prepaid expenses, deferred charges, advance payments, security deposits and prepaid items of the Seller with respect to the Business (other than the Huntsville Business);
- (s) all Permits (including applications for issuance or renewal thereof) of the Seller with respect to the Business (other than the Huntsville Business).

2.3 Assumption of Liabilities.

(a) <u>Assumed Liabilities</u>. On the terms and subject to the conditions set forth in this Agreement, at the Closing, the Buyer shall assume effective as of the Closing, and shall thereafter pay, perform, be responsible for and discharge as and when due only the following Liabilities of the Seller, but excluding any Excluded Liabilities (the "<u>Assumed Liabilities</u>"):

1 /

- (i) all Liabilities to be performed under the Sold Contracts after the Closing Date (but not (A) any Liability thereunder arising out of or in connection with any breach of any such Sold Contract occurring on or prior to the Closing Date or (B) any Liability relating to any product sold by the Business, the Seller or any Affiliate of the Seller on or prior to the Closing Date); and
- (ii) all Liabilities of the Seller with respect to the Huntsville Business for accounts payable and other current liabilities, but only to the extent included in the calculation of the Final Net Working Capital;
- (iii) all Liabilities arising out of or relating to accidents, occurrences and other incidents (including all Proceedings relating thereto) occurring after the Closing that result in (A) personal injury, (B) property damage or (C) any other Losses but only to the extent, in each case, such Liabilities result from, are caused by or arise out of, or are alleged to have resulted from, been caused by or arisen out of, directly or

indirectly, use of, exposure to or otherwise on account of, any Product that is both (x) manufactured by Seller or any of its Affiliates at the Seller Owned Real Property prior to the Closing Date and (y) sold, shipped or distributed by or on behalf of Buyer or any of its Affiliates on or after the Closing Date; and

(iv) all Liabilities relating to recalls, product liability claims or warranty claims with respect to any Product that is both (A) manufactured by Seller or any of its Affiliates at the Seller Owned Real Property prior to the Closing Date and (B) sold, shipped or distributed by or on behalf of the Buyer or any of its Affiliates on or after the Closing Date.

Notwithstanding anything herein the contrary, Buyer shall have no Liability with respect to high pressure cylinders manufactured in whole in part at any facility of Seller other than the Seller Owned Real Property.

- (b) <u>Excluded Liabilities</u>. The Buyer shall not assume, and shall not have been deemed to assume, any Liabilities other than the Assumed Liabilities. The term "<u>Excluded Liabilities</u>" shall mean all Liabilities of the Seller whether arising on or before the Closing Date, other than the Assumed Liabilities, including:
 - (i) all Liabilities arising out of Excluded Assets;
 - (ii) all Liabilities under any Benefit Plan of Seller or any ERISA Affiliate;
 - (iii) all employee and/or labor related Liabilities (including workers' compensation Liabilities);
 - (iv) all Liabilities (including with respect to loss of life, personal injury, damage to any real property, Environmental Claims, or natural resource damages) arising out of or resulting from any violation of any Environmental Law, Release of Hazardous Materials into the Environment or storage or off-site disposal of any Hazardous Materials: (A) occurring at or emanating from the Seller Owned Real Property

15

at any time prior to the Closing, or occurring at or emanating from the Seller Leased Real Property at any time whether at, before or after the Closing (except to the extent that the Liability arises out of or relates to a matter for which the Buyer is responsible pursuant to Section 2.9(c)), or (B) arising from the operation of the Business at any time prior to the Closing;

- (v) (A) all Liabilities of, or imposed on, the Seller related or attributable to Taxes and (B) all Periodic Taxes related or attributable to the Sold Assets for all Pre-Closing Tax Periods (determined in accordance with Section 5.3(d));
 - (vi) all Liabilities arising out of any Debt Obligations of the Seller;
- (vii) all Liabilities for any Third Party Expenses, severance pay obligations with respect to terminations on or prior to the Closing Date, or Change of Control Payments;
- (viii) all Liabilities arising out of or relating to accidents, occurrences and other incidents (including all Proceedings relating thereto), whether known or unknown and whether or not reported, that result in (A) personal injury, (B) property damage or (C) any other Losses, but only to the extent that, in each case, such Liabilities result from, are caused by or arise out of, or are alleged to have resulted from, been caused by or arisen out of, directly or indirectly, (1) use of, exposure to or otherwise on account of any Product manufactured, sold or distributed, or any service rendered, by or on behalf of the Seller on or prior to the Closing Date (other than any Product that is both (x) manufactured by Seller or any of its Affiliates at the Seller Owned Real Property prior to the Closing Date and (y) sold, shipped or distributed by or on behalf of Buyer or any of its Affiliates on or after the Closing Date); (2) automobile liability occurrences relating to the Seller on or prior to the Closing Date; or (3) workers' compensation occurrences relating to the Seller on or prior to the Closing Date;
 - (ix) all Liabilities for outstanding checks and other draws and drafts (including overdrafts) of the Seller;
- (x) all Liabilities owed to any Affiliate of Seller, other than the obligations under the TWI Supply Agreement made as of January 1, 2009 between the Seller and Sherwood Valve, LLC, as amended (but not any Liability thereunder arising out of or in connection with any breach of such agreement occurring on or prior to the Closing Date);
- (xi) all Liabilities relating to recalls, product liability claims or warranty claims with respect to products sold by the Business on or prior to the Closing Date, including the Aviation Recall;
 - (xii) all Liabilities under any Contract which is not a Sold Contract;
- (xiii) all Liabilities relating to any damage to the Seller Leased Real Property or any assets or Persons located at the Seller Leased Real Property (i) occurring

16

on or before the Closing, or (ii) occurring after the Closing, except to the extent the Buyer is responsible for such Liability pursuant to Section 2.9(c);

- (xiv) all Liabilities of Seller relating to any agreement to provide any customer any rebate, discount or refund;
- (xv) except as provided in Section 2.9(c), all Liabilities related to any of Seller's operations not located at the Seller Owned Real Property;

- (xvi) all Liabilities of Seller and its Affiliates under COBRA, including, but not limited to, any Liabilities on any type of "successor employer" basis under COBRA with respect to any current or former Benefit Plan or any current or former practice by Seller or any Affiliate of Seller to provide or make available group health plan coverage to any former employee or non-employee service provider or other non-employee individual (and their spouses and dependents).
- (c) <u>Cure Costs.</u> To the extent any amount is required to be paid to cure any monetary defaults which exists as of the Closing Date with respect to any executory contracts (as defined in the Bankruptcy Code) included in the Sold Contracts, Seller shall cure such monetary defaults at or prior to the Closing to the extent such cure is required by Section 365 of the Bankruptcy Code (any such amounts paid to cure any defaults, being referred to as the "<u>Cure Costs</u>"). For the sake of clarity, Buyer shall be responsible to pay all amounts due with respect to Sold Contracts, together with accounts payable and other current liabilities, to the extent the obligations to pay such amounts constitute Assumed Liabilities, it being understood that nothing in this Section 2.3(c) is intended to expand the Assumed Liabilities.
- (d) <u>Further Assurances</u>. Each of the Buyer and the Seller covenants that it will do, execute and deliver, or will cause to be done, executed and delivered, all such further acts and instruments that the other party hereto or any of its successors or permitted assigns may reasonably request in order to more fully evidence the assumption of the Assumed Liabilities provided for in this Section 2.3 and the sale and transfer of the Sold Assets.

2.4 <u>Purchase Price.</u>

On the Closing Date and subject to the terms and conditions set forth in this Agreement, in consideration of the sale, assignment and transfer of the Sold Assets, the Buyer shall (a) pay to the Seller an aggregate amount equal to \$11,000,000 in cash, minus Cure Costs not paid by Seller prior to the Closing, minus the Escrow Amount, minus the Harrisburg Holdback, plus the amount, if any, by which the Estimated Net Working Capital is greater than the Reference Working Capital or minus the amount, if any, by which the Estimated Net Working Capital is less than the Reference Working Capital (such amount, the "Closing NWC Adjustment"), by wire transfer of immediately available funds in U.S. dollars to one or more accounts of the Seller designated at least two Business Days prior to the Closing Date (the amount determined pursuant to this first sentence of Section 2.4, the "Closing Purchase Price"), and (b) assume the Assumed Liabilities (the payment, delivery and assumption described in subparts (a) and (b) of this Section 2.4, collectively, the "Total Consideration"). The Closing Purchase Price shall be

17

adjusted after the Closing Date pursuant to Section 2.5. The Closing Purchase Price, plus or minus the adjustment amounts determined pursuant to Section 2.5, shall be the "Purchase Price."

2.5 <u>Net Working Capital.</u>

- (a) <u>Estimated Net Working Capital</u>. No later than three Business Days prior to the Closing Date, the Seller shall prepare and deliver to the Buyer a good faith estimate of the Net Working Capital as of the Closing Date, together with all calculations related thereto (the "<u>Estimated Net Working Capital</u>" shall mean (i) the total current assets of the Seller included in the Sold Assets less any reserves against such current assets determined in accordance with the Agreed Principles, <u>less</u> (ii) the current Liabilities of the Seller included in the Assumed Liabilities, in each case, as of 12:01 a.m. (Eastern Time) on the Closing Date, determined in accordance with the Agreed Principles.
- (b) Physical Inventory; Closing Balance Sheet; Net Working Capital Statement. On the Closing Date, Buyer or its representatives shall have the right to conduct a physical inventory count of the Inventory of the Huntsville Business. Seller shall have the right to observe any such physical inventory. Within 90 days after the Closing Date, the Buyer shall cause to be prepared and delivered to the Seller and the Seller will reasonably cooperate with Buyer in connection with such preparation as reasonably requested by the Buyer: (i) a balance sheet (the "Closing Balance Sheet") of the Seller as of 12:01 a.m. (Eastern Time) on the Closing Date in accordance with the Agreed Principles; and (ii) a net working capital statement (the "Net Working Capital Statement"), setting forth the Net Working Capital (the "Final Net Working Capital") determined based on the Closing Balance Sheet. During the 30 days following receipt by the Seller of the Net Working Capital Statement, the Buyer shall, at the request of the Seller, afford the Seller reasonable access to the books and records with respect to the Sold Assets, the Assumed Liabilities and the Huntsville Business (to the extent relevant to the evaluation of the Net Working Capital Statement) and otherwise reasonably cooperate with the Seller in connection with its evaluation of the Net Working Capital Statement.
- (c) <u>Dispute</u>. Within 30 days following receipt by the Seller of the Net Working Capital Statement, the Seller shall deliver written notice to the Buyer of any dispute the Seller has with respect to the Net Working Capital Statement (the "<u>Net Working Capital Objection</u>") setting forth a specific description of the basis of the Net Working Capital Objection, the adjustments to the Net Working Capital Statement that the Seller believes should be made, and the Seller's calculation of the Final Net Working Capital. During such 30-day period, the Buyer shall, at the request of the Seller, afford the Seller reasonable access to the books and records with respect to the Sold Assets, the Assumed Liabilities and the Huntsville Business (to the extent relevant to the determination of the Final Net Working Capital) and otherwise reasonably cooperate with the Seller in connection with its review of the Net Working Capital Statement and preparation of any Net Working Capital Objection. The Seller shall be deemed to have accepted the Net Working Capital Statement except to the extent specifically disputed in the Net Working Capital Objection. Failure to so notify the Buyer within such 30-day period shall constitute acceptance and approval of the Buyer's calculation of the Final Net Working Capital Objection. If the Seller and the Buyer are unable to resolve all of their disagreements with respect to the items

18

specified in the Net Working Capital Objection by the 30th day following the Buyer's response thereto, after having used their commercially reasonable efforts to reach a resolution, they shall refer their remaining differences to PricewaterhouseCoopers, LLP or, if such firm refuses to accept such engagement (or such firm is, at the relevant time, doing any work for the Buyer or the Seller), another nationally recognized firm of independent public accountants as to which the Seller and the Buyer mutually agree acting promptly and in good faith (in either case, the "CPA Firm"), which shall, acting as experts in accounting, determine on a basis consistent with the requirements of this Agreement, and only with respect to the specific remaining accounting-related differences so submitted, whether and to what extent, if any, the Net Working Capital Statement requires adjustment. The Buyer and the Seller each agree to execute, if requested by the CPA Firm, a reasonable engagement letter. The Seller and the Buyer shall request the CPA Firm to render its determination within 45 days. All fees and expenses of the CPA Firm relating to this work shall be borne equally by the Seller and the Buyer. All determinations made by

the CPA Firm will be limited to the matters submitted to the CPA Firm by the Buyer and the Seller and shall be final, conclusive and binding on the parties and neither the Buyer nor the Seller nor any of their respective Affiliates shall seek further recourse to courts or other tribunals, other than to enforce the CPA Firm's determination. Judgment may be entered to enforce such determination in any court of competent jurisdiction. The Seller and the Buyer shall make reasonably available to the CPA Firm all relevant books and records, any work papers (including those of the parties' respective accountants) and supporting documentation relating to the Net Working Capital Statement and all other items reasonably requested by the CPA Firm. The "Final Statement of Net Working Capital" shall be (i) the Net Working Capital Statement in the event that (A) no Net Working Capital Objection is delivered to the Buyer during the initial 30-day period specified above with respect thereto or (B) the Seller and the Buyer so agree, (ii) the Net Working Capital Objection during the 30-day period specified above following receipt by the Buyer of the Net Working Capital Objection or (B) the Seller and the Buyer so agree or (iii) the Net Working Capital Statement, as adjusted pursuant to the agreement of the Buyer and the Seller or as adjusted by the CPA Firm together with any other modifications to the Net Working Capital Statement agreed upon by the Seller and the Buyer.

(d) Adjustment. For purposes of this Agreement, (i) the term "Final NWC Adjustment" shall mean the amount by which the Final Net Working Capital (as set forth on the Final Statement of Net Working Capital) is greater than or less than the Reference Working Capital, and (ii) the term "NWC True-Up" shall mean the amount equal to the Final NWC Adjustment (expressed as a positive number, if such adjustment amount was positive, and as a negative number, if such adjustment amount was negative) minus the Closing NWC Adjustment (expressed as a positive number, if such adjustment amount was positive, and as a negative number, if such adjustment amount was negative). For example, if the Closing NWC Adjustment was a three million dollar increase, then the NWC True-Up would be negative one million dollars; and if the Closing NWC Adjustment was a three million dollar decrease, and the Final NWC Adjustment is a two million dollar increase, then the NWC True-Up would be a positive five million dollars (i.e., subtracting a negative number converts it into a positive number). If the NWC True-Up amount is a positive number, the Buyer shall pay such positive amount, plus interest computed at the rate declared on the Closing Date by PNC Bank, National Association as its "base rate" (calculated

19

on the basis of 365 days and the actual number of days elapsed, the "Prime Rate") for the period from the Closing Date to the date of such payment, by wire transfer of immediately available funds to an account designated by the Seller. If the NWC True-Up amount is a negative number, the Seller shall pay such negative amount, plus interest computed at the Prime Rate for the period from the Closing Date to the date of such payment, by wire transfer of immediately available funds to an account designated by the Buyer. In each case, such payment shall be made within three Business Days after the date on which the Final Net Working Capital Statement is determined.

2.6 Escrow.

- (a) On the Closing Date, (i) the Buyer will deposit a portion of the Purchase Price equal to \$1,750,000 (the "Escrow Amount") with Wells Fargo Bank N.A., or another bank mutually acceptable to Buyer and Seller the "Escrow Agent"), by wire transfer of immediately available funds, to secure any obligation of the Seller, and TW International to make a payment under Section 2.5 above or Section 9.1 below or as provided in Section 5.3(h) below (ii) the Buyer, the Seller and TW International shall enter into an escrow agreement substantially in the form of Exhibit 2.6 with the Escrow Agent (the "Escrow Agreement"), which shall provide, among other things, for the release of \$875,000 to Seller (less the sum of any unresolved claims against the Escrow Amount and payments made from the Escrow Amount to Buyer) on the first anniversary of the Closing Date and the release of the balance of the Escrow Amount to Seller (including any accrued interest on the Escrow Amount, less the amount of any unresolved claims made against the Escrow Amount) on the date which is 18 months after the Closing Date; provided that any Escrow Amount that is not released in accordance with this Section 2.6 due the presence of one or more unresolved claims shall be promptly released to Seller upon the resolution of such claims to the extent that such claims have been resolved in favor of Seller in accordance with the provisions of the Escrow Agreement.
- (b) Seller grants Buyer a continuing security interest in all of Seller's right, title and interest in the Escrow Amount and the Escrow Agreement to secure the payment of amounts owing or which may become owing to Buyer under this Agreement or the Escrow Agreement; provided that, upon release of all or any part of the Escrow Amount to Seller in accordance with Section 2.6(a), the security interest granted by this Section 2.6(b) with respect to the amount so released will be immediately terminated without any further action by either party.
- (c) Buyer will not enforce any of its rights against the Escrow Amount as collateral for the obligations secured and has no right to payment or distribution from the Escrow Account except as set forth in this Agreement.
- (d) Seller authorizes Buyer to file one or more UCC financing statements describing the Escrow Amount. Seller also will execute such other documents as are necessary to perfect the security interest in favor of Buyer granted herein. Notwithstanding the foregoing, upon termination of Buyer's security interest in all or any part of the Escrow Amount, to the extent necessary to terminate Buyer's security interest in the portion of the Escrow Amount released, Buyer shall immediately file UCC termination statements with respect to that part of the Escrow Amount and execute such other documents as are necessary to terminate all of its security interest that part of the Escrow Amount.

20

(e) For so long as a part of the Escrow Amount remains subject to Buyer's security interest granted pursuant to this Section 2.6, Seller shall not grant any other security interest in or lien on that part of the Escrow Amount to any other Person.

2.7 <u>Allocation of Total Consideration.</u>

Within 90 days after the Closing Date, the Buyer shall prepare or have prepared and deliver to the Seller a schedule (an "Allocation Schedule") allocating the Total Consideration among the Sold Assets immediately after Closing, in such amounts reasonably determined by the Buyer to be consistent with Section 1060 of the Code and the Treasury Regulations thereunder; provided however the Seller and the Buyer will agree upon the portion of the Total Consideration to be allocated to the Seller Owned Real Property at or before the Closing.

(a) The Seller shall have a period of five Business Days after the delivery of the Allocation Schedule (the "Response Period") to present in writing to the Buyer notice of any objections the Seller may have to the allocations set forth therein (an "Objections Notice"). Unless the Seller

timely objects, such Allocation Schedule shall be binding on the parties without further adjustment, absent manifest error.

- (b) The Seller shall raise any objections within the Response Period, and the Buyer and the Seller shall negotiate in good faith and use their reasonable best efforts to resolve such dispute. If the parties fail to agree within 15 days after the delivery of the Objections Notice, then the disputed items shall be resolved by the CPA Firm, whose determination shall be final and binding on the parties; provided that the CPA Firm shall not adjust the Allocation Schedule except to the extent the Allocation Schedule is inconsistent (and no more than necessary to make the Allocation Schedule consistent) with Agreed Principles. The CPA Firm shall resolve the dispute within 30 days after the item has been referred to it. The costs, fees and expenses of the CPA Firm shall be borne equally by the Seller and the Buyer.
- (c) The Buyer and the Seller, and their respective Affiliates, shall report, act and file Tax Returns (including, but not limited to, Internal Revenue Service Form 8594) in all respects and for all purposes consistent with the Allocation Schedule, as modified by agreement of the parties or by determination above (the "Final Allocation Schedule"). Each of the Buyer and the Seller hereby covenants and agrees that such Person and its respective Affiliates will not take a position on any federal, state, local or foreign income, franchise, excise, sales or other Tax Return, before any Governmental Authority charged with the collection of any income, franchise, excise, sales or other Tax, or in any Proceeding, that is in any way inconsistent with the terms of this Section 2.7 and the Final Allocation Schedule. The Seller and the Affiliates of the Seller shall cooperate with the Buyer, and provide to the Buyer such documents, forms and other information as the Buyer may reasonably request to prepare such allocation.

2.8 The Closing.

Unless this Agreement shall have been terminated pursuant to ARTICLE VIII subject to ARTICLE VI and ARTICLE VII, the closing (the "<u>Closing</u>") of the transactions contemplated by this Agreement shall take place at the executive offices of the Seller, 4718 Old Gettysburg Rd., Suite 300, Mechanicsburg, PA 17055, on the third Business Day following the satisfaction

21

or waiver of all of the conditions set forth in ARTICLE VI and ARTICLE VII (other than those conditions that are to be satisfied at the Closing) (the "Closing Date"), or at such other place and time as may be agreed upon by the parties hereto. All proceedings to be taken and all documents to be executed and delivered by all parties at the Closing shall be deemed to have been taken and executed and delivered simultaneously and no proceedings shall be deemed to have been taken nor documents executed or delivered until all have been taken, executed and delivered. Legal title, equitable title and risk of loss with respect to the Sold Assets will be deemed transferred to or vested in the Buyer, and the transactions contemplated by this Agreement will be deemed effective for Tax, accounting and other computational purposes, and the parties will treat the Closing as if it had occurred, as of 12:01 a.m. (Eastern Time) on the Closing Date.

2.9 <u>Deliveries at the Closing.</u>

- (a) <u>Deliveries by the Seller</u>. At or prior to the Closing, the Seller shall deliver or cause to be delivered to the Buyer the following:
- (i) the Ancillary Agreements to which the Seller or any of its Affiliates is a party, duly executed by the Seller or such Affiliates, including the following documents pertaining to the transfer of the Seller Owned Real Property:
 - (A) Seller Owned Real Property Documents:
 - (1) <u>Deed.</u> The Deed in recordable form executed and acknowledged by the Seller in favor of the Buyer, in form and substance reasonably acceptable to the Buyer, its counsel and First American Title Insurance Company (the "<u>Title Insurer</u>"), the delivery and recording of which will vest in the Buyer good, marketable and indefeasible fee title in and to such real property and improvements, subject only to the Permitted Encumbrances;
 - Title Policy. Prior to Closing, at the Seller's and the Buyer's equally shared cost and expense, a commitment for an owner's policy of title insurance, in the form and condition as set forth below in this paragraph; at the Seller's and the Buyer's equally shared cost and expense, an ALTA Form 2006 owner's policy of title insurance, with standard exceptions, dated the date and time of the Closing (or a written binding commitment from the Title Insurer to deliver the policy of title insurance at a future date), with an amount of insurance equivalent to the allocation of Purchase Price as set forth herein, insuring the Buyer as sole owner of good, marketable and indefeasible fee title to said real property and improvements, subject only to the Permitted Encumbrances, and containing such customary endorsements as are reasonably acceptable to the Seller and the Buyer, of which the following shall be deemed reasonably acceptable if available in the applicable jurisdiction: (a) extended coverage over standard or general exceptions; (b) access; (c) location; (d) survey equivalency; (e) utility facility; (f) zoning with parking; (g) restrictions; (h) encroachment, if applicable; (i) subdivision, if applicable; (j) mineral rights, if applicable; and (k) contiguity, if applicable;

22

- (3) <u>Survey.</u> Prior to Closing, the most recent land title survey of the Seller Owned Real Property in Seller's possession, which survey was prepared in accordance with ALTA/ACSM standards and made by a surveyor or civil engineer that was duly licensed in the jurisdiction in which the Seller Owned Real Property is located as of the time the survey was prepared;
- (4) <u>Transfer Tax Forms</u>. All state, city and/or county Transfer Tax forms and returns required to be completed, filed or recorded at the Closing with respect to the Seller Owned Real Property;
- (5) Other Title and Escrow Documents. Such other documents as may be reasonably necessitated by the Title Insurer in connection with effectuating the issuance of the Title Policy and any related closing escrows, including an owner's affidavit or statement in customary and commercially reasonable form; provided, that this delivery shall not require any special or exception-specific indemnifications to permit the deletion or "insuring over" of any non-standard title exception; and

- (6) <u>Assignment of Rights</u>. Subject to Section 5.10, general assignment from the Seller to the Buyer with respect to any and all rights and claims in and to any Permits applicable to the Seller Owned Real Property, plans and specifications in the Seller's possession, if any, applicable to the Seller Owned Real Property, and any manufacturers' or contractors' warranties applicable to the Seller Owned Real Property;
 - (ii) a bill of sale in the form attached hereto as Exhibit 2.9(a)(ii);
 - (iii) an assignment and assumption agreement in the form attached hereto as Exhibit 2.9(a)(iii);
 - (iv) possession and occupancy of the Seller Owned Real Property, and possession of the Sold Assets;
- (v) a certificate of good standing (A) of the Seller, issued by the Secretary of State of the State of Delaware, dated within 10 Business Days of the Closing; and (B) of the Seller issued by the Secretary of State of the State of Alabama, dated as of the most recent practicable date;
- (vi) certified copies of resolutions duly adopted by the board of managers of the Seller, evidencing the taking of all limited liability company action necessary to authorize the execution, delivery and performance of this Agreement and the Ancillary Agreements to which the Seller is a party and the consummation of the transactions contemplated hereby and thereby;
- (vii) a certificate stating that the Seller is not a "foreign" person within the meaning of Section 1445 of the Code, which certificate shall set forth all information required by, and otherwise be executed in accordance with, Treasury Regulation Section 1.1445-2(b)(2), in a form reasonably acceptable to the Buyer;

23

- (viii) a certificate of the Secretary or Assistant Secretary of the Seller identifying the name and title and bearing the signatures of the officers of the Seller authorized to execute this Agreement and the other agreements, documents, certificates and instruments contemplated hereby;
 - (ix) a receipt for the Closing Purchase Price paid on the Closing Date;
- (x) amendments/releases to the TWI Loan Documents, duly executed by the parties thereto, which provide that the Seller is no longer a party thereto and confirm that the Seller has no Liabilities of any kind with respect thereto;
- (xi) forms of UCC-3 Termination Statements in proper form for filing upon the Closing and such other release documents and/or forms necessary to terminate validly or release all Encumbrances (other than Permitted Encumbrances) granted by the Seller or any of its Affiliates in favor of any Person against the Sold Assets;
 - (xii) evidence of the consents set forth on <u>Schedule 2.9(a)(xii)</u>;
- (xiii) Intellectual Property assignments, executed by the Seller or any of its Affiliates, assigning the Seller's or any of its Affiliates' entire right, title and interest in, to and under the trademarks, patents and domain names and their associated goodwill listed on subsection 3.11(b) of <u>Schedule 3.11</u> (but excluding those set forth on subsection 3.11(m) of <u>Schedule 3.11</u>);
- (xiv) an amendment to the TWI Supply Agreement made as of January 1, 2009 between the Seller and Sherwood Valve, LLC in the form attached hereto as Exhibit 2.9(a)(xiv); and
- (xv) a certificate in a form reasonably acceptable to Buyer dated the Closing Date and signed by the Seller certifying that the representations and warranties of the Seller contained in ARTICLE III are true, complete and correct in all material respects as of the Closing Date and representing and warranting that the conditions set forth in Sections 7.1, 7.2, 7.4, 7.7 and 7.8 have been satisfied; provided however if the Effective Date of the Plan occurs after June 12, 2010 and Buyer nevertheless elects to Close the transactions contemplated by this Agreement, such certificate shall not be required to refer to the conditions contained in Sections 7.4, 7.7 or 7.8.
 - (b) <u>Deliveries by the Buyer</u>. At or prior to the Closing, the Buyer shall deliver or cause to be delivered to the Seller the following:
- (i) the Closing Purchase Price, by wire transfer of immediately available funds to an account or accounts designated by the Seller, payable in accordance with Section 2.4 hereof;
 - (ii) the Ancillary Agreements to which the Buyer is a party, duly executed by the Buyer;

24

- (iii) a certificate of good standing of the Buyer, issued by the Secretary of State (or similar Governmental Authority) of its jurisdiction of incorporation, dated as of the most recent practicable date;
- (iv) a copy of the certificate of incorporation (or equivalent document) of the Buyer, certified by the Secretary of State (or similar Governmental Authority) of its jurisdiction of incorporation, dated as of the most recent practicable date;
- (v) certified copies of resolutions duly adopted by the board of directors of the Buyer evidencing the taking of all corporate action necessary to authorize the execution, delivery and performance of this Agreement and the Ancillary Agreements to which the Buyer is a party and the consummation of the transactions contemplated hereby and thereby;

- (vi) an amendment to the TWI Supply Agreement made as of January 1, 2009 between the Seller and Sherwood Valve, LLC in the form attached hereto as Exhibit 2.9(a)(vi);
- (vii) an assignment (in form and substance reasonably satisfactory to Buyer) by TWI to the Buyer of all right, title and interest in and to the Acquired Intellectual Property; and
- (viii) a certificate of the Secretary or Assistant Secretary of the Buyer identifying the name and title and bearing the signatures of the officers of the Buyer authorized to execute this Agreement and the other agreements, documents, certificates and instruments contemplated hereby.
- (c) <u>Post-Closing Delivery of Harrisburg Assets; Harrisburg Holdback.</u> Promptly following the Closing the Seller and the Buyer will agree upon on a date mutually convenient for the Seller and the Buyer, but not more than 45 days after the Closing Date, when the Buyer will remove the Harrisburg Assets from the Seller Leased Real Property (the "<u>Plant</u>"). Seller will provide sufficient access to the Plant so that Buyer may remove the Harrisburg Assets. In the event the Buyer is prevented by Seller or any third party, from accessing or reasonably removing the Harrisburg Assets, the above 45 day period will be extended until such time as Buyer is provided with the required access to the Plant to remove the Harrisburg Assets. The Buyer shall remove the Harrisburg Assets from the Seller Leased Real Property by means reasonably acceptable to Seller and dispose of the Harrisburg Assets in compliance with all applicable Laws. The Buyer shall be responsible for any damage to the Seller Leased Real Property caused by the removal of the Harrisburg Assets from the Seller Leased Real Property; but not any damage or condition which existed on, at, or under the Seller Leased Real Property prior to the date the Harrisburg Assets are removed from the Seller Leased Real Property. By way of further clarification, Buyer will not be responsible for any repair costs, restoration costs, testing costs, reporting costs or remediation costs which may result from a preexisting condition which is revealed upon removal of the Harrisburg Assets; provided that Buyer will be responsible for any such repair costs, restoration costs, testing costs, reporting costs or remediation costs to the extent that Buyer's negligence or willful misconduct exacerbates or otherwise adversely affects the preexisting condition. TW International and the Seller will be

25

solely responsible for resolving any dispute with the owner of the Seller Leased Real Property relating to Buyer's access to the Seller Leased Real Property for purposes of removing the Harrisburg Assets in a reasonable manner. At the sole cost and expense of TW International and Seller, TW International and Seller will take all actions, including, if reasonably necessary, instituting legal action, as may be reasonably necessary to provide Buyer the access to the Plant required to permit Buyer to remove the Harrisburg Assets. In regard to any legal actions required in connection with the immediately preceding sentence, Seller will use counsel reasonably acceptable to Buyer. If requested by Seller, Buyer will reasonably cooperate in Seller's efforts to resolve such dispute at Seller's and TWI's sole cost and expense. Upon receipt by Buyer of possession of the Harrisburg Assets on or prior to October 31, 2010, it will pay to Seller the Harrisburg Holdback less any amounts it was required to expend if Seller fails to pursue its obligations under this Section 2.9(c) diligently in good faith or otherwise expend at Seller's request in connection with the resolution of any dispute regarding its access to the Plant to remove the Harrisburg Assets. In the event that Buyer does not receive possession of the Harrisburg Assets due to Seller's failure to provide Buyer with access to the Plant reasonably necessary to permit Buyer to remove the Harrisburg Assets on or before October 31, 2010, Seller will pay to Buyer \$125,000 and Buyer may irrevocably retain the Harrisburg Holdback. Irrespective of whether Buyer is paid \$125,000 and entitled to retain the Harrisburg Holdback, Buyer may thereafter continue efforts to obtain access to the Harrisburg Assets and gain possession of the Harrisburg Assets, however after, October 31, 2010, it must do so at its own cost and expense.

2.10 Further Assurances.

Each party hereto covenants that it will do, execute and deliver, or will cause to be done, executed and delivered, all such further acts and instruments that the other party hereto or any of its successors or permitted assigns may reasonably request in order to more fully evidence the sale and transfer of the Sold Assets. To the extent any payment of any amounts owed to the Buyer are received by the Seller after the Closing, the Seller will, within five Business Days, (i) notify the Buyer of such receipt and (ii) transfer such funds and related supporting customer remittance or other lockbox receipt details received to the Buyer, without set off, recoupment, holdback, or debit or any kind or nature. To the extent any payment of any amounts owed to the Seller are received by the Buyer after the Closing, the Buyer will, within five Business Days, (i) notify Seller of such receipt and (ii) transfer such funds and related supporting customer remittance or other lockbox receipt details received to the Seller, without set off, recoupment, holdback, or debit or any kind or nature. In the event either of the Seller or the Buyer receive any checks made payable to the other in payment of any amounts owed to the other party, each party will promptly endorse such checks to the order of the other party and take such other actions to enable the proper party to deposit such checks in the party's accounts.

ARTICLE III. REPRESENTATIONS AND WARRANTIES OF THE SELLER

The Seller and TW International (with respect to Sections 3.1(b), 3.2(b) and 3.3) represent and warrant to the Buyer as of the date of this Agreement and as of the Closing Date as follows:

26

3.1 <u>Organization.</u>

(a) The Seller is a limited liability company duly formed, validly existing and in good standing under the Laws of the State of Delaware. The Seller has all requisite power and authority to own, lease and operate its assets and to carry on its business as now being conducted and is duly qualified or licensed to do business and is in good standing in the jurisdictions in which the ownership, lease or operation of its property or the conduct of its business requires such qualification or license, except where the failure to be so qualified or licensed would not have and could not reasonably be expected to have a Material Adverse Effect or a material adverse effect on the ability of the Seller to consummate the transactions contemplated by this Agreement. The Seller is qualified or licensed to do business in the State of Alabama and the Commonwealth of Pennsylvania.

(b) TW International is a limited liability company duly formed, validly existing and in good standing under the Laws of the State of Delaware. TW International has all requisite power and authority to own, lease and operate its assets and to carry on its business as now being conducted and is duly qualified or licensed to do business and is in good standing in the jurisdictions in which the ownership, lease or operation of its property or the conduct of its business requires such qualification or license, except where the failure to be so qualified or licensed would not have and could not reasonably be expected to have a material adverse effect on the ability of TW International to perform its obligations under this Agreement.

3.2 <u>Authorization; Enforceability.</u>

- (a) The Seller has the limited liability company power and authority to execute and deliver this Agreement and each Ancillary Agreement to which it is a party and perform its obligations hereunder and thereunder. The execution and delivery of this Agreement and the Ancillary Agreements to which the Seller is a party and the performance by it of its obligations hereunder and thereunder have been duly authorized by all necessary limited liability company action on the part of the Seller. This Agreement has been duly executed and delivered by the Seller and, assuming due authorization, execution and delivery by the Buyer, constitutes a valid and binding agreement of the Seller, enforceable against it in accordance with its terms, subject to General Enforceability Exceptions.
- (b) TW International has the limited liability company power and authority to execute and deliver this Agreement and each Ancillary Agreement to which it is a party and perform its obligations hereunder and thereunder. The execution and delivery of this Agreement and the Ancillary Agreements to which TW International is a party and the performance by it of its obligations hereunder and thereunder have been duly authorized by all necessary limited liability company action on the part of TW International. This Agreement has been duly executed and delivered by TW International and, assuming due authorization, execution and delivery by the Buyer, constitutes a valid and binding agreement of TW International, enforceable against it in accordance with its terms, subject to General Enforceability Exceptions.

3.3 <u>Membership Interests.</u>

TW International owns all of the membership interests of the Seller.

27

3.4 <u>Financial Statements; Accounting Controls and Procedures.</u>

- (a) Attached as Schedule 3.4 are (i) the unaudited statement of assets and liabilities of the Huntsville Business as of December 31, 2008, and the unaudited statement of income of the Huntsville Business for the year then ended, (ii) the unaudited statement of assets and liabilities of the Huntsville Business as of December 31, 2009, and the unaudited statement of income of the Huntsville Business for the year then ended (collectively, the "Financial Statements"), and (iii) unaudited statement of assets and liabilities of the Huntsville Business as of March 31, 2010 and the unaudited statement of income of the Huntsville Business for the three-month period then ended (the "Interim Financial Statements"). The foregoing income statements and statements of assets and liabilities included in the Financial Statements and Interim Financial Statements include the corporate allocations (such as insurance and selling, general and administrative expenses) described on Schedule 3.4. The Financial Statements and Interim Financial Statements present fairly, in all material respects, the financial condition and the results of operations of the Huntsville Business as of their respective dates and for the respective periods then ended. The Financial Statements have been prepared in accordance with GAAP, except with respect to (A) accounting for inventory and the capitalization of manufacturing labor and overhead variances into inventory and the testing of lower cost or market, (B) the recognition of pension accounting and (C) other minor deviations from GAAP that are not material in the aggregate.
- (b) The Seller, with respect to the Huntsville Business, maintains a separation of duties in the management of its financial books and records. The Seller maintains responsibility for control over its cash receipts and disbursements separate from the accounting organization in its treasury department. The treasury department scrutinizes the Seller's cash receipts and other receivables. The treasury department requires a three way match of purchase order, receipt of goods, and invoice before a payment is made and any deviations from this practice require the approval of the Seller's treasurer. The Seller maintains its cash holdings at a physical location separate from the Seller Owned Real Property. The information technology enterprise resource planning systems of the Seller include controls that limit user access to the functions and information necessary for each user to perform his or her responsibilities. The Seller's accounting organization reconciles accounts and Seller's financial management periodically (though not systematically) reviews key accounts. The Seller performs a count of all inventories one time per year and reconciles the count to its ledgers. The Seller performs a count on certain inventory items on a more regular basis. The Seller's controller compiles the financial results of the Seller. These results are reviewed periodically with the Seller's management and board of managers and delivered to the Seller's lenders with an attestation as to the validity thereof from the Seller's chief financial officer. To the Knowledge of the Seller, the foregoing controls and procedures provide reasonable assurance that material transactions are duly authorized, executed and recorded on the books and records of the Seller. The Seller does not have an internal audit department and there is no independent audit of the Seller's books and records. Nothing in this Section 3.4(b) shall be deemed to limit the representations and warranties contained in Section 3.4(a).

28

3.5 No Approvals or Conflicts.

(a) Except as set forth on Schedule 3.5, the execution, delivery and performance by the Seller of this Agreement and the Ancillary Agreements to which it is a party and the consummation by the Seller of the transactions and performance of its obligations contemplated hereby and thereby do not and will not (i) violate, conflict with or result in a breach by the Seller of its organizational documents (including its certificate of formation and limited liability company agreement), (ii) violate, conflict with or result in a breach of, or constitute a default by the Seller (or create an event that, with notice or lapse of time or both, would constitute a default) or give rise to any payment or other penalty or any right of termination, cancellation or acceleration that could be materially adverse to the transactions contemplated by this Agreement, or result in the creation of any Encumbrance upon any of the Sold Assets, under any material note, bond, mortgage, indenture, deed of trust, license, franchise, permit (including the permits listed on Schedule 3.13), lease, contract or other instrument to which the Seller or any of the Sold Assets may be bound, (iii) violate or result in a material breach of any Governmental Order or Law applicable to the Seller or any of the Sold Assets or (iv) except for filings with the Bankruptcy Court and the issuance of the Sale Order and as may be required by any unique aspect of Buyer, require any order, consent, approval or authorization of, or notice to, or declaration, filing, application, qualification or registration with, any Governmental Authority.

(b) The execution, delivery and performance by TW International of this Agreement and the Ancillary Agreements to which it is a party and the performance of its obligations contemplated hereby and thereby do not and will not (i) violate, conflict with or result in a breach by TW International of its organizational documents (including its certificate of formation and limited liability company agreement), (ii) violate, conflict with or result in a breach of, or constitute a default by TW International (or create an event that, with notice or lapse of time or both, would constitute a default) or give rise to any payment or other penalty or any right of termination, cancellation or acceleration, under any material note, bond, mortgage, indenture, deed of trust, license, franchise, permit, lease, contract or other instrument to which TW International or its assets may be bound, (iii) violate or result in a material breach of any Governmental Order or Law applicable to TW International or any of the Sold Assets or (iv) except for filings with the Bankruptcy Court and the issuance of the Sale Order and as may be required by any unique aspect of Buyer, require any order, consent, approval or authorization of, or notice to, or declaration, filing, application, qualification or registration with, any Governmental Authority.

3.6 <u>Compliance with Law; Permits.</u>

Except as set forth on <u>Schedule 3.6(i)</u> (and except with respect to compliance with Environmental Laws, which is covered solely by Section 3.13), since December 7, 2007, the Seller has conducted the Huntsville Business, and the Sold Assets have been maintained, in compliance in all material respects with all Laws. Except with respect to Permits required under Environmental Laws (which are covered solely by Section 3.13), the Seller possesses all material Permits necessary to conduct the Huntsville Business as conducted since December 7, 2007, and as necessary to own, lease or operate the Sold Assets as owned, leased or operated by Seller since December 7, 2007. All Permits described in the immediately preceding sentence are listed

29

on <u>Schedule 3.6(ii)</u>. All such Permits necessary to conduct the Huntsville Business as conducted since December 7, 2007 and operate the Sold Assets as operated since December 7, 2007 are in full force and effect and, except as set forth on <u>Schedule 3.5</u> or <u>Schedule 3.6</u>, are transferable to the Buyer at the Closing. Since December 7, 2007, the Huntsville Business has been conducted, in all material respects, in accordance with the requirements of such Permits.

3.7 <u>Proceedings.</u>

Except as set forth on <u>Schedule 3.7</u> (and except with respect to Environmental Claims, which are covered solely by Section 3.13), there are no Proceedings pending or, to the Knowledge of the Seller, threatened against the Huntsville Business or the Seller.

3.8 <u>Tax Matters.</u>

Except as set forth on Schedule 3.8:

- (a) (i) All Tax Returns of the Seller and all other income, sales and other material Tax Returns that relate to the Huntsville Business or the Sold Assets have been timely filed, and all such Tax Returns have been properly completed in compliance with all applicable Laws, and are true, correct and complete; and (ii) all Taxes shown to be due on any such Tax Return, and all other Taxes due and payable by Seller or that relate or are attributable to the Huntsville Business or the Sold Assets, have been timely paid (whether or not shown or required to be shown on any Tax Return) except for Taxes specifically listed on Schedule 3.8 being contested in good faith and for which adequate reserves have been established and maintained in accordance with the Agreed Principles of the Seller and specifically listed on Schedule 3.8.
- (b) The Seller has duly and timely collected and remitted all sales, use, excise or similar Taxes related or attributable to the Sold Assets or the Huntsville Business in accordance with applicable Law. The Seller does not have any Liability for the Taxes of any third Person with respect to the Sold Assets as a transferee or successor, by contract or otherwise.
- (c) No material reassessments (for property, ad valorem or other Tax purposes) of any of the Sold Assets have been proposed in writing.
 - (d) There are no Encumbrances relating to Taxes encumbering any Sold Assets, except for Permitted Encumbrances.
- (e) The Seller is, and has been for its entire existence, a disregarded entity within the meaning of Treasury Regulations Section 301.7701-3(b)(1)(ii).
- (f) No claim has ever been made by any Governmental Authority in a jurisdiction where the Seller does not file Tax Returns that the Seller is or may be subject to taxation by that jurisdiction. All Taxes that the Seller is required to have withheld or collected have been duly withheld or collected and, to the extent required, have been paid to the appropriate Taxing Authority, including, without limitation, all Taxes required to have been withheld or collected and paid in connection with amounts paid or owing to any employee, former employee, partner, independent contractor, creditor, shareholder, member, affiliate, customer, supplier or other third party, and all Forms W-2 and 1099 required with respect thereto

30

have been properly completed and timely filed. None of the Sold Assets (i) is stock of any corporation or other entity classified for federal income tax purposes as an association taxable as a corporation, (ii) is "tax-exempt use property" within the meaning of Section 168(h) of the Code, or (iii) secures any debt, the interest on which is exempt from Tax under Section 103 of the Code. The Seller is not a party to any "safe harbor lease" that is subject to the provisions of Section 168(f)(8) of the Internal Revenue Code as in effect prior to the Tax Reform Act of 1986. No Change in Control Payments will be nondeductible under Section 280G or 162(m) of the Code or subject to Tax under Section 4999 of the Code. No member of the Seller and no manager, director or officer (or employee responsible for Tax matters) of Seller expects any Governmental Authority to assess any additional Taxes for any period for which Tax Returns have been filed. To the Knowledge of the Seller there is no dispute or claim concerning any Tax Liability of the Seller.

(g) The unpaid Taxes of the Seller (A) did not, as of March 31, 2010, exceed the reserve for Tax Liability (rather than any reserve for deferred Taxes established to reflect timing differences between book and Tax income) set forth on the face of the balance sheet included in the Interim Financial Statements (rather than in any notes thereto) and (B) do not exceed that reserve as adjusted for the passage of time through the Closing Date in accordance with the past custom and practice of Seller in filing their Tax Returns.

3.9 <u>Employee Benefits.</u>

Schedule 3.9 sets forth a list of (a) each "employee benefit plan" (within the meaning of Section 3(3) of ERISA), (b) all other severance, salary continuation, Change of Control Payment, employment, incentive, bonus, option, membership interest purchase, restricted membership interest, retirement, pension, redundancy, profit sharing or deferred compensation plans (whether qualified or nonqualified), programs, agreements or policies and (c) all other employee benefit plans or programs, in each case (i) in which Huntsville Employees participate (other than any such plans, programs, agreements or policies required by Law to be provided to any such employees, including workers' compensation or similar benefits) sponsored or maintained by the Seller, (ii) with respect to which the Seller has made or is required to make payments, transfers or contributions for Huntsville Employees or (iii) or with respect to which the Seller has any current or contingent Liability (collectively, the "Benefit Plans"). There is no lien pursuant to Code Sections 412(n) (as in effect through the date of its repeal) or 430(k) or pursuant to ERISA Sections 303(k), 4062 or 4068 in favor of, or enforceable by, the Pension Benefit Guaranty Corporation or any other entity with respect to any of the Sold Assets. For purposes of this Section 3.9 only, the term "Seller" includes any ERISA Affiliate.

3.10 <u>Labor Relations.</u>

(a) The Seller is not a party to any collective bargaining agreement applicable to the Huntsville Employees, nor is any such contract or agreement presently being negotiated; (b) there is no unfair labor practice charge or complaint pending or, to the Knowledge of the Seller, threatened against the Seller with respect to the Huntsville Employees; (c) there have been no grievances, arbitrations or other similar proceedings since December 7, 2007 under, or pertaining to, any collective bargaining agreement or any associated side letters or agreements applicable to employees of the Seller; and (d) there is, and since December 7, 2007 there has

31

been, no labor strike, slowdown, work stoppage, or lockout in effect, or, to the Knowledge of the Seller, threatened against or otherwise affecting the Seller. To the Knowledge of the Seller, there is no effort to organize employees of the Seller that is pending or threatened as of the date hereof.

3.11 <u>Intellectual Property.</u>

- (a) Notwithstanding anything to the contrary contained in this Agreement, only the representations and warranties contained in this Section 3.11 shall apply to the Acquired Intellectual Property.
- (b) <u>Schedule 3.11</u> sets forth a complete and correct list of all of the registered forms (including all applications for registration) and all material unregistered forms of the Acquired Intellectual Property.
- (c) Except as otherwise specifically identified on <u>Schedule 3.11</u>, all registered forms (including all applications for registration) of the Acquired Intellectual Property identified on <u>Schedule 3.11</u> have been duly issued (or in the case of applications for registration, duly filed) and have not been cancelled, abandoned or otherwise terminated.
 - (d) None of the unregistered forms of the Acquired Intellectual Property identified on <u>Schedule 3.11</u> has been abandoned.
- (e) Except as otherwise specifically set forth on Schedule 3.11, all actions required to have been taken from and after December 7, 2007 to record each owner throughout the chain of title of all of the Acquired Intellectual Property required to have been listed on Schedule 3.11, with each applicable Governmental Authority up to the date of this Agreement, have been taken, including payment of all costs, fees, taxes and expenses associated with such recording activities.
- (f) Except as otherwise specifically set forth on Schedule 3.11 or Schedule 3.12(a), the Seller is the sole owner of all right, title and interest in and to all of the registered forms (including all applications for registration) and all of the material unregistered forms of the Acquired Intellectual Property, free and clear of all Encumbrances other than Permitted Encumbrances, and all governmental fees associated therewith and due as of the date of this Agreement have been paid in full.
- (g) Except as otherwise specifically set forth on <u>Schedule 3.11</u> or <u>Schedule 3.12(a)</u>, to the Knowledge of the Seller, since December 7, 2007, the Seller has used commercially reasonable efforts to protect the secrecy and confidentiality of the trade secrets used or held for use primarily in the Huntsville Business.
 - (h) Set forth on <u>Schedule 3.11</u> is a complete and correct list of all proprietary software included in the Acquired Intellectual Property.
- (i) Except pursuant to a Contract set forth in subsection 3.12(a)(v) of <u>Schedule 3.12(a)</u>, since December 7, 2007, the Seller has not licensed any material Intellectual Property with respect to the Huntsville Business from any Person.

32

- (j) Except pursuant to a Contract set forth in subsection 3.12(a)(v) of Schedule 3.12(a), since December 7, 2007, the Seller has not granted any license or other right that does or that will, subsequent to the Closing, permit or enable any third Person other than the Buyer to use any Acquired Intellectual Property, and no such licenses or rights are in effect.
- (k) Except as set forth on Schedule 3.7, since December 7, 2007, the Seller has not received any written notice of any claim and, to the Knowledge of the Seller, there is no threatened claim against the Seller asserting that any of the Acquired Intellectual Property infringes upon or otherwise conflicts with the Intellectual Property of any Person, and the Seller has not given any notice to any Person asserting infringement by such Person of any of the Acquired Intellectual Property.
- (l) Except as otherwise specifically set forth on <u>Schedule 3.11</u> or in subsection 3.12(a)(v) of <u>Schedule 3.12(a)</u>, immediately upon Closing, the Buyer shall own all registered forms of (including all applications for registration) and all unregistered forms of Intellectual Property included in

the Acquired Intellectual Property.

(m) Subsection 3.11(m) of <u>Schedule 3.11</u> is a complete and correct list of all registered trademarks and domain names (including applications for registration) owned by the Seller that include the "Taylor-Wharton" name and abbreviations thereof (including "TW"). The continued ownership and the use after the Closing of such trademarks and domain names shall be governed by Sections 2.2(b) and 5.8, hereof.

3.12 Contracts.

- (a) <u>Schedule 3.12(a)</u>sets forth all of the following contracts and agreements ("<u>Contracts</u>") to which the Seller is a party or by which it is bound (in either case, with respect to the Huntsville Business) as of the date of this Agreement, other than Benefit Plans (collectively, the "<u>Material Contracts</u>"):
 - (i) Contracts involving the expenditure by the Seller (with respect to the Huntsville Business) of more than \$10,000 in any instance for the purchase of materials (other than raw materials, which is covered by clause (xii) below), supplies, equipment or services;
 - (ii) indentures, mortgages, loan agreements, capital leases, security agreements, or other Contracts for the borrowing of money by the Seller, other than any of the foregoing from which the Sold Assets will be released in connection with the transactions contemplated hereby;
 - (iii) guarantees of the obligations of other Persons, other than any of the foregoing from which the Sold Assets will be released in connection with the transaction contemplated hereby;
 - (iv) Contracts that restrict the assignees of the Seller after the date of this Agreement from engaging in any line of business in any geographic area or competing with any Person;

33

- (v) Contracts under which the Seller has licensed material Intellectual Property (other than shrink-wrap, click-wrap and off-the-shelf software) to or from any other Person (including Affiliates of the Seller);
- (vi) partnership, limited liability company, joint venture agreements or other agreements involving a sharing of the profits or expenses by the Seller;
- (vii) Contracts under which the Seller will have obligations or contingent Liabilities after the date of this Agreement relating to the acquisition or sale of any business enterprise;
- (viii) Contracts by and among the Seller, on the one hand, and any of its Affiliates, on the other hand, including the Supply Agreement, dated January 1, 2009, by and between the Seller and Sherwood Valve LLC;
- (ix) any Contract (including employment and consulting Contracts) with any current or former manager, officer or employee of the Seller;
 - (x) distributor, sales representative, agency, dealer or similar Contracts;
- (xi) except as set forth with respect to clause (v) above, any Contract providing that the Seller will receive future payments (with respect to the Huntsville Business) aggregating more than \$10,000 prior to the expiration of such Contract, other than purchase orders for the sale of Products by the Seller in the ordinary course of business;
- (xii) Contracts involving the expenditure by the Seller (with respect to the Huntsville Business) of more than \$10,000 for the purchase of raw materials;
 - (xiii) any Contract which obligates the Seller or the Huntsville Business to provide any rebates or discounts; and
- (xiv) any Contract containing any "take or pay" provisions or minimum purchase conditions or requirements involving the expenditure by the Seller (with respect to the Huntsville Business) of more than \$10,000 in any instance for the purchase of raw materials, excluding any such Contracts that are terminable by the Seller without penalty on not more than 30 days' notice.
- (b) Except for the TWI Loan Documents, true and complete copies (or, if oral, written summaries) of each of the Material Contracts have been provided to the Buyer or its representatives.
- (c) Except as set forth on <u>Schedule 3.12(c)</u>, each Material Contract is in full force and effect, and is a valid and binding agreement of the Seller and, to the Knowledge of the Seller, each of the other parties thereto, enforceable by or against the Seller and, to the Knowledge of the Seller, each of such other parties thereto in accordance with its terms, subject to the General Enforceability Exceptions. Except for the Proceedings in the Bankruptcy Court

34

and as set forth on Schedule 3.12(c), no condition exists or event has occurred since December 7, 2007 that (whether with or without notice or lapse of time or both) would constitute a material default by (i) the Seller under any Material Contract or (ii) to the Knowledge of the Seller, any other party to any Material Contract.

To the extent that Seller or TW International makes any representation or warranty in this Section 3.13 with respect to any period of time, or any event, occurrence or other matter that occurred, prior to December 7, 2007, that representation or warranty is hereby qualified by and made solely with respect to the Knowledge of the Seller. Except as otherwise disclosed in any audit, study, report, analysis or result of investigation identified on <u>Schedule 3.13</u> or the report attached hereto as <u>Exhibit 3.13</u>:

- (a) The Seller with respect to the Seller Owned Real Property is in compliance with all applicable Environmental Laws. All licenses and permits currently held by the Seller pursuant to Environmental Laws with respect to the Seller Owned Real Property are identified on Schedule 3.13(a). The Seller is in full compliance with such licenses and permits and, to the extent required prior to Closing, timely and complete application has been or will be made for renewal, extension, or reissuance of all such licenses and permits, and no Governmental Authority has provided the Seller with any written statement to the effect that such renewal, extension, or reissuance will be denied. To the Knowledge of the Seller, no Governmental authority has indicated that any license or permit held by Seller with respect to the Seller Owned Real Property, the Sold Assets or the Huntsville Business may not be renewed, extended or reissued in due course and as requested without the imposition of a penalty.
- (b) The Seller is not subject to an order of any court or Governmental Authority relating to any Environmental Law and there is no Environmental Claim pending or, to the Knowledge of the Seller, threatened against the Seller with respect to the Seller Owned Real Property.
- (c) No Hazardous Materials have been stored, transported, treated or disposed of on, in, under, or from the Seller Owned Real Property in any manner that violates applicable Environmental Laws.
- (d) Except for routine inspections, there have been and are no investigations of the Seller Owned Real Property under any Environmental Law, or any activities conducted thereon, by any Governmental Authority or other Person.
- (e) <u>Schedule 3.13(e)</u> lists all environmental audits, studies, reports, analyses and results of investigations related to the past or present environmental condition or compliance status of the Sold Assets and/or the Seller Owned Real Property ("Reports") in Seller's possession, custody or control. The Seller has provided to the Buyer true and complete copies of all such Reports.
- (f) To the Knowledge of Seller, there are not and have never been located at the Seller Owned Real Property any underground storage tanks, above-ground storage tanks, dry cleaning or laundry operations, gasoline filling stations, automobile repair operations, waste

35

treatment, storage or disposal areas, asbestos-containing materials, polychlorinated biphenyls, or any other condition that may be a basis for Liability under any Environmental Law.

(g) The Seller has not disposed or arranged for the disposal of any Hazardous Material at any off-site location in any manner that may give rise to Liability under any Environmental Law, including, but not limited to, the Comprehensive Environmental Response, Compensation, and Liability Act, as amended, 42 USC 9601 et seq.

No representations or warranties in this Agreement other than those set forth in this Section 3.13 will be deemed to relate to Environmental Laws, Releases of Hazardous Materials or other environmental matters.

3.14 Personal Property Assets.

- (a) The Seller has good and valid title to, or holds by valid and existing lease or license, all the material tangible personal property assets included in the Sold Assets, reflected as assets on the Statement of Assets and Liabilities, or acquired after December 31, 2009, except with respect to inventory disposed of in the ordinary course of business consistent with past practice since such date, free and clear of all Encumbrances except for Permitted Encumbrances. The Harrisburg Assets are not subject to any lien in favor of the landlord of the Seller Leased Real Property.
- (b) Except as set forth on Schedule 3.14, all such personal property assets included in the Sold Assets (other than the Harrisburg Assets) are (x) free from any material defects, (y) in reasonably good maintenance, operating condition and repair, normal wear and tear excepted and (z) reasonably suitable for the purposes for which such personal property is has been used since December 7, 2007.
- (c) As of January 27, 2010, the forge and draw bench which are included in the Harrisburg Assets were in good working condition. Such forge and draw bench have not been used since January 27, 2010 and, to the Knowledge of the Seller, remain in the same condition as they were in on January 27, 2010. Except as otherwise set forth in this Agreement, Seller makes no representation or warranty of any kind whatsoever, whether express or implied, oral or written, regarding any Harrisburg Assets, and, except as set forth in this Agreement, Buyer acknowledges and agrees that it is purchasing the tangible personal property assets listed on Schedule 2.1(a)(ii) on an AS IS WHERE IS basis. All of the Harrisburg Assets are located at the Seller Leased Real Property.

3.15 Seller Owned Real Property.

With respect to the Seller Owned Real Property, except as otherwise specified on Schedule 3.15:

(a) the Seller has good and marketable fee simple title to the Seller Owned Real Property, free and clear of any Encumbrances, except for Permitted Encumbrances:

36

(b) there are no pending or, to the Knowledge of the Seller, threatened Proceedings (including condemnation or land use or zoning related actions) affecting the Seller Owned Real Property, except for such Proceedings as are listed on <u>Schedule 3.15(b)</u>;

- (c) all real property Taxes and assessments are current and have been paid when due and the Seller has not received notice of any pending or threatened special assessment proceedings affecting the Seller Owned Real Property;
- (d) except for Permitted Encumbrances, the Seller Owned Real Property is not subject to a lease, sublease, license or other agreement, written or, to the Knowledge of the Seller, oral, granting any Person any right to the use, occupancy or enjoyment thereof (or any portion thereof);
- (e) to the Knowledge of the Seller, water, electric, gas and sewer utility services and septic tank and storm drainage facilities currently available are adequate in all material respects for the present use of the Seller Owned Real Property in the conduct of the Huntsville Business;
- (f) the electrical, mechanical, plumbing, heating, air conditioning, ventilation, fire detection and sprinkler systems in the buildings, and the boilers, and the roofs and walls and foundations of the buildings, are in reasonably good maintenance, operating condition and repair, subject to ordinary wear and tear; and
- (g) there are no existing violations of any Laws applicable to the Seller Owned Real Property, including zoning ordinances, regulations, orders or requirements of departments of housing, building, fire, labor, health, or other municipal departments or other Governmental Authority having jurisdiction against or affecting the Seller Owned Real Property, that could have a Material Adverse Effect on the Huntsville Business.

3.16 No Brokers' or Other Fees.

No Person has acted, directly or indirectly, as a broker, finder, financial advisor or investment banker for the Seller in connection with the transactions contemplated by this Agreement and no Person is entitled to any fee or commission or like payment in respect thereof.

3.17 <u>Undisclosed Liabilities.</u>

Except for Liabilities (a) expressly reflected or reserved against on the Interim Financial Statements, (b) arising in the ordinary course of business since the date of the Interim Financial Statements under any Contract (excluding any Liability for a breach of any such Contract) to which the Seller (with respect to the Huntsville Business) is a party, (c) otherwise incurred in the ordinary course of business since the date of the Interim Financial Statements or (d) specifically disclosed on Schedule 3.17 or any other Disclosure Schedule to this Agreement, the Seller (with respect to the Huntsville Business) does not have any Liabilities required to be set forth on a balance sheet prepared in accordance with GAAP or in the notes to such a balance sheet.

37

3.18 <u>Customers and Suppliers.</u>

Schedule 3.18 lists the 10 largest customers of the Seller with respect to the Huntsville Business (based on gross sales) during the 2008 and 2009 fiscal years (the "Customers") and the 10 largest suppliers of goods or services (the "Suppliers") to the Seller with respect to the Huntsville Business (based on expenditures) during the 2008 and 2009 fiscal years, and, with respect to each, the name and dollar volume involved. Since December 31, 2009, no Customer or Supplier has terminated or materially and adversely altered its relationship with the Seller (with respect to the Huntsville Business). To the Knowledge of the Seller no Customer or Supplier has any intention to terminate or materially and adversely alter its relationship with the Seller (with respect to the Huntsville Business), including, without limitation, to:

- (a) cease the use of the Products or services of the Huntsville Business;
- (b) substantially reduce the use of the Products or services of the Huntsville Business;
- (c) seek to reduce the price it will pay for the Products or services of the Huntsville Business;
- (d) seek to increase the price it will charge for Products or services supplied to the Business;
- (e) threaten Seller or place Seller on probation or put Seller on the "no bid" or similar list for future orders;
- (f) seek to materially change the volume, timing or specifications of any present or new business; or
- (g) seek to alter the specifications of any Products, in any material respect.

The parties agree that Seller will not be in breach of this representation if, between the date of this Agreement and the Closing, a Customer notifies Seller or Seller otherwise becomes aware, that a Customer intends to take an action described in this Section 3.18 as a result of the termination of the rebate agreement set forth on Schedule 2.2(c) with such Customer, if any; provided that Seller complies with its obligation to provide Buyer with prompt written notice of the receipt of such notification or that it has become aware of such intention.

3.19 Product and Services Liability.

(a) A complete and correct copy of the standard terms and conditions of sale for each of the Products or services of the Seller constituting a part of the Huntsville Business (containing applicable guaranty, warranty and indemnity provisions) is attached as <u>Schedule 3.19(a)</u>. Except as set forth on <u>Schedule 3.19(a)</u>, no product manufactured, sold, or delivered by, or service rendered by or on behalf of, the Seller with respect to the Huntsville Business is subject to any guaranty, warranty or other indemnity, express or implied, other than as expressly provided under such standard terms and conditions attached hereto.

38

pending or, to the Knowledge of the Seller, threatened, against the Seller in respect of any product, component or other item manufactured, sold, designed or produced or service rendered by the Seller in connection with the Huntsville Business. Set forth on Schedule 3.19(b) is a list of every claim (whether based on strict liability, negligence, breach of warranty (express, implied or as required by applicable law), breach of contract, tort or otherwise) that, to the Knowledge of the Seller, has been made or is pending or threatened, against Harsco Corporation, in respect of any product, component or other item manufactured, sold, designed or produced or service rendered by Harsco Corporation in respect of its business conducted at the Seller Owned Real Property.

- (c) Except as set forth on <u>Schedule 3.19(c)</u>, Seller has not entered into, or offered to enter into, any agreement, contract commitment or other arrangement (whether written or oral) pursuant to which the Huntsville Business or the Seller is or will be obligated to make any rebates, discounts, promotional allowances or similar payments or arrangements to any customer of the Huntsville Business ("<u>Rebate Obligations</u>"). All Rebate Obligations are fully and accurately reflected in the Financial Statements.
- (d) Except as set forth on Schedule 3.19(d), there is and there has been, no pending or, to the Knowledge of the Seller, threatened claim, action, suit, proceeding, arbitration or investigation against the Seller for injury to person or property of employees or any third parties suffered as a result of the sale of any product or performance of any service by the Seller in connection with the Huntsville Business, including claims arising out of the alleged defective or unsafe nature of its products or services.
- (e) Except for the Aviation Recall, there is no pending or, to the Knowledge of the Seller, threatened recall (whether government required or initiated by Seller) or investigation of any product designed, manufactured or sold by the Seller in connection with the Huntsville Business.

3.20 <u>Acquired Inventory.</u>

All of the Acquired Inventory are in good and usable condition, are useable or saleable in the ordinary course of business and are carried on the books and records of the Seller (including the Financial Statements and the Interim Financial Statements) at the lower of cost (determined on a first-in-first-out basis) or market value in accordance with the Agreed Principles, subject to any reserves (determined in accordance with the Agreed Principles) for obsolete or slow-moving inventory set forth on such financial statements. Since December 31, 2009, there has not been a material change in the method of valuing the Acquired Inventory or in the determination of how and whether costs or other items are capitalized into inventory. Schedule 3.20 sets forth all material consignment agreements pursuant to which Acquired Inventory are now consigned or have been consigned to others since December 7, 2007. All Acquired Inventory (other than Acquired Inventory in transit or Acquired Inventory consigned to others) are located at the Seller Owned Real Property. None of the Acquired Inventory was manufactured in whole or in part by Seller at a location other than the Seller Owned Real Property.

39

3.21 <u>Accounts Receivable.</u>

- (a) Schedule 3.21(a) sets forth a true, correct and complete list as of the date set forth thereon of the accounts, notes and other receivables of the Seller with respect to the Huntsville Business (collectively, the "Acquired Accounts Receivable"). Schedule 3.21(a) includes an aging of all the Acquired Accounts Receivable showing amounts due in 30-day aging categories. Not less than five Business Days prior to the Closing Date, the Seller shall deliver to the Buyer a true, complete and correct list of all Acquired Accounts Receivable, including an aging in 30-day categories, as of a date not more than five Business Day prior to the Closing Date, which shall be attached to Schedule 3.21(a). None of the Acquired Accounts Receivable arose from the sale of Inventory manufactured in whole or in part by Seller at a location other than the Seller Owned Real Property.
- (b) The Acquired Accounts Receivable represent or will represent valid obligations arising solely out of bona fide sales, performance of services and other business transactions in the ordinary course of business consistent with past practice, and are not subject to set-offs, counterclaims or valid defenses, subject to allowances for bad debt recorded (i) on the Interim Financial Statements as of the date of such Interim Financial Statements or (ii) on the Closing Balance Sheet as of the date of such Closing Balance Sheet. No person has made any request or agreement for deduction or discount under any Acquired Accounts Receivable. No such Acquired Accounts Receivable are owed by a Person or entity that has sought the protection of any bankruptcy or insolvency Law or is the subject of any dispute as to payment. All Acquired Accounts Receivable, net of rebates, are current and collectible.

3.22 <u>Acquisitions and Divestitures.</u>

The Seller was formed solely to acquire the Business and, prior to the Seller's acquisition of the Business on December 7, 2007, the Seller had no assets or liabilities and did not conduct any operations. Seller does not own any equity interest in any other Person.

3.23 <u>Certain Business Relationships with the Seller.</u>

Except as disclosed on Schedule 3.23, neither the Seller, any Affiliate of Seller nor any current director, manager or officer of the Seller (each, a "Related Party"): (a) owns, directly or indirectly, any interest in any Person that is a competitor, supplier or customer of the Huntsville Business; (b) owns, directly or indirectly, in whole or in part, any material property, asset or right, real, personal or mixed, tangible or intangible that is utilized by or in connection with the Huntsville Business (including any of the Acquired Intellectual Property); (c) is a customer or supplier of the Huntsville Business; or (d) directly or indirectly has an interest in or is a party to any contract, agreement, lease, arrangement or understanding, whether or not in writing, pertaining or relating to the Huntsville Business, except for employment, consulting or other personal service agreements; provided, however, the beneficial ownership of not more than 2% of securities of any entity that are traded on a national securities exchange or over-the-counter market shall not be deemed to breach this Section 3.23.

40

3.24 <u>Employees; Employment Matters.</u>

(a) <u>Compliance with Laws</u>. Except as set forth on <u>Schedule 3.24(a)</u>, since December 7, 2007, the Seller (with respect to Huntsville Business) has complied in all material respects with all applicable Laws relating to labor or labor relations and employment standards, including any provisions thereof relating to wages, hours, immigration control, discrimination, accommodation, retaliation or "whistle-blowing", employee safety and

health, termination pay, vacation pay, fringe benefits, employee benefits, collective bargaining and the payment and/or accrual of the same and all insurance and all other costs and expenses applicable thereto.

(b) WARN Act. With respect to the transactions contemplated by this Agreement, any notice required under any Law has been given or will be timely given. Except as set forth on Schedule 3.24(b), the Seller has not implemented any plant closing or layoff of employees with respect to the Huntsville Business that could implicate the Worker Adjustment and Retraining Notification Act of 1988, as amended, or any similar foreign, state or local law, regulation or ordinance (collectively, the "WARN Act"); provided, however, that the foregoing representation shall not be deemed breached by the Seller because of any action taken by the Buyer on or after the Closing that, when combined with any action taken by the Seller prior to the Closing, triggers, results in or causes to arise a Liability or obligation of the Seller under the WARN Act.

3.25 <u>Sufficiency of Sold Assets.</u>

The Sold Assets are adequate and sufficient for the operation of the Huntsville Business, and constitute all of the assets, rights and properties necessary to conduct the Huntsville Business, as conducted by the Seller since December 7, 2007. The Sold Assets constitute all of the assets, rights and properties owned by the Seller or its Affiliates and used or held for use in the operation of the Huntsville Business.

3.26 Absence of Changes.

Except as disclosed on <u>Schedule 3.26</u> and except for the transactions contemplated by this Agreement, the Huntsville Business has been operated only in the ordinary course consistent with the practice of the Seller since December 7, 2007. Since December 31, 2009, and with respect to the Huntsville Business, except as disclosed on <u>Schedule 3.26</u> and except for the transactions contemplated by this Agreement, Seller has not with respect to the Huntsville Business:

- (a) suffered any change in its condition (financial or otherwise), operations (present or prospective), business (present or prospective), properties, assets or liabilities, other than changes in the ordinary course of business which have not had a Material Adverse Effect;
- (b) suffered any damage, destruction or loss of real or personal property (whether or not covered by insurance) having a Material Adverse Effect;
- (c) incurred or agreed to incur any indebtedness for borrowed money (excluding working capital borrowings in the ordinary course of business consistent with past practice);

41

- (d) paid nor became obligated to pay in excess of \$5,000 in the aggregate for fixed assets;
- (e) sold, purchased, transferred or otherwise disposed of, or agreed to sell, purchase, transfer or otherwise dispose of, through merger, consolidation, sale of assets, purchase of assets or otherwise, any assets, or canceled, or agreed to cancel, any debts or claims, other than the sale of inventory in the ordinary course of business consistent with past practice;
- (f) mortgaged, pledged or subjected to any Encumbrance, or agreed to mortgage, pledge or subject to any Encumbrance, any of the Sold Assets:
 - (g) entered into any new agreement or arrangement with any Affiliates;
- (h) been threatened with or placed on probation or put on the "no bid" or similar list by any Customer or Supplier or had any material order canceled or, to the Knowledge of the Seller, been threatened with cancellation of any material order;
- (i) entered into, made or permitted any material amendment or termination of any of the Material Contracts other than in the ordinary course of business;
- (j) experienced any shortage or difficulty in obtaining any raw materials or components which caused a stoppage in the operations of a customer:
- (k) made any change in its Tax or accounting methods or practices with respect to its condition, operations, business, properties, assets or liabilities:
- (l) made any change in its practices with respect to the collection of its Accounts Receivables, payment of accounts payable or purchase and sale of Inventory;
 - (m) deferred or delayed any capital expenditures; or
 - (n) instituted, settled or agreed to settle any material litigation, action or Proceeding before any court or Governmental Authority.

3.27 <u>Capital Expenditures.</u>

Set forth on <u>Schedule 3.27</u> is a list of all expenditures made by the Seller with respect to the Huntsville Business in 2008 and 2009 that should be treated as capital expenditures on a balance sheet prepared in accordance with GAAP.

3.28 <u>Disclosure.</u>

No statement, representation or warranty made by the Seller in this Agreement or any Ancillary Agreement schedules, attachments or exhibits attached to this Agreement or any Ancillary Agreement, contains any untrue statement of any material fact or omits a material fact necessary to make the statements contained in this Agreement, the Ancillary Agreements or such schedules, attachments or exhibits, not misleading.

3.29 No Other Representations or Warranties.

Except for the representations and warranties contained in this ARTICLE III (as modified by the Disclosure Schedules hereto), neither the Seller nor any other Person makes any other express or implied representation or warranty with respect to the Seller, TW International, the Huntsville Business, the Sold Assets, the Assumed Liabilities or the transactions contemplated by this Agreement, and, except for the representations and warranties contained in this ARTICLE III (as modified by the Disclosure Schedules hereto), the Seller disclaims any other representations or warranties, whether made by the Seller, any Affiliate of the Seller or any of their respective officers, directors, employees, agents or representatives, including any representation or warranty regarding the probable success or profitability of the Huntsville Business or any projections, forecasts or budgets provided with respect thereto, and including any confidential memoranda distributed on behalf of the Seller relating to the Huntsville Business or other publications or data room information provided to the Buyer or its representatives, or any other document or information in any form provided to the Buyer or its representatives, including management presentations, in connection with the sale of the Huntsville Business and the Sold Assets, the assumption of the Assumed Liabilities and the transactions contemplated hereby. Except for the representations and warranties contained in this ARTICLE III (as modified by the Disclosure Schedules hereto), the Seller (a) expressly disclaims any representation or warranty, express or implied, at common law, by statute or otherwise relating to the condition of the Huntsville Business or the Sold Assets (including any implied or expressed warranty of merchantability or fitness for a particular purpose, or of conformity to models or samples of materials) and (b) disclaims all liability and responsibility for any representation, warranty, statement or information made, communicated or furnished (orally or in writing) to the Buyer or its Affiliates or representatives (including any opinion, information or advice that may have been or may be provided to the Buyer by any director, manager, officer, employee, agent, consultant or representative of the Seller or any of its Affiliates). The Seller makes no representations or warranties to the Buyer regarding the probable success or profitability of the Huntsville Business.

ARTICLE IV. REPRESENTATIONS AND WARRANTIES OF THE BUYER

The Buyer hereby represents and warrants to the Seller as follows:

4.1 Organization.

The Buyer is a corporation duly formed, validly existing and in good standing under the Laws of the State of Delaware. The Buyer has all requisite power and authority to own, lease or operate its assets and to carry on its business as now being conducted and is duly qualified or licensed to do business and is in good standing in the jurisdictions in which the ownership of its property or the conduct of its business requires such qualification or license, except where the failure to be so qualified or licensed would not reasonably be expected, individually or in the aggregate, to have a material adverse effect on the ability of the Buyer to consummate the transactions contemplated by this Agreement.

43

4.2 <u>Authorization; Enforceability.</u>

The Buyer has the corporate power and authority to execute and deliver this Agreement and the Ancillary Agreements to which it is a party and perform its obligations hereunder and thereunder. The execution and delivery of this Agreement and the Ancillary Agreements to which it is a party by the Buyer and the performance by it of its obligations hereunder and thereunder have been duly authorized by all necessary corporate action by the Buyer. This Agreement has been duly executed and delivered by the Buyer and, assuming due authorization, execution and delivery by the Seller, constitutes a valid and binding agreement of the Buyer, enforceable against it in accordance with its terms, subject to the General Enforceability Exceptions.

4.3 <u>No Approvals or Conflicts.</u>

The execution, delivery and performance by the Buyer of this Agreement and the Ancillary Agreements to which it is a party and the consummation by the Buyer of the transactions contemplated hereby and thereby do not and will not (a) violate, conflict with or result in a breach by the Buyer of (i) its organizational documents (including its certificate of incorporation and bylaws) or (ii) any resolution(s) adopted by the board of directors of the Buyer, (b) violate, conflict with or result in a breach of, or constitute a default by the Buyer (or create an event that, with notice or lapse of time or both, would constitute a default) or give rise to any payment or other penalty or any right of termination, cancellation or acceleration under, or result in the creation of any Encumbrance upon any of the properties of the Buyer under, any material note, bond, mortgage, indenture, deed of trust, license, franchise, Permit, lease, contract, agreement or other material instrument to which the Buyer or any of its properties may be bound, (c) violate or result in a material breach of any Governmental Order or Law applicable to the Buyer or any of its properties or (d) require any order, consent, approval or authorization of, or notice to, or declaration, filing, application, qualification or registration with, any Governmental Authority.

4.4 <u>Proceedings.</u>

There are no Proceedings pending or, to the Knowledge of the Buyer, threatened against the Buyer or any of its Subsidiaries that would have a material adverse effect on the ability of the Buyer to consummate the transactions contemplated by this Agreement. The Buyer is not subject to any Governmental Order that would have a material adverse effect on the ability of the Buyer to consummate the transactions contemplated by this Agreement.

4.5 <u>Compliance with Laws; Permits.</u>

Neither the Buyer nor any of its Subsidiaries is in violation of any Governmental Order or Law applicable to them or any of their respective properties, except where noncompliance would not have a material adverse effect on the ability of the Buyer to consummate the transactions contemplated by this Agreement.

4.6 <u>Financing.</u>

The Buyer has sufficient funds to pay the Purchase Price and all related transaction expenses incurred by or on behalf of the Buyer and to consummate the transactions contemplated hereby (the "<u>Transaction Financing</u>"). There are no conditions precedent or other contingencies related to the funding of the full amount of the Transaction Financing.

4.7 No Brokers' or Other Fees.

No Person has acted, directly or indirectly, as a broker, finder, financial advisor or investment banker for the Buyer in connection with the transactions contemplated by this Agreement and no Person is entitled to any fee or commission or like payment in respect thereof.

4.8 Condition of the Business.

Notwithstanding anything contained in this Agreement to the contrary, the Buyer acknowledges and agrees that the Seller is not making any representations or warranties whatsoever, express or implied, beyond those expressly given by the Seller in ARTICLE III (as modified by the Disclosure Schedules hereto as supplemented or amended), and the Buyer acknowledges and agrees that, except for the representations and warranties contained in ARTICLE III, the Sold Assets and the Huntsville Business are being transferred on a "where is" and, as to condition, "as is" basis. Any claims the Buyer may have for breach of representation or warranty shall be based solely on the representations and warranties of the Seller set forth in ARTICLE III (as modified by the Disclosure Schedules hereto as supplemented or amended). The Buyer further represents that neither the Seller nor any of its Affiliates nor any other Person has made any representation or warranty, express or implied, as to the accuracy or completeness of any information regarding the Seller, the Huntsville Business, the Sold Assets, the Assumed Liabilities or the transactions contemplated by this Agreement not expressly set forth in ARTICLE III, and none of the Seller, any of its Affiliates or any other Person will have or be subject to any liability to the Buyer or any other Person resulting from the distribution to the Buyer or its representatives or the Buyer's use of any such information, including any confidential memoranda distributed on behalf of the Seller relating to the Huntsville Business, the Sold Assets, the Assumed Liabilities or other publications or data room information provided to the Buyer or its representatives, or any other document or information in any form provided to the Buyer or its representatives, including management presentations, in connection with the sale of the Huntsville Business and the Sold Assets, the assumption of the Assumed Liabilities and the transactions contemplated hereby.

4.9 Solvency.

In completing the transactions contemplated by this Agreement, the Buyer does not intend to hinder, delay or defraud any present or future creditors of the Buyer or the Seller. Immediately after giving effect to the consummation of the transactions contemplated by this Agreement, and assuming the accuracy of the Seller's representations and warranties contained herein:

45

- (a) the fair saleable value (determined on a going concern basis) of the assets of the Buyer shall be greater than the total amount of its Liabilities (including all Liabilities, whether or not reflected in a balance sheet prepared in accordance with GAAP); and
 - (b) the Buyer shall be able to pay its debts and obligations in the ordinary course of business as they become due.

ARTICLE V. COVENANTS AND AGREEMENTS

5.1 <u>Conduct of Business Prior to the Closing.</u>

Without the consent of the Buyer, which consent shall not be unreasonably withheld, conditioned or delayed, except as contemplated by this Agreement or as disclosed on Schedule 5.1, from and after the date of this Agreement and until the Closing, the Seller shall (i) conduct the operations of the Huntsville Business in the ordinary course consistent in all material respects with past practice and (ii) use its commercially reasonable efforts to maintain satisfactory relationships with suppliers, customers and others having material business relationships with the Huntsville Business. Without limiting the generality of the foregoing, except as contemplated by this Agreement and except as set forth on Schedule 5.1, the Seller (with respect to the Sold Assets and the Huntsville Business) shall not do any of the following without the prior written consent of the Buyer, which consent shall not be unreasonably withheld, conditioned or delayed:

- (a) sell or otherwise dispose of any of the Sold Assets having an aggregate value exceeding \$10,000, excluding sales of inventory in the ordinary course of business consistent with past practice;
- (b) acquire assets that will be included in the Sold Assets having an aggregate value exceeding \$10,000, excluding (i) acquisitions of inventory in the ordinary course of business consistent with past practice, and (ii) capital expenditures permitted by clause (d) below;
 - (c) merge or consolidate with any Person;
 - (d) make any material capital commitments with respect to the Huntsville Business in excess of \$10,000;
 - (e) incur, assume or guarantee any material Debt Obligation that would become an Assumed Liability;
 - (f) incur any Encumbrance on any material Sold Assets, other than Permitted Encumbrances;
- (g) increase the cash compensation of Huntsville Employees other than (i) in the ordinary course of business or (ii) as required by any Contract in effect as of the date hereof and listed on Schedule 3.12(a) or as required by Law;

- (h) enter into any Contract that restricts or will restrict the Huntsville Business after the date of this Agreement from engaging in any line of business in any geographic area or competing with any Person that materially impairs the operation of the Huntsville Business;
- (i) enter into any partnership, limited liability company or joint venture agreement that materially impairs the operation of the Huntsville Business;
 - (j) enter into, amend or renew any collective bargaining agreements regarding the Huntsville Employees;
 - (k) terminate or make any material amendment to or waive any material rights under a Material Contract;
 - (l) permit the Effective Date of the Plan to occur on or prior to June 12, 2010; or
 - (m) agree or commit to do any of the foregoing.

Notwithstanding anything to the contrary contained herein, the Seller shall be permitted (without notice or consent to Buyer) to distribute all Cash held by it as of the Closing Date to TW International. Notwithstanding the foregoing, Seller shall not be deemed to have breached this Section 5.1, if (a) such breach was inadvertent, (b) is capable of being cured and (c) Seller cures such breach within five days after such breach occurs. Seller shall promptly notify Buyer if it discovers it has breached any provision of this Section 5.1.

5.2 Access to Books and Records; Cooperation.

Except as provided in Section 5.3 and in clause (f) of this Section 5.2:

- (a) From the Closing and until the seventh anniversary of the Closing, during normal business hours, upon at least five days prior notice, the Buyer shall permit, at the Seller's sole cost and expense, the Seller and its counsel, accountants and other authorized representatives to have reasonable access to the officers, directors, employees, accountants and other advisors and agents, properties, books, records and contracts of the Buyer related to the Huntsville Business and the Sold Assets, and the right, at Seller's expense, to make copies and extracts from such books, records and contracts, to the extent necessary to facilitate the resolution of any claims made by or against or incurred by the Seller with respect to the Sold Assets, the Huntsville Business or the Excluded Liabilities.
- (b) From the Closing and until the seventh anniversary of the Closing, during normal business hours, upon at least five days prior notice, the Seller shall permit, at the Buyer's sole cost and expense, the Buyer and its counsel, accountants and other authorized representatives to have reasonable access to the officers, managers, employees, accountants and other advisors and agents, properties, books, records and contracts of the Seller as they relate to the Huntsville Business, the Sold Assets or the Assumed Liabilities, and the right, at Buyer's expense, to make copies and extracts from such books, records and contracts, to the extent necessary to facilitate the resolution of any claims made by or against or incurred by the Buyer with respect to the Sold Assets, the Huntsville Business or the Assumed Liabilities.

47

- (c) Until the seventh anniversary of the Closing, the Buyer shall not, and shall cause its Affiliates not to, destroy, at any time any files or records that are subject to Section 5.2(a) without giving written notice to the Seller and giving the Seller 60 days following receipt of such notice to request in writing that all or a portion of the records intended to be destroyed be delivered to the Seller at the Seller's expense.
- (d) During the period commencing on the date hereof and ending on the Closing, the Seller shall afford the Buyer and its counsel, accountants and other authorized representatives, reasonable access, during normal business hours, upon reasonable advance notice, to the officers, managers, employees, accountants and other advisors and agents, properties, books, records and contracts of the Seller with respect to the Huntsville Business, Sold Assets or Assumed Liabilities, including such access as is reasonably necessary for Buyer and its consultants to conduct a Phase II environmental review of the Seller Owned Real Property in accordance with the scope of work set forth in Exhibit 5.2(d) (the "Phase II Review"); provided, that such access does not interfere in any material respect with the Seller's normal business operations. The parties agree that the provisions of the Confidentiality Agreement shall continue in full force and effect following the execution and delivery of this Agreement as provided in Section 5.13. All information obtained by the Buyer and its counsel, accountants and representatives pursuant to this Section 5.2(d) shall be kept confidential in accordance with Section 5.13.
- Notwithstanding any provision in Section 5.2(d) to the contrary, the Buyer's right to conduct the Phase II Review shall be subject to the following conditions: (i) the Buyer shall provide evidence that the Buyer and each consultant, representative or agent of the Buyer that performs or participates in the Phase II Review maintain the minimum insurance coverages specified in Exhibit 5.2(e), (ii) upon performance and upon completion of the Phase II Review, the Buyer shall be responsible and shall restore or cause its representatives to restore, at Buyer's sole cost and expense, the property affected by the Phase II Review to the condition it was in immediately before the Phase II Review and repair any damage occasioned by the Phase II Review; provided that the foregoing obligation shall not be deemed to include any obligation with respect to any latent condition on, under or affecting Seller Owned Real Property which is exacerbated, expanded, discovered or modified, by the testing conducted within the agreed scope of the Phase II Review, (iii) the Buyer and its consultant, representative or agent shall not perform any testing, drilling, sampling or any other work at the Seller Owned Real Property except pursuant to the scope of work set forth in Exhibit 5.2(d), (iv) the Seller and its consultants, agents and representatives shall have the right to inspect and monitor all aspects of the Phase II Review, (v) at Seller's request not less than 24 hours prior to commencement of the Phase II Review, the Buyer shall provide the Seller with sufficient quantities of all samples and other materials obtained by the Buyer and its consultants, representatives and agents in connection with the Phase II Review to enable the Seller to perform its own independent testing of the same, (vi) upon Seller's request, the Buyer shall provide the Seller with a copy of any final reports produced as a result of the Phase II Review upon the Buyer's receipt of the same and any such final reports produced will be certified for Seller's use and reliance to the same extent as for Buyer. Seller agrees that any report, sample or other information provided to Seller in connection with the Phase II Review is subject to the terms of the Confidentiality Agreement and Seller agrees to hold it in strict confidence, not to disclose it to any third party and to use it solely for purposes of exercising or enforcing its rights under this Agreement.

- (f) The Seller shall, on the one hand, use, and the Buyer shall, on the other hand, use, all commercially reasonable efforts to obtain and to cooperate in obtaining any consent, approval, authorization or order of, and in making any registration or filing with, any Governmental Authority or other Person required in connection with the execution, delivery or performance of this Agreement by such party.
- Nothing in this Agreement shall impose obligations on the Seller to give the Buyer or its counsel, accountants or other authorized representatives access to information if such access could reasonably be expected to cause the Seller or any of its Affiliates to be in breach of any duty of confidence or any other duty or obligation under applicable Law (including Laws affecting privacy, personal information and the collection, handling, storage, processing, use or disclosure of data). Nothing in this Agreement shall impose obligations on the Buyer to give the Seller copies of Tax Returns of the Buyer or any Affiliate of Buyer. Nothing in this Agreement shall restrict any party to this Agreement from destroying any information in accordance with its record retention policies that are generally applicable to such information or require it to provide any notice of such destruction.

5.3 <u>Tax Matters: Cooperation; Preparation and Filing of Tax Returns; Transfer Taxes and other Tax Matters.</u>

- (a) <u>Cooperation</u>. Subject to the other provisions of this Agreement, the Buyer and the Seller agree to furnish or cause to be furnished to each other, upon request, such information and assistance relating to the Huntsville Business and the Sold Assets (including access to books and records, employees, contractors and representatives) as is reasonably necessary for the filing of all Tax Returns, the making of any election related to Taxes, the preparation for any audit or other examination by any Taxing Authority, and the prosecution or defense of any Proceeding relating to any Tax Return. Further, the Seller shall be permitted to retain, in its discretion, copies of any such books and records relating to the Huntsville Business and the Sold Assets as are necessary for any of such purposes as set forth above. The Buyer and the Seller shall retain, and the Seller shall cause TWI Holdings and TW International to retain, all books and records with respect to Taxes pertaining to the Huntsville Business and the Sold Assets until the expiration of all relevant statutes of limitations (and, to the extent notified by the Buyer or the Seller, any extensions thereof). At the end of such period, each party shall provide the other with at least 60 days prior written notice before destroying any such books and records, during which period the party receiving such notice can elect to take possession, at its own expense, of such books and records.
- (b) <u>Pre-Closing Taxes</u>. All Liabilities for Periodic Taxes attributable to the Huntsville Business or the Sold Assets for Pre-closing Tax Periods shall be the responsibility of the Seller, and any refunds or credits or rights to refunds or credits with respect thereto shall be the property of the Seller.
- (c) <u>Post-Closing Taxes</u>. All Liabilities for Periodic Taxes attributable to the Huntsville Business or the Sold Assets for Tax periods (or the portion of any Straddle Period) beginning on or after the Closing Date shall be the responsibility of the Buyer, and any refunds or credits or rights to refunds or credits with respect thereto shall be the property of the Buyer.

49

- (d) Proration of Taxes. All real property Taxes, personal property Taxes and similar ad valorem obligations levied with respect to the Huntsville Business or the Sold Assets for a Straddle Period, whether such Taxes are payable to a Governmental Authority, a landlord or other third party ("Periodic Taxes") shall be apportioned between Sections 5.3(b) and 5.3(c) as of the Closing Date based upon, respectively, the number of calendar days in the portion of such Straddle Period ending on the Closing Date and the number of calendar days in the portion of such Straddle Period commencing on the day following the Closing Date. Such prorations shall be based upon the most recent available Tax statement and latest Tax valuation. If the Closing occurs before the Tax rate is fixed for the then current Tax period, the proration of the corresponding Taxes shall be on the basis of the Tax rate for the last preceding Tax period applied to the latest assessed valuation. The Seller shall be responsible for filing all Tax Returns relating to such Taxes with respect to the Huntsville Business and the Sold Assets that are required to be filed on or prior to the Closing Date, and the Buyer shall be responsible for filing all Tax Returns relating to such Taxes with respect to the Huntsville Business and the Sold Assets required to be filed after the Closing Date. The portion of any Periodic Taxes which were paid by Seller in advance but relate to portion of the Straddle Period commencing on the day after the Closing Date will be treated as a prepaid current asset on the Closing Balance Sheet, and the portion of any Periodic Taxes which relate to a portion of the Straddle Period ending on or prior to the Closing Date but which are required to be paid by Buyer after the Closing Date, will be treated as a accrued current liability on the Closing Balance Sheet.
- <u>Transfer Taxes</u>. The Buyer shall pay 50% of and the Seller shall pay 50% of any (i) transfer, real property transfer, documentary, sales, use, stamp, recording and similar Taxes (including all applicable real estate transfer Taxes, but excluding any Taxes based on or attributable to income or gains) and related fees (including any penalties, interest and additions to Tax) incurred in connection with the sale and transfer of the Sold Assets by the Seller to the Buyer pursuant to the terms of this Agreement and (ii) Duties that may be imposed by any Governmental Authority in connection with the sale and transfer of the Sold Assets by the Seller to the Buyer pursuant to the terms of this Agreement (together, "Transfer Taxes"). The Seller shall be responsible for preparing and filing all Tax Returns or other applicable documents in connection with the Transfer Taxes, to the extent permitted by applicable Law. The Buyer shall cooperate with the Seller in the preparation and filing of all Tax Returns or other applicable documents for or with respect to Transfer Taxes. The Seller shall (A) unless the Closing occurs after the Effective Date of the Plan, use commercially reasonable efforts to cause the transactions contemplated by this Agreement to be exempt from Transfer Taxes under 11 U.S.C. § 1146, and (B) in any event, make any and all filings (including applications for tax clearance certificates) reasonably available under applicable Law to be relieved of (and to relieve Buyer from) any Transfer Taxes otherwise applicable to the contemplated transactions. Seller shall, at least five (5) days prior to Closing, deliver to Buyer for review and comment all Tax Returns and other documentation which Seller is, pursuant to this Section 5.3(e), required to prepare. To the extent any such Tax Returns (adjusted to take into account Buyer's comments that are reasonably acceptable to Seller) reflect that any Transfer Taxes will be due, Buyer may elect to withhold the 50% of the amount of such Taxes that is payable by the Seller pursuant to this Section 5.3(e) from the Purchase Price, and cause such Taxes to be paid directly to the appropriate taxing authority in connection with the filing of such Tax Returns (which Buyer may, by notice to Seller, elect to file and, to the extent so filed on behalf of Seller, shall be executed by Seller as

- (f) <u>Reimbursements.</u> If the Buyer (or an Affiliate of the Buyer) or the Seller (or an Affiliate of the Seller) pays any Taxes to be borne by another Person under this Section 5.3, the other Person shall promptly reimburse the payor for the Taxes paid. If any party (or an Affiliate thereof) receives any refunds or credits that are the property of another party under this Section 5.3, the party receiving such refunds or credits shall promptly pay (or cause its Affiliate to pay, as the case may be) the amount of such refunds or credits to the other party hereto.
- (g) <u>Employment Taxes</u>. Forms W-2 for those employees of the Business hired by the Buyer shall be prepared and filed in accordance with the "alternate procedure" pursuant to Internal Revenue Service Revenue Procedure 2004-53, I.R.B. 2004-34, and any comparable provisions of applicable state and local Laws. Additionally, as required by the "alternate procedure," Schedule D of Form 941 shall be prepared and timely filed by both the Buyer and the Seller to explain any discrepancies between Form W-2 and Form 941 reporting.
- Tax Clearance. Set forth on Schedule 5.3(h) is a list of each jurisdiction in which the Seller files any income, franchise, sales, use or property Tax Returns together with a good faith estimate of the revenues Seller received from each such jurisdiction during the year ended December 31, 2009. To the extent that, in Delaware, Pennsylvania, Alabama or any other taxing jurisdiction, with respect to which Seller has paid or is required to pay that jurisdiction an amount more than \$5,000 with respect to income, franchise, sales, use or property, Taxes with respect to calendar year 2009 (each such jurisdiction a "Taxing Jurisdiction"), any tax clearance or other procedure is available for obtaining certification that all of such Taxes have been paid to that Taxing Jurisdiction, Seller shall apply for such tax clearance certificates and obtain preliminary or conditional tax clearances or such similar assurance as may be available in that Taxing Jurisdiction as to the amount of any Tax due in that Taxing Jurisdiction. In addition, Seller shall conduct lien searches in Delaware, Pennsylvania and Alabama not less than 30 days after the Closing Date nor more than 60 days after the Closing Date. The Seller shall, prior to Closing, provide Buyer with a copy of all such applications to be made by the Seller and any preliminary or conditional tax clearances or similar assurance received by Seller. In the event that any Taxing Jurisdiction to which application is made pursuant to this Section 5.3(h) indicates that any Taxes of Seller are due and owing or the above referenced preliminary or conditional tax clearances or the lien searches otherwise indicate that any Taxes of Seller are due and owing, a portion of the Escrow Amount equal to the amount of any such Taxes shall remain in the escrow and shall not be released to Seller until such time as both of the following are true: (i) the amount would otherwise be released from the escrow but for this Section 5.3(h), and (ii) Seller provides Buyer with (A) a copy of the check or wire transfer in the amount of such Taxes delivered to the Taxing Jurisdiction, (B) other evidence, reasonably satisfactory to Buyer, that the amount of such Taxes has been paid in full or (C) a statement from the Taxing Jurisdiction reasonably satisfactory to Buyer (such as a tax clearance certificate) that such Taxes are not due and owing.

51

5.4 <u>Treatment of Payments.</u>

The Buyer and the Seller will treat any amounts payable after the Closing by the Seller to the Buyer (or by the Buyer to the Seller) pursuant to this Agreement as an adjustment to the Purchase Price, unless a final determination by the appropriate Taxing Authority or court causes any such payment not to be treated as an adjustment to the Purchase Price for Tax purposes.

5.5 <u>Employees; Employment Matters.</u>

- (a) Employees. For purposes of this Agreement, a "Huntsville Employee" means (i) any employee of the Seller with respect to the Huntsville Business who are still employed on the Closing Date, including the following employees, each of whom will be listed on Schedule 5.5(a)(i), which shall be delivered to the Buyer at least five Business Days prior to the Closing Date: (A) who are temporarily absent due to the Family Medical Leave Act, military or other approved leave or absence in compliance with the applicable written policies of the Seller or applicable Law; (B) who are on short term or long term disability leave; or (C) who are receiving workers' compensation payments as required by Law and have the right to re-employment in accordance with applicable Law. Subject to Section 5.5(e) and subject to Buyer's standard hiring practices and procedures, on or prior to the Closing Date, the Buyer shall, make offers of ongoing employment (which offers shall be made in compliance with applicable Law) at substantially the same level of compensation as in effect immediately prior to the Closing to all Huntsville Employees (conditioned upon the Closing) other than those persons set forth on Schedule 5.5(a) (ii), employees on lay-off, suspension or who have been terminated prior to the Closing and other than Persons who have received notice of termination prior to the Closing and who are still employed on the Closing Date. All Huntsville Employees described in the immediately preceding sentence who accept the Buyer's offer of employment as of the Closing Date are hereinafter referred to collectively as the "Transferred Employees".
- (b) <u>Cessation of Active Participation in Benefit Plans</u>. Effective as of the Closing Date, all Transferred Employees will cease active participation in, and any benefit accrual under, each of the Benefit Plans, except as required by Law.
- (c) <u>No Right to Employment</u>. Nothing herein expressed or implied shall confer upon any of the employees of the Seller, the Buyer or any of their respective Affiliates any additional rights or remedies, including any additional right to employment, or continued employment for any specified period, of any nature or kind whatsoever under or by reason of this Agreement.
- (d) <u>No Third Party Beneficiary.</u> No provision in this Section 5.5 shall (i) create any third party beneficiary or other rights in any employee or former employee (including any beneficiary or dependent thereof) of the Seller, the Buyer or any other Person other than the parties hereto and their respective successors and permitted assigns, (ii) constitute or create, or be deemed to constitute or create, an employment agreement or (iii) constitute or be deemed to constitute an amendment to any employee benefit plan sponsored or maintained by the Seller, the Buyer or any of their respective Affiliates.

52

(e) <u>No COBRA Successor Liability</u>. Buyer shall not be a "successor employer" of Seller or any Affiliate of Seller under COBRA with respect to any current or former Benefit Plan or any current or former practice by Seller or any Affiliate of Seller to provide or make available group health plan coverage to any former employee or non-employee service provider or other non-employee individual (and their spouses and dependents).

5.6 Labor Matters.

The Seller shall take, or cause to be taken, any and all actions in connection with any required notification to, or any required consultation with, the employees and relevant government agencies concerning the transactions contemplated by this Agreement with respect to the employees of the Seller, and the Buyer will reasonably cooperate with the Seller in connection with the foregoing, including by providing any such notification requested by the Seller,

whether before or after the Closing. Set forth on <u>Schedule 5.6</u> is a list of all Employees of Seller terminated by Seller within the period beginning 30 days prior to the date of this Agreement and ending on the Closing Date.

5.7 <u>Contact With Customers and Suppliers.</u>

Prior to the Closing, the Seller shall use commercially reasonable efforts to cooperate with and assist the Buyer in preserving the business organization and operations of the Huntsville Business and the goodwill of those having business relationships with the Huntsville Business. Consistent with applicable Competition/Foreign Investment Laws, upon the Seller's prior written consent (which the Seller may not unreasonably withhold, but which the Seller may condition upon a designee of the Seller being present at any meeting or conference referenced in this sentence) the Buyer and its representatives shall be permitted to contact and communicate with the employees, customers, suppliers and licensors of the Huntsville Business in connection with the transactions contemplated by this Agreement.

5.8 <u>Non-Solicitation by Buyer and its Affiliates.</u>

For a period of two years following the Closing Date, without the prior written consent of TW International, the Buyer shall not, and shall cause its Affiliates not to, solicit or hire any employee of the TW International or its Affiliates, excluding the Huntsville Employees, for employment or consulting services by the Buyer or any of its Affiliates. An employee shall be deemed not to have been solicited for employment if such employee responded to a general solicitation. Seller and TW International will cause their sales agent Richardo Fechi (A) to provide Buyer (either through TW International or directly) with the names and contact information of his contacts at customers for sale of Products and (B) personally introduce Buyer's designated sales representative to each customer with respect to whom Mr. Fechi has been Seller's representative with respect to the Products, at such reasonable times as may be mutually convenient to Mr. Fechi and Buyer's designee.

5.9 Corporate Names.

(a) The Buyer shall remove or cover the names "Taylor-Wharton", "TW" and any trademarks, corporate names, trade names, logos, domain names, brandmarks, brand names

53

or trade dress relating to such names, including those set forth on subsection 3.11(m) of Schedule 3.11, from all: (i) invoices, sales acknowledgement forms and other shipping documents (including bills of lading, packing lists and export documents) of the Sold Assets, the Huntsville Business and/or the Buyer no later than 90 days after the Closing Date, unless such period is extended with the consent of the Seller, such consent not to be unreasonably withheld; and (ii) signage, letterhead (including internal memo forms and fax forms), envelopes, business cards, sales literature, exhibits and displays and promotional items of the Sold Assets, the Huntsville Business and/or the Buyer no later than 90 days after the Closing Date, unless such period is extended with the consent of the Seller, such consent not to be unreasonably withheld. The Buyer shall have the right to sell products manufactured prior to Closing (including identification plates) and packaging (including shipping boxes and packaging materials) bearing the name "Taylor-Wharton" or any abbreviation to exceed 90 days following the Closing Date, as reasonably required, to exhaust the inventory of such products and packaging existing as of the Closing Date. The Buyer shall have the right to use product instruction manuals and instruction sheets bearing the name "Taylor-Wharton" or any abbreviation thereof (including "TW") following the Closing to exhaust the inventory of such instruction manuals and instruction sheets existing as of the Closing Date. Except as provided in this Section 5.9, the Buyer shall neither use nor permit the Huntsville Business or any of its Affiliates to use the name "Taylor-Wharton" or any abbreviation thereof (including "TW") or any trademarks, corporate names, trade names, logos, domain names, brandmarks, brand names or trade dress relating or confusingly similar to such names, in connection with the Huntsville Business or otherwise.

(b) As soon as commercially practical and in any event no later than 30 days after the Closing Date, the Seller shall, and shall cause its Affiliates to, immediately cease all use of the trademarks and the domain names included in the Acquired Intellectual Property.

5.10 <u>Further Actions.</u>

Each of the parties hereto shall use commercially reasonable efforts to take, or cause to be taken, all appropriate action, do or cause to be done all things necessary, proper or advisable under applicable Law, and execute and deliver such documents and other papers, as may be reasonably required to consummate the transactions contemplated by this Agreement. Without limiting the generality of the foregoing, the Seller agrees that it shall use its best commercially reasonable efforts to assist in the transfer to Buyer of all Permits relating to the Huntsville Business and Sold Assets and to obtain prior to the Closing all consents and to deliver prior to the Closing all notices required in connection with the transactions contemplated hereby, including those listed on Schedule 3.5. Notwithstanding anything in this Agreement to the contrary, this Agreement shall not constitute an agreement to sell, convey, assign or transfer any asset if any attempted sale, conveyance, assignment or transfer of such asset, without the consent of another Person to such transfer, would constitute a breach by the Seller or the Buyer with respect to such asset if such consent shall not have been received. If any required consent is not obtained on or prior to the Closing, the Seller shall use its commercially reasonable efforts to (i) provide to the Buyer the material benefits of the applicable contract, agreement, permit or other asset, (ii) cooperate in any reasonable and lawful arrangement designed to provide such material benefits to the Buyer and (iii) enforce at the request of the Buyer and for the account of the Buyer and at the Buyer's expense any rights of Seller arising from any such contract or

54

agreement (including the right to elect to terminate or renew such contract or agreement in accordance with the terms thereof upon the request of the Buyer).

5.11 <u>Elimination of Debt Obligations</u>.

At or prior to the Closing, the Seller shall cause all Debt Obligations, if any, of the Seller that relate to the Sold Assets or Assumed Liabilities (other than letters of credit entered into in the ordinary course of business) to be repaid in full or otherwise satisfied or eliminated without any continuing liability or obligation of the Buyer.

5.12 <u>Bulk Transfer Laws.</u>

The Buyer and the Seller hereby waive compliance with any bulk transfer laws applicable to the transactions contemplated by this Agreement.

5.13 Confidentiality.

- (a) The Buyer hereby confirms and agrees that, with respect to any information directly or indirectly furnished by or on behalf of the Seller or any of its Affiliates, whether before, on or after the date hereof, the Buyer shall continue to be bound by the terms of the Confidentiality Agreement.
- (b) The Buyer understands and agrees that the Seller is making available confidential information and trade secrets to the Buyer concerning the operations of the Seller, which information would be damaging to the Seller and its Affiliates if disclosed to a competitor or made available to any other Person, and that such information has been divulged in confidence. The Buyer acknowledges that after the Closing the Seller and its Affiliates could be irreparably damaged if any nonpublic or proprietary information about the Seller or its Affiliates that does not relate to the Sold Assets or Assumed Liabilities were disclosed by the Buyer or its Affiliates after the Closing to any Person other than the Seller or its Affiliates, and the Buyer will not, and will cause its officers, directors, employees and other Affiliates not to, without the prior written consent of the Seller, disclose or use (or permit to be disclosed or used) in any way any such information, unless (i) compelled to disclose such confidential information by judicial or administrative process or, in the opinion of its counsel, by other requirements of Law and, in any such event, the Buyer shall, to the extent practicable, give the Seller prompt written notice of any such requirement prior to any such disclosure, (ii) such confidential information is generally available to the public through no fault of the Buyer or any of its Affiliates, or (iii) such confidential information is publicly disclosed by the Buyer or one of its Affiliates with the Seller's prior written consent, or (iv) such confidential information is or becomes available to Buyer or one of its Affiliates on a nonconfidential basis from a source other than the Seller or one of its Affiliates, which source is entitled to disclose such information without breach of any obligation of confidentiality known to the Buyer or one of its Affiliates.
- (c) The Seller understands and agrees that the Buyer is making available confidential information to the Seller concerning the finances and operations of the Buyer, which information would be damaging to the Buyer and its Affiliates if disclosed to a competitor or made available to any other Person, and that such information has been divulged in confidence. In addition, from and after the Closing Date, the Seller understands and agrees that the Seller and

55

its Affiliates (i) will not use any confidential information of the Buyer or regarding the Huntsville Business, Sold Assets or Assumed Liabilities for the Seller's or its Affiliates' own purposes or for the purposes of any Person other than the Buyer or its Affiliates and (ii) will not disclose any confidential information of the Buyer or regarding the Huntsville Business, Sold Assets or Assumed Liabilities to any Person other than the Buyer or its Affiliates, unless (1) compelled to disclose such confidential information by judicial or administrative process or, in the opinion of its counsel, by other requirements of Law and, in any such event, the Seller shall, to the extent practicable, give the Buyer prompt written notice of any such requirement prior to any such disclosure, (2) such confidential information is generally available to the public through no fault of the Seller or any of its Affiliates, (3) such confidential information is publicly disclosed by the Seller or one of its Affiliates with the Buyer's prior written consent, or (4) such confidential information is or becomes available to Seller or one of its Affiliates on a nonconfidential basis from a source other than the Seller or one of its Affiliates, which source is entitled to disclose such information without breach of any obligation of confidentiality known to the Seller or one of its Affiliates. From and after the Closing Date, the Seller agrees, and shall cause its Affiliates, to take all reasonable precautions to prevent inadvertent disclosure or use of such confidential information and shall maintain confidential and in complete secret such confidential information, except as set forth above. The Seller understands the importance of this obligation of confidentiality to the Buyer and its Affiliates and acknowledges that the use or disclosure of such confidential information in contravention of this Section 5.13(c) could be damaging to the business or operations of the Buyer or its Affiliates, particularly if such use or disclosure is to

5.14 No Exclusivity.

From the date of the issuance of the Bidding Procedures Order until completion of the Auction, the Seller shall have the right, directly or indirectly:

(a) to solicit, initiate or encourage any inquiry, proposal or offer from any Person relating to any transaction involving the sale of the Sold Assets, all or any material part of the Business (including the Huntsville Business) or the Membership Interests, or any merger, consolidation, business combination, or similar transaction involving the Business or the Seller (an "Acquisition Transaction"); (b) to participate in any discussions or negotiations or enter into any agreement with, or provide any non-public information to, any Person relating to or in connection with a possible Acquisition Transaction or facilitate an Acquisition Transaction in any manner; or (c) to accept any proposal or offer from any Person relating to a possible Acquisition Transaction.

5.15 <u>Non-Solicitation and Non-Competition by Seller and its Affiliates.</u>

(a) From and after the Closing Date until the fifth anniversary of the Closing Date, none of the Seller, TW International or any direct or indirect Subsidiaries of TW International (including new Subsidiaries of TW International acquired while TW International is Controlled by its current beneficial owners) ("Prohibited Parties") shall, anywhere in the world, directly or indirectly, either for themselves or any other Person, own, manage, operate, join, consult with, render services for, participate in, control, finance, permit their name to be used by, or otherwise assist in any manner, in any business that manufactures, engineers, designs, markets or sells (i) steel constructed acetylene cylinders up to 450 cubic foot in capacity, (ii) steel constructed high pressure cylinders up to 150 cubic foot capacity, or (iii) steel

56

constructed high pressure carbon dioxide cylinders up to 20 pounds in capacity (the "Competitive Business").

- (b) The foregoing Section 5.15 shall not apply to, or be deemed to restrict, a Person that operates a Competitive Business prior to acquiring a Prohibited Party from continuing such business after its acquisition of a Prohibited Party; provided that, if the Prohibited Party is acquired via a sale of substantially all of its equity interests or is merged with and into an entity with no prior business operations, the Prohibited Party or surviving entity of such merger shall not directly or indirectly conduct a Competitive Business in violation of Section 5.15.
- (c) From and after the Closing Date until the second anniversary of the Closing Date, none of the Prohibited Parties shall, anywhere in the world, directly or indirectly, either for themselves or any other Person, solicit, attempt to solicit, hire or attempt to hire, directly or indirectly, any

employee of the Buyer without the prior written consent of the Buyer, provided that an employee shall not be deemed solicited for hire by a general, non targeted, solicitation.

(d) The parties hereto specifically acknowledge and agree that the remedy at law for any breach of the provisions of this Section 5.15 will be inadequate and that the Buyer, in addition to any other relief available to it, shall be entitled to temporary and permanent injunctive relief, specific performance or other equitable relief without the necessity of proving actual damage. In the event that the provisions of this Section 5.15 should ever be deemed to exceed the limitations provided by applicable Law, then the parties hereto agree that such provisions shall be reformed to set forth the maximum limitations permitted. The Seller acknowledges and agrees that this Section 5.15 is a significant inducement for the Buyer to enter into and perform its obligations under this Agreement and this Section 5.15 is reasonable and necessary to protect and preserve legitimate business interests of the Buyer.

5.16 <u>Bankruptcy Conditions.</u>

(a) Unless the parties otherwise agree, the Seller shall file or cause to be filed (such filing date shall be referred to as the "Sale Motion Filing Date") with the Bankruptcy Court a motion (the "Sale Motion") seeking (i) approval of an order (the "Bidding Procedures Order") that provides for the conduct of the sale of the Sold Assets in accordance with the bidding procedures that will contain such provisions described more fully on Exhibit 5.16(a) (i) (the "Bidding Procedures"); (ii) to schedule a hearing (the "Sale Hearing") for the approval of the sale of the Sold Assets pursuant to the terms and conditions of this Agreement and deadlines for filing and service of objections and responses to the relief requested in the Sale Motion; (iii) to approve the form, manner and sufficiency of notice of the Sale Motion and Sale Hearing to be given and published by the Seller; and (iv) entry of an non-appealable order (the "Sale Order") that: (1) approves the sale of the Sold Assets in accordance with the terms and conditions of this Agreement, (2) authorizes and requires the Seller to sell the Sold Assets pursuant to 11 U.S.C. § 363(b) and pursuant to and in accordance with the terms of this Agreement free and clear of all liens, claims and encumbrances, whether arising pre-petition or post petition, and (3) finds that the Buyer is a good faith purchaser pursuant to 11 U.S.C. § 363(m). The Sale Order shall be in the form attached as Exhibit 5.16(a)(iii), with only such changes as are mutually agreed upon by

57

Buyer and Seller. If this Agreement is filed with the Bankruptcy Court it will be filed without attaching the Exhibits and Schedules.

(b) The Seller's Sale Motion shall contain, and the transactions contemplated by this Agreement shall be subject to, the following provisions:

- (i) <u>Due Diligence by Additional Bidders</u>. Until entry of the Bidding Procedures Order, Seller shall not solicit bids from any other Person and shall not permit any other Person other than Buyer to conduct due diligence in respect of the Huntsville Business, the Sold Assets, the Membership Interests or substantially all of the Seller's assets. From and after the entry of the Bidding Procedures Order, unless otherwise ordered by the Bankruptcy Court for cause shown, if any Person other than the Buyer (a "<u>Potential Bidder</u>") delivers to the Seller a request for information or a proposal for the purchase of the Membership Interests, the Huntsville Business, the Sold Assets or substantially all of the Seller's assets, the Seller shall have the right to permit the Potential Bidder to conduct due diligence with respect to the Membership Interests, the Huntsville Business, the Sold Assets or otherwise with respect to Seller's assets. Such due diligence access may include management presentations as may be scheduled by the Seller, access to data rooms, delivery of written materials by the Seller, on-site inspections by the Potential Bidder or its authorized representatives and such other matters that a Potential Bidder may request and as to which the Seller, in its sole discretion, may accept.
- (ii) Sale Hearing. Subject to the Seller's right to consummate the Plan and insure that the Effective Date thereunder has occurred at any time after June 12, 2010, the Seller will conduct an auction of the Membership Interests, the Huntsville Business, the Sold Assets or substantially all of the Seller's assets, as the case may be (the "Auction") in accordance with the Bidding Procedures Order. Notwithstanding any provision in this Agreement to the contrary, if the successful bidder at the Auction is a Person other than the Buyer, the Seller may proceed to sell the Membership Interests, the Huntsville Business, the Sold Assets, the Business or substantially all of the Seller's assets, as the case may be, to the successful bidder (an "Alternate Transaction"). If the Buyer or another bidder is the successful bidder at the Auction, but the purchase price is higher than that provided for in this Agreement, then this Agreement shall be modified accordingly, as provided in an appropriate order of the Bankruptcy Court authorizing the sale to the Buyer or such other bidder, and the parties will proceed to consummate the transactions contemplated by this Agreement in accordance with the terms hereof, with such changes as shall be necessitated by the applicable order of the Bankruptcy Court to enter the Sale Order and the parties will proceed to consummate the transactions contemplated by this Agreement in accordance with the terms hereof.

If the Seller enters into a written agreement regarding an Alternate Transaction with a Person other than the Buyer or any of its Affiliates before the termination of this Agreement, whether pursuant to a sale, plan of reorganization or otherwise, the Seller shall pay to the Buyer concurrently with the closing of such transaction, a break-up fee in an amount equal to \$352,500

58

(the "Break-up Fee"). If this Agreement is terminated pursuant to Section 8.1(b), 8.1(e), 8.1(g) or 8.1(h), then the Seller shall pay the Buyer an amount equal to the actual costs and expenses reasonably incurred by the Buyer in connection with the negotiation, execution and performance of this Agreement, provided that such amount shall in no event exceed \$250,000 (the "Expense Reimbursement"). In no event will the Buyer be entitled to payment of both the Expense Reimbursement and the Break-up Fee, but, if the Buyer would otherwise be entitled to payment of both the Expense Reimbursement and the Break-up Fee, the Buyer will be paid the greater thereof. Any Break-up Fee or Expense Reimbursement payable to the Buyer pursuant to this section shall be paid as an allowed administrative claim under Section 507(a)(2) of the Bankruptcy Code against the Seller's bankruptcy estate.

(c) Notwithstanding any provision in this Agreement to the contrary, the Seller shall have the right to consummate the Plan and insure that the Effective Date thereunder has occurred at any time after June 12, 2010; provided that such right shall not effect the Seller's obligation to file the Sale Motion within three Business Days after the execution of this Agreement or Buyer's right to terminate this Agreement pursuant to Section 8.1(i) if the Bidding Procedures Order is not approved by the Bankruptcy Court within 20 days after the date Sale Motion is filed; provided further, that the parties agree that Buyer may terminate this Agreement in such event under Section 8.1(b) if the Effective Date of the Plan occurs prior to June 12, 2010 and under Section 8.1(e) if the Effective Date of the Plan occurs after June 12, 2010. Any Break-Up Fee or Expense Reimbursement payable to the Buyer pursuant to

this section shall be paid as an allowed administrative claim under Section 507(a)(2) of the Bankruptcy Code with priority over any and all administrative claims other than claims based on the DIP Obligations and the Carve-Out (each as defined in Final Order (I) Authorizing Debtors (A) To Obtain Postpetition Secured Financing Pursuant to 11 U.S.C. §§ 105, 361, 362, 364(c)(1), 364(c)(2), 364(c)(3), 364(d)(1) and 364(e), and (B) To Utilize Cash Collateral Pursuant To 11 U.S.C. §§ 363, And (II) Granting Adequate Protection To Prepetition Secured Parties Pursuant to 11 U.S.C. §§ 361, 362, 363, and 364) entered by the Bankruptcy Court in the Bankruptcy Case.

5.17 Advice of Changes.

Between the date of this Agreement and the Closing Date or the earlier termination of this Agreement pursuant to ARTICLE VIII, the Seller will promptly notify the Buyer in writing of any fact that, had the fact had existed or actually been known by the persons identified in the definition of Knowledge of the Seller as the date of this Agreement, would have been required to be set forth in this Agreement or disclosed pursuant to this Agreement or which would materially affect or change any of the information set forth in the Disclosure Schedules. If any such fact requires any change to the Disclosure Schedules, then within three Business Days of becoming aware of the fact (but in any event prior to the Closing Date), the Seller shall deliver to the Buyer a supplement to the Disclosure Schedules specifying such change; provided, that any such supplement shall not be taken into account for purposes of determining whether the condition contained in Section 7.1 has been satisfied; and provided further, that if any supplement to the Disclosure Schedules reflects any state of facts, change, occurrence or event that would result in the failure of the condition precedent set forth in Section 7.1, the Buyer may, in accordance with Section 8.1(b), cause this Agreement to be terminated by giving written notice to the Seller. Neither the termination of this Agreement by Buyer pursuant to Section 5.17 shall affect

59

Buyer's rights and remedies with respect to any breach of representations or warranties arising from the matters disclosed under this Section 5.17; provided that if a representation and warranty was true at the time this Agreement was executed but becomes untrue at or prior to the Closing Date as the result of an occurrence which is outside of the Seller's reasonable control and the Buyer elects to terminate this Agreement, Buyer may not also seek indemnification from Seller for breach of such representation and warranty.

5.18 Contracts Related to the Huntsville Business Held by TW International.

To the extent any Contract, Intellectual Property or other asset or right related to, arising out of, used in or held for use in the Huntsville Business is held by or in the name of TW International, TW International will assign such Contract, Intellectual Property or other asset or right to the Seller prior to the Closing, and such Contract will be deemed to be a Contract, Intellectual Property right or other asset or right held by the Seller in the name of the Seller for all purposes of this Agreement.

5.19 <u>Servicing Warranty Claims.</u>

Buyer shall, at Seller's request and at Seller's sole cost and expense, use Buyer's commercially reasonable efforts to make any repairs to, or provide replacements for, Products that are the responsibility of the Seller (but not Products manufactured at a location other than the Seller Owned Real Property); provided that this obligation of Buyer is conditioned upon Buyer having adequate capacity and resources and the necessary tooling and equipment. Buyer shall not charge Seller an amount greater than Buyer's cost of providing such repairs or replacement Products. For purposes of this Section 5.19, Buyer's cost will include to the extent incurred by Buyer and not paid directly by Seller: all out-of-pocket costs, return freight charges, the cost of producing a replacement cylinder or to otherwise repair such cylinder based on Buyer's then current standard costs for material, labor (including overtime), overhead incurred to produce such replacement cylinder or to perform such repair, any refund to customer for return and cancellation, and/or any customer related additional charges associated with the warranty claim and return freight to customer. Repair services are understood to include, but not be limited to, repainting, retesting hydro, re-valve (includes new valve at current cost/price), repair threads, internal/ external shot blast and re-Stamp / buffing head markings. Specific costs which are outside of the ordinary practice of Buyer for addressing a given type of claim must be approved in advance by Seller, and if not approved, Buyer shall not be required to incur such cost or entitled to reimbursement for such cost under this Section 5.19.

5.20 <u>Harrisburg Assets.</u>

The Seller will not allow any Person to operate the forge or draw bench included in the Harrisburg Assets and will permit to have access to the building in which the Harrisburg Assets are located only Buyer, Harsco Corporation, their respective employees, representatives and agents, and such other Persons who need such access for purposes of preparing any assets for removal from the Seller Leased Real Property or for purposes of conducting an evaluation of the assets located at the Seller Leased Real Property in connection with such other Person's participation in an auction with respect to any such assets. All such access will be supervised by the Seller or its authorized employees, representatives or agents.

60

ARTICLE VI. CONDITIONS TO THE SELLER'S OBLIGATIONS

The obligation of the Seller to effect the Closing under this Agreement is subject to the satisfaction, at or prior to the Closing, of each of the following conditions (any and all of which may be waived by the Seller in whole or in part to the extent permitted by applicable Law); provided, however, that the Seller may not rely on the failure of any condition set forth in this ARTICLE VI if such failure was caused by the Seller's failure to comply with any provision of this Agreement:

6.1 <u>Representations and Warranties.</u>

The representations and warranties made by the Buyer in this Agreement shall be true and correct (provided that any representation and warranty contained herein that is subject to a materiality, material adverse effect or similar qualification will not be so qualified for purposes of determining the existence of any breach thereof on the part of the Buyer) as of the Closing Date as though such representations and warranties were made at such date (except to the extent such representations and warranties are made as of a specified date, which representations and warranties, subject to the elimination of materiality qualifications as provided in the parenthetical above, shall be true and correct as of such earlier date), except for such breaches that would not,

individually or in the aggregate with any other breaches of representations and warranties on the part of the Buyer, reasonably be expected to materially and adversely affect the ability of the Buyer to consummate the transactions contemplated by this Agreement.

6.2 <u>Performance.</u>

The Buyer shall have performed and complied in all material respects with all agreements and obligations required by this Agreement to be so performed or complied with by it prior to the Closing and shall have delivered the items in Section 2.9(b) hereof.

6.3 Governmental Orders.

At the Closing, (a) there shall not be in effect any Governmental Order restraining, enjoining or otherwise prohibiting the transactions contemplated hereby and (b) if the Seller is subject to the jurisdiction of the Bankruptcy Court as of the Closing, the Bankruptcy Court shall have issued a final, nonappealable Sale Order in form and substance reasonably acceptable to the parties hereto that, among other things (i) approves and authorizes the sale of the Sold Assets to the Buyer in accordance with the terms and conditions of this Agreement, and (ii) provides for the assignment of this Agreement to the reorganized Seller and the reorganized TW International and the assumption of this Agreement by the reorganized seller and reorganized TW International.

ARTICLE VII. CONDITIONS TO THE BUYER'S OBLIGATIONS

The obligation of the Buyer to effect the Closing under this Agreement is subject to the satisfaction, at or prior to the Closing, of each of the following conditions (any and all of which may be waived by the Buyer in whole or in part to the extent permitted by applicable Law);

61

<u>provided</u>, <u>however</u>, <u>that</u> the Buyer may not rely on the failure of any conditions set forth in this ARTICLE VII if such failure was caused by the Buyer's failure to comply with any provision of this Agreement:

7.1 <u>Representations and Warranties.</u>

The representations and warranties made by the Seller in this Agreement shall be true and correct (provided that any representation or warranty of the Seller contained herein that is subject to a materiality, Material Adverse Effect, material adverse effect or similar qualification will not be so qualified for purposes of determining the existence of any breach thereof on the part of the Seller) as of the Closing Date as though such representations and warranties were made at such date (except to the extent such representations and warranties are made as of a specified date, which representations and warranties, subject to the elimination of any materiality qualifications as provided in the parenthetical above, shall be true and correct as of such earlier date), except for such breaches that would not, individually or in the aggregate with any other breaches of representations and warranties on the part of the Seller, reasonably be expected to have a Material Adverse Effect.

7.2 <u>Performance.</u>

The Seller shall have performed and complied in all material respects with all agreements and obligations required by this Agreement to be performed or complied with by it prior to the Closing and shall have delivered the items in Section 2.9(a) hereof.

7.3 Governmental Orders.

At the Closing, (a) there shall not be in effect any Governmental Order restraining, enjoining or otherwise prohibiting the transactions contemplated hereby and (b) if the Seller is subject to the jurisdiction of the Bankruptcy Court as of the Closing, the Bankruptcy Court shall have issued a final, nonappealable Sale Order approving and authorizing the sale of the Sold Assets to the Buyer in accordance with the terms and conditions of this Agreement.

7.4 <u>Plan not Effective.</u>

The Effective Date of the Plan shall not have occurred.

7.5 <u>Key Employee.</u>

The Buyer and Mike Camp shall have reached agreement on the terms of Mike Camp's employment with the Buyer in accordance with a mutually acceptable offer letter.

7.6 No Material Adverse Effect.

Since December 31, 2009, there shall not have been any change in or with respect to the Huntsville Business or the Sold Assets which constitutes, or could reasonably be expected to constitute, a Material Adverse Effect.

62

7.7 <u>Assumption and Assignment of Sold Contracts.</u>

Unless the Closing will occur after the Effective Date of the Plan, Seller shall have obtained a final and non-appealable order of the Bankruptcy Court in the Bankruptcy Case, in form and substance reasonably acceptable to the Buyer, providing for the assumption and assignment, pursuant to section 365 of the Bankruptcy Code, of the Sold Contracts.

7.8 <u>Approval of the Break-up Fee and Expense Reimbursement.</u>

Entry of the Bidding Procedures Order and approval of the Break-up Fee and Expense Reimbursement.

7.9 Satisfaction with Certain Due Diligence Items.

The Buyer shall have been satisfied in its reasonable discretion with the results of the Phase II Review, which it will commence promptly following the date the bankruptcy Court enters on its docket the Bidding Procedures Order; provided that, if the Buyer does not provide notice to the Seller that it desires to terminate this Agreement as a result of the failure of this condition on or before the date that is 15 Business Days after the date that the Bankruptcy Court enters on its docket the Bidding Procedures Order, the Buyer shall be deemed to have irrevocably waived this condition.

ARTICLE VIII. TERMINATION

8.1 <u>Termination.</u>

This Agreement may be terminated at any time prior to the Closing:

- (a) by the mutual written consent of the Seller and the Buyer;
- (b) by the Buyer, at any time prior to the Closing in the event that the Seller is in breach of any representation, warranty or covenant made by it in this Agreement and such breach renders the conditions set forth in ARTICLE VII incapable of being satisfied absent a written waiver of such conditions by the Buyer;
- (c) by the Buyer, at any time prior to the expiration of the 15 Business Day period commencing on the date that the Bankruptcy Court enters on its docket the Bidding Procedures Order, if the condition contained in Section 7.9 has not been satisfied by such date;
- (d) by the Seller, at any time prior to the Closing in the event that the Buyer is in breach of any representation, warranty or covenant made by it in this Agreement and such breach renders the conditions set forth in ARTICLE VI incapable of being satisfied absent a written waiver of such conditions by the Seller;
- (e) by the Seller or the Buyer if the Closing has not occurred on or before the later of (i) June 12, 2010 or (ii) the first to occur of the Effective Date of the Plan or July 31,

63

2010, unless the failure of such consummation is due to the failure of the party attempting to terminate to comply in all material respects with the agreements and covenants contained herein;

- (f) by either the Seller or the Buyer if any Governmental Authority of competent jurisdiction shall have issued an order, decree or ruling or taken any other action, which order, decree or ruling or other action shall have become final and nonappealable, (i) restraining, enjoining or otherwise prohibiting the transactions contemplated hereby, or (ii) approving or authorizing the sale of the Sold Assets, substantially all of the assets of the Seller or the Membership Interests to another Person;
- (g) by the Buyer if a motion to dismiss the Bankruptcy Case or a motion to convert the Bankruptcy Case to a case under Chapter 7 of the Bankruptcy Code or the appointment of a trustee, receiver, liquidator or other similar person for the purpose of liquidating any of the Sold Assets other than pursuant to this Agreement is granted in the Bankruptcy Case;
- (h) by Buyer if the Seller fails to file the Sale Motion within three (3) Business Days after the execution of this Agreement; provided that the Buyer must deliver its written notice of termination pursuant to this subsection prior to the date that the Seller files the Sale Motion;
- (i) by the Buyer if the Bidding Procedures Order is not approved by the Bankruptcy Court within 20 days after the date Sale Motion is filed with the Bankruptcy Court; provided that Buyer must deliver its written notice of termination pursuant to this subsection prior to the date that the Bidding Procedures Order is approved by the Bankruptcy Court; or
- (j) by the Buyer if the provisions of this Agreement with respect to the Break-Up Fee and the Expense Reimbursement are not approved by the Bankruptcy Court within 20 days after the date the Seller files the Sale Motion is filed with the Bankruptcy Court; provided that the Buyer must deliver its written notice of termination pursuant to this subsection prior to the date that the Break-Up Fee and the Expense Reimbursement are approved by the Bankruptcy Court.

This Agreement shall terminate without any further action by any party hereto upon the approval by the Bankruptcy Court of an Alternative Transaction.

8.2 Procedure and Effect of Termination.

- (a) A party desiring to terminate this Agreement pursuant to Section 8.1 must give written notice of such termination to the other party in accordance with Section 10.7, specifying the provision hereof pursuant to which such termination is effective. If this Agreement is terminated as provided herein:
 - (i) the Buyer will redeliver to the Seller all documents, work papers and other material of the Seller relating to the transactions contemplated hereby, whether so obtained before or after the execution hereof;

- (ii) the provisions of the Confidentiality Agreement shall continue in full force and effect in accordance with its terms; and
- (iii) no party to this Agreement shall have any liability under this Agreement to any other party except (A) nothing herein shall relieve any party from any liability for any willful breach of any of the representations or warranties or breach of any covenants or agreements set forth in this Agreement (and the damages recoverable by the non-breaching party shall include all attorneys' fees reasonably incurred by such party in connection with the transactions contemplated by this Agreement), (B) the payment of the Expense Reimbursement or Break-up Fee as contemplated by Section 5.17 and (C) as contemplated by Section 5.14 and by clause (ii) above.

ARTICLE IX. INDEMNIFICATION

9.1 <u>Indemnification by the Seller.</u>

- (a) From and after the Closing, the Seller and TW International, jointly and severally, shall indemnify, defend, and hold harmless the Buyer and its Affiliates and each of their respective equity holders, officers, directors, managers, members, employees, agents and representatives (collectively, the "Buyer Indemnified Persons") from and against any and all claims, losses, damages, costs and expenses (including reasonable fees and expenses of attorneys) (collectively, "Losses"), that any Buyer Indemnified Person suffers or incurs arising out of or resulting from:
 - (i) any breach or inaccuracy of any representation or warranty made by the Seller or TW International in ARTICLE III;
 - (ii) the failure to perform any covenant or agreement of the Seller or TW International contained in this Agreement; or
 - (iii) any Excluded Liability.
- (b) Notwithstanding anything to the contrary contained in this ARTICLE IX, (i) none of the Buyer Indemnified Persons shall be entitled to recover from the Seller or TW International for any Losses under Section 9.1(a)(i) unless the total of all Losses exceeds \$62,500 (the "Basket"), in which event the Buyer Indemnified Persons will be entitled to indemnification only for such Losses in excess of the Basket; (ii) the Buyer Indemnified Persons shall not be entitled to recover more than an aggregate amount equal to \$2,200,000 (the "Cap" and, together with the Basket, the "Indemnity Limitations") from the Seller with respect to all Losses indemnifiable pursuant to Section 9.1(a)(i); provided however, the Indemnity Limitations shall not apply to breaches, inaccuracies or misrepresentations with respect to any Special Warranties or to any claim based on fraud or intentional misrepresentation.
- (c) Notwithstanding anything in this Section 9.1 or elsewhere in this Agreement to the contrary, except in the case of fraud or an intentional misrepresentation by the Seller in any representation or warranty made by the Seller with respect to the matters set forth in this Section 9.1(c), neither the Seller nor TW International shall be obligated to indemnify,

65

defend, or hold harmless any Buyer Indemnified Person from and against any Losses that any Buyer Indemnified Person suffers or incurs arising out of or resulting from:

- (i) any Liabilities (including with respect to loss of life, personal injury, damage to any real property, Environmental Claims, or natural resource damages) arising out of or resulting from any violation or alleged violation of any Environmental Law, Release of Hazardous Materials into the Environment or storage or off-site disposal of any Hazardous Materials: (1) occurring at or emanating from the Seller Owned Real Property at any time before December 7, 2007, or (2) arising from the operation of the business conducted by Harsco Corporation or any other Person at the Seller Owned Real Property or Seller Leased Real Property at any time before December 7, 2007;
- (ii) any Liabilities arising out of or relating to accidents, occurrences and other incidents (including all Proceedings relating thereto), whether known or unknown and whether or not reported, that result in (1) personal injury, (2) property damage or (3) any other Losses and, in each case, that result from, are caused by or arise out of, or are alleged to have resulted from, been caused by or arisen out of, directly or indirectly, use of, exposure to or otherwise on account of any product manufactured (in whole or in part), sold, shipped or distributed, or any service rendered, (A) by or on behalf of Harsco Corporation or (B) in connection with the operation of the business conducted by Harsco Corporation or any other Person at the Seller Owned Real Property at any time before December 7, 2007; or
- (iii) any Liabilities relating to recalls, product liability claims or warranty claims with respect to products manufactured (in whole or in part), sold or distributed (1) by or on behalf of Harsco Corporation or (2) in connection with the operation of the business conducted by Harsco Corporation or any other Person at the Seller Owned Real Property at any time before December 7, 2007.
- (d) The parties acknowledge and agree that Seller's and TWI's responsibility to indemnify Buyer for Excluded Liabilities shall not be affected by the fact that Seller may have made representations and warranties in this Agreement which have not been breached.

9.2 Indemnification by the Buyer.

From and after the Closing, the Buyer shall indemnify and hold harmless the Seller and its Affiliates and their respective officers, managers, directors, employees, agents and representatives (collectively, the "Seller Indemnified Persons") from and against any and all Losses that any Seller Indemnified Person actually suffers or incurs arising out of or resulting from:

- (a) any material breach, inaccuracy or misrepresentation of any representation or warranty of the Buyer contained in ARTICLE IV;
- (b) the failure to perform any covenant or agreement of the Buyer contained in this Agreement;
- (c) any Assumed Liability; and

(d) the possession, ownership, use, operation and management of the Sold Assets or the Huntsville Business by the Buyer after the

9.3 Exclusive Remedy.

Closing.

The indemnification provided in this ARTICLE IX and the recourse to the Escrow Amount under the Escrow Agreement, subject to the limitations set forth in this Agreement and the Escrow Agreement, shall be the sole and exclusive post-Closing remedies available to any party in connection with any Losses arising out of or resulting from this Agreement, the transactions contemplated hereby, or the Buyer's ownership or operation of the Sold Assets or the Huntsville Business, whether based in contract or tort or otherwise; provided, however, that the provisions of this Section 9.3 shall not prevent or limit a cause of action at Law or in equity (a) to obtain an injunction or injunctions to prevent breaches of this Agreement and to enforce specifically the terms and provisions of this Agreement or (b) based upon fraud or intentional misrepresentation by the Seller of any representation or warranty made by the Seller in ARTICLE III or by the Buyer of any representation or warranty made by the Buyer in ARTICLE IV, nor shall such provision prevent or limit the rights of the parties hereto with respect to Sections 2.5(c) or 2.6(c). Notwithstanding the immediately preceding sentence or anything else in this Agreement to the contrary, if the Buyer is paid the Break-up Fee or Expense Reimbursement, then that amount shall be the Buyer's sole and exclusive remedy in connection with any Losses arising out of or resulting from this Agreement, the transactions contemplated hereby, or the Buyer's ownership or operation of the Sold Assets or the Huntsville Business, whether based in contract or tort or otherwise. Buyer's right of recourse against the Harrisburg Holdback, and the obligation of Seller to pay to Buyer \$125,000, pursuant to the terms of Section 2.9(c) shall be Buyer's sole and exclusive recourse against Seller and TW International with respect to Seller's obligation to provide Buyer access to the Plant to permit Buyer to remove the Harrisburg Assets therefrom and the other obligations of Seller under Section 2.9(c).

9.4 Indemnification Calculations.

The amount of any Losses for which indemnification is provided under this ARTICLE IX shall be computed net of any insurance proceeds received by the Indemnified Party from a third party insurer in connection with such Losses, including insurance proceeds or other amounts included among the Sold Assets under Section 2.1(m). If an Indemnified Party receives insurance proceeds from a third party insurer in connection with Losses for which it has received full indemnification hereunder, such party shall refund to the Indemnifying Party the amount of such insurance proceeds when received, up to the amount of indemnification received, less any increases in insurance premiums that result from the making of such claim. If an Indemnified Party receives insurance proceeds in connection with Losses for which it has received partial indemnification hereunder, such party shall refund to the Indemnifying Party the amount of such insurance proceeds when received, in excess of the amount necessary to provide the Indemnified Party with a full recovery when combined with the partial indemnification hereunder, less any increases in insurance premiums that result from the making of such claim. An Indemnified Party shall use its commercially reasonable efforts to pursue insurance claims with respect to any Losses; provided, however, that the reasonable costs and expenses associated with the pursuit of such insurance claim shall be Losses hereunder. The Buyer and the Seller agree to treat any amounts payable pursuant to this ARTICLE IX as an adjustment to the Purchase Price, unless a

67

final determination by the appropriate Taxing Authority or court causes any such payment not to be treated as an adjustment to the Purchase Price for Tax purposes.

9.5 Survival.

The representations and warranties contained in this Agreement shall survive the Closing Date until the date which is eighteen (18) months after the Closing Date, except that claims based on intentional misrepresentation shall survive indefinitely and the following representations and warranties (the "Special Warranties") shall survive for the following specified periods: (a) the representations and warranties set forth in Sections 3.1, 3.2, 3.3, 3.14(a), 3.15(a) and 3.16 shall survive indefinitely; (b) the representations and warranties set forth in Section 3.13 shall survive the Closing Date until the date which is three years after the Closing Date; and (d) the representations and warranties set forth in Section 3.8, shall survive for the applicable statute of limitations period (taking into account all extensions) plus 60 days. The covenants and agreements contained in this Agreement to be performed following the Closing Date will survive the Closing Date in accordance with their terms or if no survival period is specified, without limit. All claims for indemnification based on a breach of a representation or warranty must be asserted on or prior to the date of the termination of the respective survival periods set forth in this Section 9.5, except such claims may be pursued thereafter if written notice thereof (specifying in reasonable detail the basis for such claim) was duly given within such period. Any claim for indemnification based on a breach of a representation or warranty not made by the Buyer on or prior to the date of termination of the applicable survival period will be irrevocably and unconditionally released and waived, whether or not a longer period would be permitted by applicable Law.

9.6 <u>Notice and Opportunity to Defend.</u>

(a) If there occurs an event which a party asserts is an indemnifiable event pursuant to Section 9.1 or 9.2, the party or parties seeking indemnification (the "<u>Indemnified Party</u>") shall promptly notify the other party or parties obligated to provide indemnification (the "<u>Indemnifying Party</u>"), which notice shall specify the nature and basis of such claim and the amount thereof, to the extent known. If such event involves any claim or the commencement of any action or proceeding by a third Person (a "<u>Third Party Claim</u>"), the Indemnified Party shall give such Indemnifying Party prompt written notice (the "<u>Claim Notice</u>") of such claim or the commencement of such action or proceeding, which notice shall specify the nature and basis of such claim and the amount thereof, to the extent known, and shall be accompanied by copies of all relevant documentation with respect to such claim, including any summons, complaint or other pleadings that may have been served, any written demand or any other relevant document or instrument; <u>provided</u>, <u>however</u>, <u>that</u> the failure to provide such prompt notice will not relieve the Indemnifying Party of its obligations hereunder unless such failure prejudices the Indemnifying Party hereunder. In the case of a Third Party Claim, the Indemnifying Party shall be entitled to assume the defense thereof, with counsel selected by the Indemnifying Party and, after notice from the Indemnifying Party to the Indemnified Party of such election so to assume the defense thereof (an "<u>Indemnification Acknowledgement</u>"), the Indemnified Party in connection with the defense thereof. The

Indemnifying Party and the Indemnified Party agree to cooperate reasonably with each other and their respective counsel in connection with the defense, negotiation or settlement of any such action or asserted liability. Notwithstanding anything else set forth in this Section 9.6, the Indemnified Party shall at all times have the right to participate at its own expense in the defense of such action or asserted liability. If the Indemnifying Party assumes the defense of an action, no settlement or compromise thereof may be effected (i) by the Indemnifying Party without the written consent of the Indemnified Party (which consent shall not be unreasonably withheld or delayed) unless the settlement involves solely money damages and all such relief is paid or satisfied in full by the Indemnifying Party and the Indemnified Party receives a full release from all claimants or (ii) by the Indemnified Party without the consent of the Indemnifying Party. In no event shall an Indemnifying Party be liable for any settlement effected without its written consent.

(b) Notwithstanding anything to the contrary set forth in this Agreement, from and after the time when the aggregate amount of claims paid or potentially payable by an Indemnifying Party under this Agreement, which are subject to the Cap, exceeds or could potentially exceed the Cap based upon claims paid and pending in accordance with this Agreement, the Indemnified Party shall have the right, at its own cost and expense, to jointly control the defense and settlement of any pending indemnification claims the liability for which could potentially result in the aggregate claims exceeding the Cap. Any such time as the aggregate amount of claims subject to the Cap that have been paid or settled (subject to being paid) exceeds the Cap, the Indemnifying Party shall no longer have the right or obligation to participate in the control or defense of such claims.

9.7 Additional Limitations.

- (a) Except for Losses resulting from an action brought by a third party against a Buyer Indemnified Person or a Seller Indemnified Person, no such party shall be entitled to indemnification under ARTICLE IX for punitive damages, or for lost revenues, income or profits, consequential, incidental, exemplary or special damages.
- (b) Notwithstanding anything in this ARTICLE IX to the contrary, no Buyer Indemnified Person is entitled to make any claim under any provision of this Agreement for reimbursement or indemnification for any Losses pursuant to this ARTICLE IX to the extent such Losses have been reflected in the adjustment to the Purchase Price pursuant to Section 2.4 or Section 2.5. In addition, no party shall be entitled to be compensated more than once for the same Loss.

9.8 Subrogation.

Nothing in this Agreement shall limit or be construed to limit the right of the Seller to assert any claims, demands or rights by subrogation against appropriate Persons (other than a Buyer Indemnified Person) for any amounts paid or reimbursed in respect of Losses successfully asserted by a Buyer Indemnified Person pursuant to Section 9.1.

69

ARTICLE X. MISCELLANEOUS

10.1 <u>Fees and Expenses.</u>

Except as otherwise provided in this Agreement, each party hereto shall bear its own expenses and the expenses of its Affiliates in connection with the preparation and negotiation of this Agreement and the consummation of the transactions contemplated by this Agreement.

10.2 <u>Governing Law; Jurisdiction.</u>

The laws of the State of Delaware, without regard to principles of conflicts of laws, will govern this Agreement and its subject matter, construction and the determination of any rights, duties or remedies of the Parties arising out of or relating to this Agreement, its subject matter or any of the transactions contemplated by this Agreement. Without limiting any party's right to appeal any Order of the Bankruptcy Court, (i) the Bankruptcy Court shall retain exclusive jurisdiction to enforce the terms of this Agreement and to decide any claims or disputes which may arise or result from, or be connected with, this Agreement, any breach or default hereunder, or the transactions contemplated hereby, and (ii) any and all proceedings related to the foregoing shall be filed and maintained only in the Bankruptcy Court, and the parties hereby consent to and submit to the jurisdiction and venue of the Bankruptcy Court and shall receive notices at such locations as indicated in Section 10.7; provided, however, that if the Bankruptcy Case has closed, the Parties agree to unconditionally and irrevocably submit to the exclusive jurisdiction of any state or Federal court located in the State of Delaware with respect to any case or controversy arising out of or relating to this Agreement, its subject matter or any of the transactions contemplated by this Agreement and that all litigation arising out of or relating to this Agreement, its subject matter or any of the transactions contemplated by this Agreement will be commenced in the State of Delaware. The Parties agree that any process, summons, notice or document sent by U.S. registered or certified mail addressed to a party shall be effective service of process for any action, suit or proceeding brought against it in any such court. The Parties irrevocably and unconditionally waive any objection to the laying of venue of any such suit, action or proceeding brought in any such court has been brought in an inconvenient forum. The Parties agree that a final judgment in any suc

10.3 <u>Certain Interpretive Matters.</u>

- (a) Unless otherwise expressly provided, for purposes of this Agreement, the following rules of interpretation shall apply:
- (i) When calculating the period of time before which, within which or following which, any act is to be done or step taken pursuant to this Agreement, the date that is the reference date in calculating such period shall be excluded. If the last day of such period is a non-Business Day, the period in question shall end on the next succeeding Business Day.

- (ii) Any reference in this Agreement to \$ shall mean U.S. dollars.
- (iii) The Exhibits and Disclosure Schedules to this Agreement are hereby incorporated and made a part hereof and are an integral part of this Agreement. All Exhibits and Disclosure Schedules annexed hereto or referred to herein are hereby incorporated in and made a part of this Agreement as if set forth in full herein. Any matter or item disclosed on one Disclosure Schedule shall be deemed to have been disclosed on each other Disclosure Schedule to which it would be applicable by its terms. No disclosure on a Disclosure Schedule relating to a possible breach or violation of any contract, Law or Governmental Order shall be construed as an admission or indication that such breach or violation exists or has actually occurred. Any capitalized terms used in any Disclosure Schedule or Exhibit but not otherwise defined therein shall be defined as set forth in this Agreement.
 - (iv) Any reference in this Agreement to gender shall include all genders.
- (v) The provision of a Table of Contents, the division of this Agreement into Articles, Sections and other subdivisions and the insertion of headings are for convenience of reference only and shall not affect or be utilized in construing or interpreting this Agreement. All references in this Agreement to any "Section" are to the corresponding Section of this Agreement unless otherwise specified.
- (vi) The words such as "herein," "hereinafter," "hereof" and "hereunder" refer to this Agreement as a whole and not merely to a subdivision in which such words appear unless the context otherwise requires.
- (vii) The word "including" or any variation thereof means (unless the context of its usage otherwise requires) "including, without limitation" and shall not be construed to limit any general statement that it follows to the specific or similar items or matters immediately following it.
- (viii) For purposes of this Agreement, the term "commercially reasonable efforts" shall not be deemed to require any Person to give any guarantee or other consideration of any nature, including in connection with obtaining any consent or waiver, or to consent to any change in the terms of any agreement or arrangement.
- (b) The parties hereto have participated jointly in the negotiation and drafting of this Agreement and, in the event an ambiguity or question of intent or interpretation arises, this Agreement shall be construed as jointly drafted by the parties hereto and no presumption or burden of proof shall arise favoring or disfavoring any party by virtue of the authorship of any provision of this Agreement.

10.4 Amendment.

This Agreement may not be amended, modified or supplemented except upon the execution and delivery of a written agreement executed by the parties hereto.

71

10.5 <u>Assignment.</u>

This Agreement will bind and inure to the benefit of the parties and their successors, assigns and transferees; provided that no party hereto may assign this Agreement, any Ancillary Agreement or any rights under this Agreement or any Ancillary Agreement to any other person without the prior written consent of the other party, except that Buyer may assign this Agreement to any Subsidiary of the Buyer or to any lender to the Buyer as security for obligations to such lender in respect of the financing arrangements entered into in connection with the transactions contemplated hereby and any refinancings, extensions, refundings or renewals thereof; provided, further, that no such assignment to any Subsidiary of the Buyer or to such lender to the Buyer shall in any way affect or otherwise limit the Buyer's obligations or liabilities under this Agreement and the Buyer shall remain liable with respect to all of its Liabilities under this Agreement. Any attempted assignment in violation of this Section 10.5 will be null and void. Notwithstanding anything herein to the contrary, the Buyer may assign any of its right to indemnification or reimbursement hereunder or under any of the Ancillary Agreements to any Person that is not an Affiliate of the Buyer upon a sale or transfer of all or substantially all of the assets of the Buyer or any division of the Buyer to that Person.

10.6 Waiver.

Any of the terms or conditions of this Agreement that may be lawfully waived may be waived in writing at any time by the party that is entitled to the benefits thereof. Any waiver of any of the provisions of this Agreement by any party hereto shall be binding only if set forth in an instrument in writing signed on behalf of such party. No failure to enforce any provision of this Agreement shall be deemed to or shall constitute a waiver of such provision and no waiver of any of the provisions of this Agreement shall be deemed to or shall constitute a waiver of any other provision hereof (whether or not similar) nor shall such waiver constitute a continuing waiver. The representations and warranties of the Seller shall not be affected or deemed waived by reason of any investigation made by or on behalf of the Buyer (including, but not limited to, by any of its advisors, consultants or representatives) or by reason of the fact that the Buyer or any of such advisors, consultants or representatives knew or should have known that any such representation or warranty is or might be inaccurate.

10.7 Notices.

Any notice, demand, or communication required or permitted to be given by any provision of this Agreement shall be deemed to have been sufficiently given or served for all purposes if (a) personally delivered, (b) sent by a nationally recognized overnight courier service to the recipient at the address below indicated or (c) delivered by facsimile which is confirmed in writing by sending a copy of such facsimile to the recipient thereof pursuant to clause (a) or (b) above:

Norris Cylinder Company 4818 W. Loop 281 Longview, TX 75603 Attn: Jerry Van Auken, President

Facsimile: 903-237-7657

With a copy to:

TriMas Corporation 39400 Woodward Avenue Suite 130 Bloomfield Hills, MI 48304 Attn: Joshua A. Sherbin, Esq. Facsimile: (248) 631-5502

and

Honigman Miller Schwartz and Cohn LLP 2290 First National Building 660 Woodward Avenue Detroit, Michigan 48226 Attn: Donald J. Kunz, Esq. Facsimile: (313) 465-7455

If to the Seller:

TW Cylinder LLC 4718 Old Gettysburg Rd., Suite 300 Mechanicsburg, PA 17055 Attn: Chief Financial Officer Facsimile: 610-398-0659

With a copy to:

Taylor-Wharton International LLC 4718 Old Gettysburg Rd., Suite 300 Mechanicsburg, PA 17055 Attn: General Counsel

Facsimile: 717-731-7988

73

And:

McNees Wallace & Nurick LLC 100 Pine Street P.O. Box 1166 Harrisburg, PA 17108 Attn: Jeffrey J. van Bastelaar Facsimile: 717-260-1751

or to such other address or facsimile number as any party hereto may, from time to time, designate in a written notice given in like manner. Except as otherwise provided herein, any notice under this Agreement will be deemed to have been duly given (x) on the date such notice is personally delivered or delivered by facsimile or (y) the next succeeding Business Day after the date such notice is delivered to the overnight courier service if sent by overnight courier; provided that in each case notices received after 4:00 p.m. (local time of the recipient) shall be deemed to have been duly given on the next Business Day.

10.8 <u>Complete Agreement.</u>

This Agreement (including the Disclosure Schedules and Exhibits hereto), the Confidentiality Agreement and the Ancillary Agreements contain the entire understanding of the parties with respect to the subject matter hereof and supersede all prior agreements and understandings, both written and oral, between the parties with respect to the subject matter hereof.

10.9 <u>Counterparts.</u>

This Agreement may be executed in one or more counterparts, all of which shall be considered one and the same agreement and each of which shall be deemed an original.

10.10 Publicity.

The Seller and the Buyer will consult with each other and will mutually agree upon any publication or press release of any nature with respect to this Agreement or the transactions contemplated hereby and shall not issue any such publication or press release prior to such consultation and agreement except as may be required by applicable Law or by obligations pursuant to any listing agreement with any securities exchange or any securities exchange regulation,

in which case the party proposing to issue such publication or press release shall make all commercially reasonable efforts to consult in good faith with the other party before issuing any such publication or press release and shall provide a copy thereof to the other party prior to such issuance.

10.11 Severability.

Any provision of this Agreement that is invalid, illegal or unenforceable in any jurisdiction shall, as to that jurisdiction, be ineffective to the extent of such invalidity, illegality or unenforceability, without affecting in any way the remaining provisions hereof in such

74

jurisdiction or rendering that or any other provision of this Agreement invalid, illegal or unenforceable in any other jurisdiction. Upon such determination that any term or other provision is invalid, illegal or incapable of being enforced, the parties hereto shall negotiate in good faith to modify this Agreement so as to effect the original intent of the parties as closely as possible in an acceptable manner in order that the transactions contemplated hereby are consummated as originally contemplated to the greatest extent possible.

10.12 Third Parties.

Except as provided in ARTICLE IX, nothing herein expressed or implied is intended or shall be construed to confer upon or give to any Person, other than the parties hereto and their permitted successors or assigns, any rights or remedies under or by reason of this Agreement.

10.13 Non-Recourse.

Except to the extent provided in this Agreement with respect to TW International, no past, present or future director, manager, partner, officer, employee, incorporator, member, stockholder, agent, attorney or representative of the Seller or its Affiliates, on the one hand, or Buyer or its Affiliates, on the other hand, shall have any liability for any obligations or liabilities of the Seller or Buyer, as applicable, under this Agreement or the Ancillary Agreements or for any claim based thereon, in respect thereof, or by reason thereof.

10.14 No Successor Liability.

Upon the Closing, it is the intent of the parties that the Buyer shall not be deemed: (a) to be the successor of the Seller or any of its Affiliates (including, but not limited to, any status as a "successor employer" of Seller or any Affiliate of Seller under COBRA with respect to any current or former Benefit Plan or any current or former practice by Seller or any Affiliate of Seller to provide or make available group health plan coverage to any former employee or non-employee service provider or other non-employee individual (and their spouses and dependents)); (b) to have, de facto, or otherwise, merged with or into the Seller or any of its Affiliates; (c) to be a mere continuation or substantial continuation of the Seller or any of its Affiliates or the enterprise(s) of the Seller or any of its Affiliates; or (d) to be liable for any acts or omissions of the Seller or any of its Affiliates in the conduct of the Business or arising under or related to the Sold Assets, other than as expressly set forth in this Agreement. Without limiting the generality of the foregoing, and except as otherwise provided in this Agreement, it is the intention of the parties that the Buyer shall not be liable for any claims against the Seller or its Affiliates or any of their predecessors, and the Buyer shall have no successor or vicarious liability of any kind or character whether known or unknown as of the Closing, whether now existing or hereafter arising, or whether fixed or contingent, with respect to the Business or any obligations of the Seller or its Affiliates arising prior to the Closing of the Transactions, except as expressly provided in this Agreement, including, but not limited to, Liabilities on account of any Taxes arising, accruing, or payable under, out of, in connection with, or in any way relating to the operation of the Business prior to the Closing.

(Signature page follows)

75

IN WITNESS WHEREOF, each of the parties hereto has caused this Agreement to be executed by its duly authorized officer, in each case as of the date first above written.

TW CYLINDERS LLC

By: /s/ Bill Corbin
Name: Bill Corbin
Title: Chairman, C.E.O.

NORRIS CYLINDER COMPANY

By: /s/ Jerry Van Auken
Name: Jerry Van Auken

Title: President

Solely with respect to Sections 2.6, 3.1(b), 3.2(b), 3.3, 5.8, 5.18 and ARTICLE IX:

TAYLOR-WHARTON INTERNATIONAL LLC

By: /s/ Bill Corbin
Name: Bill Corbin
Title: Chairman, C.E.O.



Certification Pursuant to Section 302 of The Sarbanes-Oxley Act of 2002 (Chapter 63, Title 18 U.S.C. Section 1350(A) and (B))

I, David M. Wathen, certify that:

- 1. I have reviewed this quarterly report on Form 10-Q of TriMas Corporation;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I am responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
 - a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared; and
 - b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting.
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: April 30, 2010

/s/ DAVID M. WATHEN

David M. Wathen
Chief Executive Officer

QuickLinks

Exhibit 31.1

Certification Pursuant to Section 302 of The Sarbanes-Oxley Act of 2002 (Chapter 63, Title 18 U.S.C. Section 1350(A) and (B))

Certification Pursuant to Section 302 of The Sarbanes-Oxley Act of 2002 (Chapter 63, Title 18 U.S.C. Section 1350(A) and (B))

I, A. Mark Zeffiro, certify that:

- 1. I have reviewed this quarterly report on Form 10-Q of TriMas Corporation;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I am responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
 - a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared; and
 - b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting.
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: April 30, 2010

/s/ A. MARK ZEFFIRO

A. Mark Zeffiro Chief Financial Officer

QuickLinks

Exhibit 31.2

Certification Pursuant to Section 302 of The Sarbanes-Oxley Act of 2002 (Chapter 63, Title 18 U.S.C. Section 1350(A) and (B))

Exhibit 32.1

Certification Pursuant to 18 U.S.C. Section 1350, As Adopted Pursuant to Section 906 of the Sarbanes-Oxley Act of 2002

In connection with the Quarterly Report of TriMas Corporation (the "Company") on Form 10-Q for the period ended March 31, 2010 as filed with the Securities and Exchange Commission on the date hereof (the "Report"), I, David M. Wathen, Chief Executive Officer of the Company, certify, pursuant to 18 U.S.C. § 1350, as adopted pursuant to § 906 of the Sarbanes-Oxley Act of 2002, that to the best of my knowledge:

- 1. The Report fully complies with the requirements of section 13(a) or 15(d) of the Securities Exchange Act of 1934; and
- 2. The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

Date: April 30, 2010

/s/ DAVID M. WATHEN

David M. Wathen *Chief Executive Officer*

QuickLinks

Exhibit 32.1

Certification Pursuant to 18 U.S.C. Section 1350, As Adopted Pursuant to Section 906 of the Sarbanes-Oxley Act of 2002

Exhibit 32.2

Certification Pursuant to 18 U.S.C. Section 1350, As Adopted Pursuant to Section 906 of the Sarbanes-Oxley Act of 2002

In connection with the Quarterly Report of TriMas Corporation (the "Company") on Form 10-Q for the period ended March 31, 2010 as filed with the Securities and Exchange Commission on the date hereof (the "Report"), I, A. Mark Zeffiro, Chief Financial Officer of the Company, certify, pursuant to 18 U.S.C. § 1350, as adopted pursuant to § 906 of the Sarbanes-Oxley Act of 2002, that to the best of my knowledge:

- 1. The Report fully complies with the requirements of section 13(a) or 15(d) of the Securities Exchange Act of 1934; and
- 2. The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

Date: April 30, 2010

/s/ A. MARK ZEFFIRO

A. Mark Zeffiro Chief Financial Officer

QuickLinks

Exhibit 32.2

Certification Pursuant to 18 U.S.C. Section 1350, As Adopted Pursuant to Section 906 of the Sarbanes-Oxley Act of 2002